# Model Medical Staff Bylaws and Rules & Regulations



# Missouri Hospital Association

in association with

**Lathrop Gage LLP** 

Copyright 2017 Missouri Hospital Association and Lathrop Gage LLP
All rights reserved. No part of this publication may be reproduced in any form or by any means without written permission from Missouri Hospital Association and Lathrop Gage LLP.
Disclaimer
This publication is designed to provide accurate and authoritative information with respect to the subject matter covered. It is provided with the understanding that the publisher is not engaged in rendering legal or other professional services. If legal advice or other expert assistance is required, the services of a competent professional person should be sought.

# Acknowledgements

On behalf of the Missouri Hospital Association, we would like to acknowledge the invaluable contributions that Stacy Harper and Donn Herring, with Lathrop Gage LLP, made to these Medical Staff Bylaws.

Stacy Harper is of counsel in the firm's healthcare department. Ms. Harper focuses her healthcare practice on regulatory compliance and healthcare reimbursement. She regularly counsels clients on compliance with Medicare regulations, fraud and abuse laws, information privacy and cybersecurity, and state licensure requirements. Ms. Harper is a certified professional coder and a former corporate compliance officer.

Donn Herring is a partner in the firm's healthcare department. He advises healthcare clients, including hospitals, physicians, federally qualified health centers, home health providers, durable medical equipment companies and nursing homes, on the full range of regulatory and business issues. He practices extensively in the areas of healthcare mergers and acquisitions (including physician/hospital joint ventures), physician integration (including physician practice acquisition, co-management arrangements and gainsharing arrangements), provider network development (including independent provider associations and physician hospital organizations), voluntary compliance program development, implementation and auditing, HIPA.A. compliance program development and implementation, certificate of need and managed care contracting.

Lathrop Gage LLP would like to recognize the contributions of Mark Gum, who worked as a summer associate at the firm in 2017. Mr. Gum attends the Washington University School of Law in Saint Louis, Missouri. Further, no endeavor such as this can be achieved without the prior work of Charles Myers who has dedicated his legal career to the service of Missouri hospitals.

© 2017 Missouri Hospital Association and Lathrop Gage LLP All Rights Reserved

# TABLE OF CONTENTS

Copyrightii Acknowledgmentsii					
	MEDICAL STAFF BYLAWS				
PREAM	Comments	2			
ARTICL	E I - NAME	6			
ARTICL 2.1	E II - PURPOSES AND RESPONSIBILITIES				
2.2	Responsibilities	7			
ARTICL	E III - MEMBERSHIP	8			
3.1	Nature of Membership				
3.2	Qualifications for Membership	9			
3.3	Obligations of Staff Membership	.11			
3.4	Duration of Appointment				
3.5	Procedures for Appointment and Reappointment				
3.6 Contract Practitioners					
3.7	Leave of Absence	.14			
ARTICL	E IV - MEDICAL STAFF CATEGORIES	.15			
4.1	Categories	.15			
4.2	Active Staff				
4.3	Courtesy Staff				
4.4	Consulting Staff1				
4.5	Honorary Staff				
4.6	Emergency Room Physicians				
4.7	Telemedicine Practitioners				
4.8	Locum Tenens				
4.9	Dentists				
4.10	Psychologists				
4.11	Podiatrists				
4.12	Provisional Period	.24			
ARTICL	E V - ALLIED HEALTH PROFESSIONALS	.26			
5.1	Nature of Health Professionals	.27			
5.2	Qualifications	.28			
5.3	Prerogatives	.28			
5.4	Obligations of AHPs	.29			
5.5	Application for Privileges	.29			
5.6	Procedure for Specification of Services	.29			
5.7	Department [Service] Assignment				
5.8	Review of Credentials	.30			

© 2017 Missouri Hospital Association and Lathrop Gage LLP All Rights Reserved

5.9	Procedural Rights	30
ARTICI	LE VI - APPLICATION, APPOINTMENT AND REAPPOINTMENT PROCEDURES	30
6.1	General	
6.2	Pre-Application	31
6.3	Application	31
6.4	Application Contents	31
6.5	Effect of Application	34
6.6	Burden of Providing Information	35
6.7	Processing Application	35
6.8	Reappointment	
6.9	Telemedicine Applications	44
ARTICI	LE VII - THE DELINEATION OF CLINICAL PRIVILEGES	45
7.1	Exercise of Privileges	45
7.2	Basis for Privileges Determination	
7.3	Department [Service] Responsibility in Defining Privileges	
7.4	Consultation and Other Conditions	
7.5	Requests	
7.6	Procedure	
7.7	Special Conditions for Dentists	
7.8	Special Conditions for Psychologists	
7.9	Special Conditions for Podiatrists	
7.10	Special Conditions for AHPs	
7.11	Emergency Privileges	
7.12	Temporary Privileges	
7.13	Telemedicine	
	LE VIII - OFFICERS	
	Officers of the Medical Staff	
8.1 8.2		
	Qualifications of Officers	
8.3	Election of Officers	
8.4	Term of Office	
8.5	Vacancies in Office	
8.6	Removal of Officers	
8.7	Duties of Officers	
	LE IX - CLINICAL DEPARTMENTS	
9.1	Current Departments, Affiliation	
9.2	Department Director; Election Qualifications and Appointment	
9.3	Department Director; Responsibilities and Authority	
9.4	Department Functions	
9.5	Assignment	
9.6	Resignation and Removal	60
ARTICI	LE X - CLINICAL SERVICES	60
10.1	Non-Departmental Staff	60
10.2	Current Services; Affiliation	
10.3	Physician Liaison; Election Qualifications and Appointment	61
10.4	Physician Liaison; Responsibilities and Authority	
10.5	Service Functions	
10.6	Assignment	62
10.7	Pasignation and Pamoval	

ARTICL	E XI - COMMITTEES	62
11.1	Peer Review Committees	63
11.2	Medical Executive Committee	63
11.3	Credentials Committee	
11.4	Joint Conference Committee	
11.5	Performance Improvement/Quality Assurance Committee	67
11.6	Utilization Review Committee	
11.7	Pharmacy and Therapeutics Committee	
11.8	Infection Control Committee	
11.9	Physician Health Committee	
11.10	Special Committees	72
ARTICL	E XII - MEDICAL STAFF COMMITTEE AND DEPARTMENT [SERVICE]	
	MEETINGS	72
12.1	Medical Staff Error! Bookman	k not defined.
12.2	Department [Service] and Committee Meetings	74
ADTICI	E XIII - INVESTIGATION AND CORRECTIVE ACTION	
13.1	Grounds for Corrective Action	
13.1	Authorization to Initiate	
13.2		
13.3	Discretionary Interview; Resolution Prior to Investigation	
13.4	Informal Interview	
13.5	Investigation	
13.7	Medical Executive Committee Action	
13.7	Effect of Medical Executive Committee Action	
13.9	Other Action	
13.10		
	Automatic Suspension or Revocation	
	Reporting	
	E XIV - HEARING AND APPELLATE REVIEW PROCEDURE	
14.1	Purpose	
14.2	Right to Hearing	
14.3	Notice of Adverse Recommendation or Action	
14.4	Request for Hearing	
14.5	Waiver by Failure to Request Hearing	
14.6	Right to One (1) Hearing and Appellate Review	
14.7	Hearing Requirements	
14.8	Hearing Procedure	
14.9	Report and Further Action	
14.10	Initiation and Prerequisites of Appellate Review	
14.11	Appellate Review Procedure	
14.12	Final Decision of the Board	
14.13	Reporting	
14.14	General Provisions	94
ARTICL	E XV - CONFIDENTIALITY, REPORTING IMMUNITY AND RELEASES	95
15.1	Special Definitions	
15.2	Authorizations and Conditions	
15.3	Confidentiality of Information	96
15.4	Immunity from Liability	
15.5	Activities Covered	97

15.6	Releases			
15.7	Cumulative Effect	98		
ARTICI	LE XVI - GENERAL PROVISIONS	98		
16.1	Staff Rules and Regulations.	98		
16.2 Urgent Amendment to Rules and Regulations				
16.3	Management of Conflicts Between the Medical Staff and the Medical Executive Co	ommittee.99		
16.4	Physician Health			
16.5	Departments [Services]			
16.6	Corporate Compliance			
16.7	Conflicts of Interest			
16.8	Staff Dues			
ARTICI	LE XVII - ADOPTION AND AMENDMENT OF BYLAWS			
17.1	Medical Staff Authority and Responsibility			
17.2	Medical Staff Action			
17.3	Board Approval			
17.4	Conflict Resolution	102		
ARTICI	MEDICAL STAFF RULES & REGULATIONS  LE 1 - GENERAL	2		
A DTICI	LE 2 - PROVISION OF PATIENT CARE	2		
2.1	Admission to Care			
2.2	Medical Necessity and Utilization Review.			
2.3	Coordination of Care			
2.4	Radiologic Services			
2.5	Consultation			
2.6	Surgical and Anesthesia Services			
2.7	Dental Services			
2.8	Podiatry Services			
2.9	Psychologist Services			
2.10	Discharge from Care			
ARTICI	LE 3 - MEDICAL RECORDS	10		
0.1	General			
3.2	History and Physical			
3.3	Operative Records			
3.4	Anesthesia Records.			
3.5	Signature Requirements			
3.6	Delinquent Records			
	LE 4 - ORDERS			
4.1	Medication and Treatment Orders			
4.2	Standing Orders			
4.3	Verbal Orders			
4.4				
	LE 5 - ALLIED HEALTH PROFESSIONALS			
5.1	Categories of AHP			
5.2	Advanced Practice Registered Nurse			
5.3	Certified Registered Nurse Anesthetist	20		

5.4	Physician Assistant	21
ARTICI	LE 6 - RESTRAINTS AND SECLUSION	23
6.1	Patient Rights	
6.2	Use of Restraints or Seclusion	
6.3	Physician Order	23
6.4	Renewal of Order	24
6.5	Face-to-Face Evaluation	24
6.6	Monitoring	24
6.7	Documentation	24
6.8	Simultaneous Restraint and Seclusion	25
6.9	Intake and Output	25
6.10	Training	25
ARTICI	LE 7 - EMERGENCY SERVICES	25
7.1	Organization	
7.2	On-Call Coverage	
7.3	Response Time	
7.4	Screening Examination	26
7.5	Stabilizing Treatment	27
7.6	Off-Campus Departments	28
7.7	Reporting	
ARTICI	LE 8 - AUTOPSIES	28
8.1	Securing Autopsies	

### **MODEL**

### **MEDICAL STAFF**

# **BYLAWS**

	OF	
A.1 1		
Adopted:		 
Revised.		

### **General Comments**

These Bylaws are intended to be a model which can be tailored to meet the needs of individual hospitals regardless of size. Because they are a model, the Bylaws are inclusive and may have far more detail than is needed for a given hospital. Comments to a number of sections are intended to focus discussion as to whether particular language is needed for a hospital's specific situation.

Language is included in this Model for a departmentalized Medical Staff, as well as optional language for a non-departmentalized Medical Staff typical of smaller hospitals. The most significant difference between non-departmental and departmental Medical Staffs is the substitution of "departments" for "services" and "departmental chairs" for "physician liaisons." Of equal significance is the fact that the departmental chairs have increased obligations over those of physician liaisons. For example, a physician liaison's obligations are couched in terms of coordinating certain activities, whereas a departmental chair will generally exercise direct administrative responsibility. As an example of this distinction, see Article IX, Sections 9.3 and 9.4 and the related Comments. Bracketed language throughout this Model indicates where the optional language may be used.

Medical Staff Bylaws often contain separate appendices for application and reappointment of members, hearing and appellate review procedures, and allied health professionals. These Bylaws incorporate all of these provisions into a single document for purposes of simplicity and understanding. Those provisions may easily be reorganized into appendices, thereby limiting the main body of the Bylaws to Medical Staff organizational matters and relegating to the appendices, in more readily identifiable sections, those specific matters of appointment and due process hearings.

In evaluating which provisions should be retained or excluded from a hospital's Medical Staff Bylaws, a long view should be taken concerning potential community needs. Development of Medical Staff Bylaws is a time-consuming process, as it should be. Thoroughness and completeness at the outset may prevent the need for frequent amendments. An important aspect of accomplishing this is to include Medical Staff leadership in the development process from the beginning.

There are references to "Missouri Law" in the document. In 2017, the passage of SB50 and SB501 by the Missouri General Assembly mandates DHSS adopt the Medicare Conditions of Participation as baseline regulation and that the department shall not adopt regulation that is duplicative or contradictory to that of the Conditions. If you have questions regarding Missouri Statute or regulations, please reach out to your legal advisors, Missouri Hospital Association or Healthcare Services Group as applicable.

# BYLAWS OF THE MEDICAL STAFF OF

### **PREAMBLE**

WHEREAS, [] ("Hospital") is a []	1
hospital organized under the laws of the State of Missouri for the purpose of providing he	ealth
care and medical services for inpatients and outpatients and promoting the well-being of	the
citizens of [], Missouri and the surrounding area; and	

**Comment:** Hospitals generally are organized in one of three classifications — not-for-profit, for profit or public (governmental). Hospital organizational documents and legal counsel should be consulted to ensure that this definition reflects accurate information.

WHEREAS, the Board of Trustees [Directors] of Hospital has charged the Medical Staff of Hospital with the responsibility for providing, monitoring and improving patient care in the Hospital; and to that end, the Medical Staff of Hospital is continually striving to achieve quality patient care for inpatients and outpatients of Hospital and accepts and agrees to discharge its responsibilities subject to the ultimate authority of the Board of Trustees [Directors].

NOW, THEREFORE, the Physicians, Dentists, Psychologists and Podiatrists practicing in the Hospital shall organize their activities in conformity with these Bylaws to carry out the functions delegated to the Medical Staff by the Board of Trustees [Directors].

**Comment:** Missouri law limits membership on the Medical Staff to Physicians, Dentists, Psychologists and Podiatrists. A Medical Staff must include Physicians, but a hospital is not required to include the other types of Practitioners. This list assumes that all of these services are provided by the Hospital. If a type of Practitioner is not included on the Medical Staff, it should be removed from this list and other sections throughout the Bylaws.

**Comment:** An issue often arises whether or not the Medical Staff is a separate organization. Most healthcare attorneys representing hospitals take the position that the Medical Staff should not be treated as a separate organization to avoid it being viewed as an entity that may be conspiring with the Hospital on issues of competition or as a separate legal entity subject to independent liability. Therefore, throughout these Bylaws, the Medical Staff is viewed as carrying out the delegated function of providing patient care on behalf of the Hospital Governing Body. Further, relevant law requires that the Hospital Governing Body be the ultimate voice with respect to operations and governance.

**Comment:** The term "Trustee" in the phrase "Board of Trustees" can be interchanged with the term "Director" or "Special Trustee." Missouri recognizes "Special Trustee" status in some city-owned hospitals where property has been gifted to the city to establish the Hospital. Legally, each of these terms can denote differing levels of fiduciary responsibility. For purposes of this document, the terms can be used interchangeably, although consistency throughout is urged to avoid confusion.

### **DEFINITIONS**

Words used in these Bylaws shall be construed to refer to the masculine or feminine gender and to singular or plural, as the context requires. The captions or headings in these Bylaws are for convenience only and are not intended to limit or define the scope or effect of any provision of these Bylaws.

- 1. "Appellate Review Body" means the group designated to conduct an appellate review pursuant to a request properly filed and pursued by a Practitioner as provided in Article XIII of the Medical Staff Bylaws.
- 2. "Allied Health Professional" or "AHP(s)" means those individuals identified and described in Article V, Sections 5.1 and 5.2, of the Medical Staff Bylaws.
- 3. "Applicant" means a Physician, Dentist, Psychologist or Podiatrist who requests to join the Medical Staff or obtain Privileges at Hospital.

**Comment:** If the Medical Staff includes a more limited group of Practitioners, this list should be revised to reflect the categories of Practitioners that may be members of the Medical Staff.

- 4. "Application" means an application for appointment to the Medical Staff as described in Article VI, Sections 6.3 and 6.4, of the Medical Staff Bylaws.
- 5. "Chief Executive Officer" is the individual appointed by the Board of Trustees [Directors] to serve as the Board's [Director's] Representative in the overall administration of the Hospital. The Chief Executive Officer may, consistent with the Chief Executive Officer's authority granted by the Hospital Bylaws, appoint a Representative to perform certain administrative duties identified in these Bylaws.
- 6. "Chief of Staff" is the Chief Administrative Officer of the Medical Staff as described in Article VII of the Medical Staff Bylaws.
- 7. "Clinical Privileges" or "Privileges" mean the permission granted to a Practitioner or Allied Health Professional to render specific diagnostic, therapeutic, medical, dental or surgical services within the Hospital.
- 8. "Dentist" means an individual who has received a Doctor of Dental Medicine or Doctor of Dental Surgery degree and is currently licensed to practice dentistry in Missouri.
- 9. "Department" or "Clinical Department" means a grouping or division of clinical services as listed in Article IX, Section 9.1, of the Medical Staff Bylaws.

**Comment:** This is a distinguishing feature from hospitals that do not have a departmental Medical Staff. A non-departmental Medical Staff will have "Services" rather than "Departments." See optional definition for "Service" hereafter.

10. "Department Director" means an Active Medical Staff Member who has been appointed in accordance with and has the qualifications and responsibilities for Department

administration as outlined in Article IX, Sections 9.2 and 9.3, and throughout these Bylaws.

**Comment:** In non-departmental Bylaws, there is no Department Director, but a Physician Liaison carries out many of the same types of duties. See optional definition for "Physician Liaison" hereafter.

- 11. "Executive Committee" and "Medical Executive Committee" shall mean the Executive Committee of the Medical Staff provided for in Article X of the Medical Staff Bylaws.
- 12. "Good Standing" means that the Medical Staff Member, at the time the issue is raised, has met the attendance and committee participation requirements during the previous Medical Staff Year, is not in arrears in dues payments, and has not received a suspension or restriction of the Medical Staff Member's appointment, admitting or Clinical Privileges in the previous twelve (12) months; provided, however, that if a Medical Staff Member has been suspended in the previous twelve (12) months for failure to comply with Hospital's policies or regulations regarding medical records and has subsequently taken appropriate corrective action, such suspension shall not adversely affect the Medical Staff Member's Good Standing status.
- "Governing Body," "Board of Trustees [Directors]" or "Board" means the Board of Trustees [Directors] of [\_\_\_\_\_\_] organized in accordance with Title 19 Section 30-20.080 of the Missouri Code of State Regulations.
  "Hearing Committee" means the committee appointed to conduct an evidentiary hearing.
- 14. "Hearing Committee" means the committee appointed to conduct an evidentiary hearing pursuant to a request properly filed and pursued by a Practitioner in accordance with Article XIII of these Medical Staff Bylaws.
- 15. "Hospital" means [\_\_\_\_\_\_], [\_\_\_\_\_\_], Missouri.
- 16. "Hospital Bylaws" mean those Bylaws established by the Board of Trustees [Directors].
- 17. "Hospital Corporate Compliance Plan" means the Corporate Compliance Plan and any related policies and procedures adopted by the Board [Directors] to promote Hospital's compliance with applicable laws and regulations.

**Comment:** This provision assumes the Hospital has a corporate compliance plan. Hospital licensure regulations require that hospitals have in place a means for complying with mandatory federal, state and local laws, rules and standards. Further, federal guidelines strongly urge adoption of a compliance plan.

- 18. "Medical Staff" or "Staff" means the Practitioners who have obtained membership status and have been granted Privileges that allow them to attend to patients and/or to provide other diagnostic, therapeutic, teaching or research services at the Hospital.
- 19. "Medical Staff Bylaws" mean these Bylaws covering the operations of the Medical Staff of [\_\_\_\_\_].

20.	"Medical Staff Rules and Regulations" mean the rules and regulations adopted by the
	Medical Staff and approved by the Board [Directors].

21.	"Medical Staff Year" is defined as the twelve (12)-month time period beginning on
	of each year and ending

**Comment:** This would typically be a calendar year. However, there may be good business reasons for a different year or for following the Hospital fiscal year if that is not a calendar year.

- 22. "Peer Review Committee" means a committee established by the Hospital or the Medical Staff for the purposes set forth and as more fully described in 537.035 RSMo.
- 23. "Physician" means an individual who has received a Doctor of Medicine or Doctor of Osteopathy degree and is currently fully licensed to practice medicine in the State of Missouri.
- 24. ["Physician Liaison" means an Active Medical Staff Member who has been appointed in accordance with and has the qualifications and responsibilities for service administration as outlined in Article IX, Sections 9.3 and 9.4, and throughout these Bylaws.]

**Comment:** This position is designed to assist in coordinating the activities of the various Clinical Services provided by the Medical Staff. The issue of whether this position is appropriate for a given hospital, based on size and other factors, is raised in the sections cited in this definition. If Physician Liaisons are not used, this definition should be deleted.

- 25. "Podiatrist" means an individual who has received a Doctor of Podiatric Medicine degree and is currently fully licensed to practice podiatric medicine in the State of Missouri.
- 26. "Practitioner" means, unless otherwise expressly provided, any Physician, Dentist, Psychologist or Podiatrist who has either: (a) applied for appointment to the Medical Staff and for Clinical Privileges; or (b) been granted an appointment to the Medical Staff and holds specific delineated Privileges; or (c) has applied for or has been granted temporary Privileges pursuant to Article VII, Section 7.10, of these Bylaws.

**Comment:** This definition is limited to the Practitioners listed because Missouri hospital licensure regulations authorize only these Practitioners to be eligible for membership on the regular Medical Staff. If all of these categories of Practitioners are not provided membership to the Medical Staff, the list should be revised accordingly.

- 27. "Prerogative" means the right to participate, by virtue of Staff category or otherwise, granted to a Medical Staff Member or Allied Health Professional, and subject to the ultimate authority of the Board [Directors] and the conditions and limitations imposed in these Bylaws and in other Hospital and Medical Staff policies.
- 28. "Psychologist" means an individual who has obtained an approved masters or doctoral degree and is currently fully licensed to practice psychology in the State of Missouri.

- 29. "Representative" or "Hospital Representative" means the Board of Trustees [Directors] and any director or committee thereof; the Hospital Chief Executive Officer or the Chief Executive Officer's designee; other employees of the Hospital; a Medical Staff organization or any member, officer, clinical division or committee thereof; and any individual appointed or authorized by any of the foregoing Representatives to perform specific functions related to gathering, analysis, use or dissemination of information.
- 30. ["Service" or "Clinical Service" means a grouping or division of clinical services as listed in Article IX, Section 9.2, of the Medical Staff Bylaws.]

**Comment:** This definition should be substituted for the definition of "Department" in Subsection 8 above for smaller, non-departmentalized Medical Staffs.

31. "Special Notice" means written notice sent via certified mail, return receipt requested or by hand-delivery evidenced by a receipt signed by the Practitioner to whom it is directed.

### **ARTICLE I - NAME**

Thece	Rylanic	addrage	"The N	Madical.	Staff of	r 1:	,
111020	Dylaws	auuress	THEN	vicuicai	Stan Or	1.	

Comment: Usually, this means the name of the Hospital. Medicare Conditions of Participation and Joint Commission Standards allow multiple hospitals in an organized healthcare system to adopt a unified or integrated Medical Staff structure. The Medical Staff of each hospital within the system must elect, by majority vote, to either (1) adopt the unified Bylaws and join the unified system Medical Staff or (2) opt out and maintain a separate and distinct Medical Staff, and associated Bylaws, for the specific hospital facility. These Model Bylaws have been drafted for application at a single hospital, but could be modified through changes in definitions to be used as unified Bylaws for an organized system.

### **ARTICLE II - PURPOSES AND RESPONSIBILITIES**

**Comment:** This list of purposes and responsibilities of the Medical Staff is consistent with the Comment in the Preamble. Besides making it clear what the purposes and responsibilities are, the list also makes it clear that the Medical Staff is an integrated, functional part of the Hospital and not a separate legal entity.

**Comment:** This list covers most areas of responsibility for the Medical Staff as well as its purposes. Responsibilities may be added or removed to fit the specific needs and circumstances of a particular hospital. Municipal hospitals may wish to add purposes unique to the Municipal entity's charter document. However, any removal of a purpose or responsibility must be carefully considered in light of the previous Comment.

- 2.1 Purposes. The purposes of the Medical Staff are:
  - (a) To be accountable to the Board [Directors] for the appropriateness of patient care services and the professional and ethical conduct of each Practitioner appointed to

- the Medical Staff and to promote patient care at Hospital that is consistent with generally recognized standards of care;
- (b) To be the formal organizational structure through which the benefits of membership on the Medical Staff may be obtained by individual Practitioners and the obligations of Medical Staff membership may be fulfilled; and
- (c) To provide an appropriate and efficient forum for Medical Staff Member input to the Board [Directors] and Chief Executive Officer on Hospital and medical issues.

### 2.2 Responsibilities. The Medical Staff's responsibilities shall include:

- (a) To participate in the performance improvement/quality assurance, quality review and utilization management of the Hospital and conduct activities required by the Hospital to assess, maintain and improve the quality and efficiency of medical care in the Hospital, including without limitation:
  - (i) Evaluating Practitioner and institutional performance through use of a valid measurement system as developed by Hospital, and based upon clinically sound criteria;
  - (ii) Monitoring critical patient care practices on an ongoing basis, including but not limited to performance of operative and other procedures;
  - (iii) Evaluating the use of medications, blood and blood products;
  - (iv) Developing clinical practice patterns and identifying and evaluating deviations from established patterns of clinical practice;
  - (v) Establishing criteria and evaluating Practitioner credentials for appointment and reappointment to the Medical Staff and for identifying the Clinical Privileges that are assigned to individual Practitioners and Allied Health Professionals in the Hospital;
  - (vi) Initiating and pursuing corrective action with respect to Practitioners when warranted; and
  - (vii) Identifying and advancing the appropriate use of Hospital resources available for meeting patients' medical, social and emotional needs, in accordance with sound resource utilization practices.
- (b) To ensure timely completion of medical records by a Physician, Dentist or other qualified licensed individual in accordance with state law and Hospital policy, including but not limited to documentation of:
  - (i) A medical history and physical examination completed no more than thirty (30) days before or twenty-four (24) hours after admission, but prior to any surgery requiring anesthesia services. If history and physical are

- documented prior to admission, an updated physical examination and medical history must be recorded within twenty-four (24) hours of admission, but prior to any surgery or procedure requiring anesthesia; and
- (ii) Complete records of discharged patients consistent with Hospital medical record policies, not to exceed thirty (30) days after dismissal.

**Comment:** The Joint Commission requires that Medical Staff Bylaws specifically address the requirements for completing and documenting medical histories and physical examinations. History and physical examination completion requirements also are addressed under Missouri law. Missouri law requires that Medical Staff Bylaws provide for the requirements regarding the completion of medical records, including a system of disciplinary action for failure to complete the records.

- (c) To make recommendations to the Board [Directors] regarding Medical Staff appointment and reappointment, including category and Department [Service] assignments, Clinical Privileges, and corrective and/or disciplinary action.
- (d) To assist in the development, delivery and evaluation of continuing medical education and training programs.
- (e) To develop and maintain Medical Staff Bylaws and policies that promote sound professional practices, organizational principles and compliance with federal and state law requirements, and to enforce compliance with such Medical Staff Bylaws, policies and laws.
- (f) To participate in the Hospital's long-range planning activity, to assist in identifying community health needs, and to participate in developing and implementing appropriate institutional policies and programs to meet those needs.
- (g) To fulfill the obligations and appropriately use the authority granted in these Medical Staff Bylaws in a timely manner through the use of Medical Staff officers, committees and individuals and to account to the Board [Directors].
- (h) To assure that at all times at least one (1) Physician member of the Medical Staff shall be on duty or available within a reasonable period of time for emergency service.

### **ARTICLE III - MEMBERSHIP**

**Comment:** This Article III is drafted to reflect the legal obligation of the Hospital and its Board [Directors] to provide appropriate levels of clinical care to its patients through the selection and retention of a competent Medical Staff. The language sets out criteria and parameters within which the Medical Staff can function in carrying out its delegated tasks.

**Comment:** An important theme of this Article III is the necessity of members of the Medical Staff to provide and demonstrate clinical competence to perform the procedures and provide the

care they desire to provide within the Hospital. The mere attainment of academic degrees, memberships and licensure in and of itself does not necessarily support the required level of clinical competence to perform the services that the Applicant proposes.

- 3.1 <u>Nature of Membership</u>. No person, including those with a contract of employment with the Hospital, may admit or provide any health care services to patients in the Hospital unless the person is a Medical Staff Member or has been granted Clinical Privileges in accordance with the procedures set forth in these Medical Staff Bylaws.
  - (a) Medical Staff appointment shall confer only the Clinical Privileges and Prerogatives granted by the Board [Directors] in accordance with these Bylaws that are commensurate with the member's qualifications, experience, present capabilities and scope of practice.
  - (b) No Applicant shall be denied membership on the basis of sex, race, creed, color, national origin, age or a handicap unrelated to the ability to fulfill patient care and required Staff obligations.
  - (c) No Practitioner shall be granted or denied Medical Staff membership or the exercise of particular Clinical Privileges at the Hospital solely because the Practitioner:
    - (i) Holds a current license or obtained a professional degree recognized by the State of Missouri;
    - (ii) Holds a particular certification or fellowship; completed a general practice residency; or is a member of a specialty board, society or body; or
    - (iii) Has previously had Medical Staff membership or Privileges in this Hospital, or is a current or former Medical Staff Member, or holds or has held Privileges in any other hospital or other health care facility.
  - (d) No application for membership on the Medical Staff shall be denied based solely upon the Applicant's professional degree or the school or health care facility in which the Practitioner received medical, dental, psychology or podiatry schooling, postgraduate training or certification, if the schooling or postgraduate training for a Physician was accredited by the American Medical Association or the American Osteopathic Association; for a Dentist, was accredited by the American Dental Association's Commission on Dental Accreditation; for a Psychologist, was accredited with accordance to Chapter 337, RSMo; and for a Podiatrist, was accredited by the American Podiatric Medical Association.
- 3.2 <u>Qualifications for Membership</u>. Every Practitioner who applies for or holds Medical Staff appointment must, at the time of Application and initial appointment and continuously thereafter, demonstrate to the satisfaction of the Medical Staff and the Board of Trustees [Directors] that the Practitioner meets all of the following qualifications for membership and any other qualifications and requirements as set forth

in these Medical Staff Bylaws, the Medical Staff Rules and Regulations, Hospital Bylaws, policies and rules, Hospital Corporate Compliance Plan, and other requirements or policies established by the Board [Directors].

- (a) Has current valid license issued by the State of Missouri to practice as a Physician, Dentist, Psychologist or Podiatrist, and has current, valid DEA registration, if applicable.
- (b) Has documentation of graduation from a school with certification by the Accreditation Counsel for Graduate Medical Education; Educational Council for Foreign Medical Graduate and on the passage of the Foreign Medical Graduate Examination in the Medical Sciences; or other applicable accreditation. Physicians shall provide documentation of satisfactory completion of an approved internship or residency. Fellowship in an institution approved for residency training shall be regarded as residency training or internship.
- (c) Has documentation evidencing an ongoing ability to provide patient care services consistent with acceptable standards of practice and available resources, including current experience, clinical results and utilization practice patterns.
- (d) Has demonstrated ability to work with and relate to people, including other Medical Staff Members, Hospital employees and administration, the Board [Directors], patients and visitors, and the community in general, in a cooperative, professional manner that maintains and promotes an environment of quality and efficient patient care.

**Comment:** Subsection 3.2(d), above, addresses an often difficult issue for Medical Staff and Hospital management. Issues of behavior, even in the face of an acceptable level of clinical competence, can create tension between Medical Staff Members and Hospital Staff that can impact patient care both directly and indirectly. Therefore, it is important to have a statement in these Bylaws that creates an appropriate expectation of acceptable behavior.

- (e) Has been in compliance with the obligations of Staff appointment as set forth in Section 3.3 of this Article III and equitable participation in the performance of Staff obligations.
- (f) Adheres to generally recognized standards of medical and professional ethics.
- (g) Demonstrates freedom from, or adequate control over, any physical or mental impairment that would significantly affect the Practitioner's ability to practice, including, but not limited to, abuse of any type of substance or chemical that affects cognitive, motor or communication ability in any manner that interferes with, or has a reasonable probability of interfering with the qualifications for membership such that patient care is, or is likely to be, adversely affected.

**Comment:** This criterion may implicate the Americans with Disabilities Act (ADA). See Comment to Article VI, Section 6.4(e), for more details.

- (h) Has the ability to read and understand the English language, to communicate effectively and intelligibly in the English language (written and verbal), and to prepare medical record entries and other required documentation in a legible and professional manner.
- (i) Has professional liability insurance of such kind, in such amount, and underwritten by such insurers as required by Missouri law or as required from time to time by resolution of the Board of Trustees [Directors] after consultation with the Medical Executive Committee, whichever requirement is more stringent.
- (j) Has never been convicted of a felony or a misdemeanor related to the Practitioner's suitability to practice the Practitioner's profession.
- (k) Is not currently excluded from or sanctioned by the Medicare or Medicaid programs or any other state or federal governmental program, and is not on the U.S. Department of Health and Human Services Office of Inspector General list of excluded providers.
- (l) In the case of new Applications for Medical Staff appointment and Clinical Privileges and with respect to Applications for changes in Clinical Privileges, the requested appointment/Privileges/affiliation must be compatible with any policies, plans or objectives formulated by the Board of Trustees [Directors] concerning:
  - (i) The Hospital's patient care needs, including current needs and projected needs;
  - (ii) The Hospital's ability to provide the facilities, personnel and financial resources that will be necessary if the Application is approved; and
  - (iii) The Hospital's decision to contract exclusively for the provision of certain medical services with a Physician or a group of Physicians other than the affected Practitioner.
- 3.3 <u>Obligations of Staff Membership</u>. Each member of the Medical Staff and each Practitioner granted temporary Privileges under these Bylaws must:
  - (a) Provide the Practitioner's patients with generally recognized professional services consistent with the recognized standards of practice in the same or similar communities and the resources locally available;

**Comment:** This provision describes a "local" standard of care. Because many areas of medical practice and hospital care are now standard throughout the United States, adopting a "national" standard of care may be worth consideration.

(b) Comply with these Bylaws; the Medical Staff Rules and Regulations; the Hospital Bylaws, policies and rules; the Hospital's Corporate Compliance Plan; and all

other standards, policies and rules of the Staff, the Hospital, and state and federal law;

**Comment:** This provision is designed to put members of the Medical Staff on notice of the need to comply with all appropriate laws and policies. Anti-discrimination laws create some special problems, particularly with respect to harassment issues. EMTALA concerns also are covered by this language. Special reference in the language of the Bylaws to specific federal and state anti-discrimination statutes and regulations may be worth consideration.

- (c) Perform any Staff, committee and Hospital functions for which the Practitioner is responsible;
- (d) Complete medical records and other records in such manner and within the time period required by Hospital for all patients the Practitioner admits or in any way provides care for in the Hospital;

**Comment:** With the implementation of electronic medical records and more specific data tracking regarding the completion of each component of documentation, increased emphasis is being placed on timely completion of medical records. Timing considerations for medical record completion and additional detail regarding medical records are found in the Model Rules and Regulations.

- (e) Abide by generally recognized standards of professional ethics; and
- (f) Satisfy the continuing education requirements established by the Medical Staff.

Practitioner's failure to satisfy any of the aforementioned qualifications or obligations may be grounds for denial of reappointment to the Staff, reduction in Staff category, restriction or revocation of Clinical Privileges, or other disciplinary action as determined in a final action of the Board [Directors] pursuant to Article XII of these Bylaws.

### 3.4 Duration of Appointment.

(a) Initial appointment shall be for a period extending to the end of the current Medical Staff year. Reappointment shall be for a period of not more than two (2) Medical Staff years. Provided, however, that the duration of any such initial appointment or reappointment shall be subject to the provisions of Article XII.

**Comment:** The duration of appointments to the Medical Staff conform to requirements in Joint Commission Standards as well as the requirement that the National Practitioner Data Bank be queried at least every two (2) years for each Physician. Missouri law also limits the term of appointment or reappointment to two (2) years.

(b) Notwithstanding the foregoing, in the case of a Practitioner providing professional services by contract with or employment by the Hospital, termination or expiration of the contract or employment or the failure to renew the contract may result in a shorter period of appointment or Clinical Privileges.

**Comment:** This language is intended to complement language that should be included in exclusive contracts and employment contracts and provides for termination of Medical Staff appointments or Clinical Privileges upon termination, expiration or failure to renew such contracts.

(c) If the Hospital adopts a policy involving a closed Department [Service] or an exclusive arrangement for a particular service or services, any Practitioner who previously held Privileges to provide such services but who is not a party to the exclusive contract/arrangement will have the Privileges for the performance of such services terminated as of the effective date of the closure of the Department [Service] or exclusive arrangement, irrespective of any remaining time on the appointment or reappointment term.

**Comment:** This language provides for termination of a Practitioner's existing Clinical Privileges if the Hospital should enter into an exclusive contract and the Practitioner whose Privileges are being terminated is not a party to that exclusive arrangement.

3.5 <u>Procedures for Appointment and Reappointment</u>. The mechanics for evaluating Applications for initial appointment and for conducting periodic reappraisals for reappointment to the Staff are outlined in Article VI of these Bylaws.

### 3.6 Contract Practitioners.

- (a) A Practitioner who is or who will be providing specified professional services pursuant to a contract with the Hospital is subject to all membership qualifications, appointment, reappointment and Clinical Privilege evaluations, and must meet all of the obligations of membership, just as any other Applicant or Staff Member.
- (b) The Staff appointment and Clinical Privileges of any Staff Member who has a contractual relationship with the Hospital, or is either an agent, employee or principal of, or partner in, an entity that has a contractual relationship with the Hospital relating to providing services to patients at the Hospital, shall terminate subject to extension by the affirmative act of the Board [Directors] upon:
  - (i) The expiration or other termination of the contractual relationship with the Hospital; or
  - (ii) The expiration or other termination of the relationship of the Staff Member with the entity that has a contractual relationship with the Hospital.

In the event of a termination of Staff appointment and/or Clinical Privileges due to Section 3.6(b)(i) or (b)(ii) above, no right to a hearing or appellate review provided in these Bylaws, including those provided in Article XIII, shall apply. Otherwise, nothing herein shall limit the Staff Member's procedural due process rights.

**Comment:** Inclusion of this provision is very important. Many hospitals have found themselves with continuing relationships with Physicians they no longer desire because either the Bylaws or contract (or perhaps both), failed to take into account Staff appointment and Privileges upon termination of a contract. The only solution in the absence of this provision and/or similar language in the contract is to utilize the fair hearing process, which can be problematic depending upon whether termination of the contract was a business decision or a quality of care decision. This language clarifies that there is no right to a fair hearing or appellate review upon "expiration or other termination" of the relevant contractual relationship. Alternative language could provide for a fair hearing and appellate review if the contract termination was based upon competency or unprofessional conduct issues.

### 3.7 Leave of Absence.

- (a) Upon a showing of good cause, Staff Members may be granted leaves of absence by the Medical Executive Committee, subject to approval by the Board [Directors], for a definitely stated period of time not to exceed six (6) months. Absence for longer than six (6) months shall constitute voluntary resignation of Medical Staff appointment and Clinical Privileges unless, upon good cause shown, the Medical Executive Committee grants an exception.
- (b) Requests for leaves of absence shall be made to the Chief of Staff and shall state the beginning and ending dates of the requested leave.
- (c) During the leave of absence, the Staff Member is not entitled to Clinical Privileges at the Hospital, and has no membership rights and responsibilities, but must continue to pay Medical Staff dues, unless otherwise waived by the Medical Staff. Prior to a leave of absence being granted, the Medical Staff Member shall have made arrangements that are acceptable to the Medical Executive Committee and Board [Directors] for the care of the Staff Member's patients during the leave of absence.
- (d) If the leave of absence is for reasons other than medical reasons, the Staff Member may be reinstated at the conclusion of the leave of absence upon filing with the Chief of Staff a written request for reinstatement and a statement summarizing continuing education, licensure or other activities related to the member's Clinical Privileges undertaken during the leave of absence. The Staff Member also shall submit such other information as requested by the Medical Executive Committee.
- (e) If the leave of absence is for medical reasons, the Staff Member must submit to the Medical Executive Committee a written request for reinstatement as well as a report from the Staff Member's attending Physician indicating that such Staff Member is physically and/or mentally capable of resuming a Hospital practice, with or without accommodation and, if with accommodation, the nature of the needed accommodation. The Staff Member also shall provide such other information as may be requested by the Medical Executive Committee.

(f) In acting upon the request for reinstatement, the Medical Executive Committee may approve reinstatement either to the same or a different Staff category and may limit or modify the Clinical Privileges to be extended to the Staff Member upon reinstatement, subject to approval by the Board [Directors].

**Comment:** The key theme in the case of leaves of absence is for the Medical Staff leadership to assure itself that the Physician on leave has not lost any of the Physician's skills or competence.

**Comment:** Other sections could be added for hospitals that experience military leaves of absence.

### **ARTICLE IV - MEDICAL STAFF CATEGORIES**

**Comment:** The categories of Medical Staff membership described in this Article IV are fairly common. The question often arises about what categories must be made available. Missouri hospital licensing regulations require an Active Staff category and permit other categories. Hospitals may have a need for categories in addition to those provided herein or may have different labels for given categories.

**Comment:** The Medicare Conditions of Participation permit hospitals to grant Medical Staff status to non-Physician Practitioners as permitted by state law and determined by the Governing Body. Under Missouri law, Medical Staff membership is limited to Physicians, Dentists, Psychologists and Podiatrists licensed in the state. The granting of Clinical Privileges to non-Medical Staff is discussed below in Article V.

- 4.1 <u>Categories</u>. The Medical Staff shall be divided into the following categories: Active, Courtesy, Consulting, Honorary, Emergency Room, Telemedicine, *Locum tenens*, Dentists, Psychologists and Podiatrists.
- 4.2 Active Staff.
  - (a) Qualifications: An Active Staff Member must:
    - (i) Reside or have a business office within sufficiently close proximity to the Hospital (as determined by the Medical Executive Committee) to enable the Staff Member to provide continuous care to the Staff Member's patients;

**Comment:** The standard for determining how close a Physician should live to the Hospital should be one of reasonableness related to patient care, as suggested in this Subsection 4.2(a) of Article IV. Arbitrary distances or geographic limitations, such as "living within the county," should be avoided because on their face they may have no rational relationship to patient care.

(ii)	Admit at least	() patients or perform
	(	() outpatient procedures at the Hospital during
	each Medical Staff year	or otherwise be regularly involved in the care of
	patients in the Hospital:	and

**Comment:** This Subsection 4.2(a)(ii) and Subsection 4.3(a)(ii) below deal with the issue of setting the number of admissions or procedures which will distinguish between Active and Courtesy Staff, respectively. Because Active Staff Members are accountable to the Governing Body for policy development and Active Staff responsibilities, such as emergency coverage and periodic meetings, they should have sufficient regular, direct contact with the Hospital and reside in sufficiently close proximity to allow them a good understanding of the Hospital's needs. On the other hand, care should be taken that the number of admissions or procedures required for Active Staff membership is not so high as to implicate anti-referral prohibitions.

- (iii) Meet the requirements as provided in Article III, Section 3.2.
- (b) Prerogatives: An Active Staff Member may:
  - (i) Admit patients without limitation, except as limited by the scope of the Staff Member's Clinical Privileges granted pursuant to Article VII or otherwise as provided in the Medical Staff Rules and Regulations;
  - (ii) Attend regular and professional meetings of the Medical Staff and of any Medical Staff committees of which the Active Staff Member is a member;
  - (iii) Vote on all matters presented at all meetings of the Medical Staff and at all committee meetings of which the Active Staff Member is a member, except as provided by resolution of the Medical Executive Committee and approved by the Board of Trustees [Directors];
  - (iv) Hold office in the Medical Staff or sit on, or act as chairperson of, any committee as determined by resolution of the Medical Executive Committee or the Board of Trustees [Directors]; and
  - (v) Exercise the Clinical Privileges granted to the Active Staff Member.
- (c) Obligations: In addition to the basic obligations set forth in Article III, Section 3.3, an Active Staff Member must:
  - (i) Contribute to the administration of the Medical Staff, including serving as a Medical Staff officer and on Hospital and Medical Staff committees as appointed or elected;
  - (ii) Participate in the performance improvement/quality assurance and utilization review activities required of the Medical Staff;
  - (iii) Discharge the recognized function of Staff membership by engaging in the Staff's teaching and continuing education programs, attending to charity patients as required, consulting with other Staff Members consistent with the Staff Member's scope of practice and delineated Privileges, supervising Practitioners during the provisional period, and fulfilling such other functions as may reasonably be required of Staff Members;

- (iv) Attend at regular and professional meetings of the Medical Staff and of any Medical Staff committees of which the Active Staff Member is a member;
- (v) Promptly pay all Medical Staff dues and assessments; and
- (vi) Serve on the on-call roster for the purpose of assignment to charity patients and provide coverage and back-up coverage in the Emergency Room as required by Hospital policies and procedures and/or Medical Staff Rules and Regulations. No Staff Member shall be entitled to a hearing or other rights of review based on denial of a request for exemption from on-call service.

**Comment:** The foregoing obligations of Active Medical Staff Members amount to minimal requirements to ensure that the Medical Staff operates smoothly and that important Hospital clinical and quality functions are adequately supported. The applicable legal standard for the legitimacy of such requirements is whether a given requirement is reasonable for carrying out the Hospital purposes, foremost of which is the provision of providing quality patient care. The same standard is applicable to all categories of Medical Staff membership, although the specific obligations may vary from category to category.

### 4.3 Courtesy Staff.

- (a) Qualifications: A Courtesy Staff Member must:
  - (i) Reside or have a business office that is within sufficiently close proximity to the Hospital (as determined by the Medical Executive Committee) to enable the Courtesy Staff Member to provide continuous care to the Courtesy Staff Member's patients, or make arrangements that are satisfactory to the Chief of Staff for alternative Physician coverage for patients for whom the Courtesy Staff Member is responsible;

(ii)	Be regularly involved in the care of patients in the Hospital with a
	maximum of() inpatient admissions or
	() outpatient procedures in any Medical Staff
	year. If a member of Courtesy Staff becomes aware that the Courtesy
	Staff Member will exceed either of these numbers, the Courtesy Staff
	Member shall make immediate Application for membership on the Active
	Staff;

**Comment:** This Subsection 4.3(a)(ii) of Article IV requires the determination of a number of admissions consistent with the standards described in the Comment to Subsection 4.2(a)(ii), above.

(iii) Have treated, admitted, consulted on, performed procedures on or
otherwise been involved in the care of such number of patients as the
Board of Trustees [Directors] shall, after receiving the recommendations
of the Medical Staff Executive Committee, determine from time to

- be reasonably required for the Physician to maintain clinical competence and familiarity with the operations of the Hospital and for the Hospital to be able to evaluate clinical competence and conduct of the Practitioner;
- (iv) Demonstrate active participation in the active Medical Staff at another hospital requiring performance improvement/quality assurance activities similar to those at this Hospital, or agree to fulfill the responsibilities of Active Staff membership specified in Subsection 4.2(c) of this Article IV concerning participation in performance improvement/quality assurance and utilization review activities at this Hospital and participation in clinical programs and attendance at committee meetings; and
- (v) Meet the requirements as provided in Article III, Section 3.2.
- (b) Prerogatives: A Courtesy Staff Member may:
  - (i) Admit patients in the same manner as an Active Staff Member as provided in Subsection 4.2(b) of this Article IV, and may exercise such Clinical Privileges as are granted to the Courtesy Staff Member. At times of full Hospital occupancy or a shortage of hospital beds or other facilities, as determined by the Chief Executive Officer, the elective patient admissions of Courtesy Staff Members shall be subordinate to those of Active Staff Members:
  - (ii) Not vote at Medical Staff meetings or hold office in the Medical Staff, except that Courtesy Staff Members may vote on any committees to which they are assigned, but shall not chair committees;
  - (iii) Not be eligible to serve on the Executive Committee; and
  - (iv) Attend regular and professional meetings of the Medical Staff.
- (c) Obligations: A Courtesy Staff Member must:
  - (i) Pay any dues or fees that may be assessed; and
  - (ii) Meet the requirements as provided in Article III, Section 3.3, and Article IV, Subsection 4.3(a).

### 4.4 Consulting Staff.

- (a) Qualifications: A Consulting Staff Member must meet the following criteria:
  - (i) Possess specialized skills needed at the Hospital for a specific project or. on an occasional basis, in consultation when requested by a Medical Staff Member;

- (ii) Demonstrate active participation on the Active Medical Staff at another hospital requiring performance improvement/quality assurance activities similar to those of this Hospital, or agree to fulfill the obligation of Active Staff membership specified in Subsection 4.2(c) of this Article IV concerning participation in performance improvement/quality assurance and utilization review activities at this Hospital and participation in clinical programs and attendance at committee meetings; and
- (iii) Meet the requirements as provided in Article III, Section 3.2.
- (b) Prerogatives: A Consulting Staff Member may:
  - (i) Perform services delineated in the Clinical Privileges granted to the Consulting Staff Member;
  - (ii) Examine patients and write consultation reports and orders for treatment or testing upon request of an Active or Courtesy Member of the Hospital;
  - (iii) Not admit patients to the Hospital, hold office on the Medical Staff or participate in any vote during meetings of the Medical Staff;

**Comment:** Subsection 4.4(b)(iii) may be modified to include any additional categorical limitations on the Privileges of Consulting Staff, such as the privilege to act as the primary surgeon for any surgical procedures.

- (iv) Serve on, but not chair, committees; and
- (v) Attend regular and professional meetings of the Medical Staff, but may not vote.
- (c) Obligations: A Consulting Staff Member must:
  - (i) Pay any dues or fees assessed; and
  - (ii) Meet the requirements as provided in Article III, Section 3.3 and Section 4.4(a) of this Article IV.

### 4.5 <u>Honorary Staff.</u>

- (a) Qualifications: The Honorary Staff shall consist of Practitioners recognized for their excellent reputations, their contributions to health and medical sciences, and/or their long-standing service to the Hospital. The Governing Body, upon recommendation of the Active Medical Staff, shall appoint Honorary Staff Members.
- (b) Prerogatives: Honorary Staff Members may not admit patients to the Hospital or exercise Clinical Privileges at the Hospital. They may, however, attend Staff meetings and any Staff or Hospital education activity. Honorary Staff Members

shall not be eligible to vote, hold office in the Medical Staff organization, and chair or serve on committees. Honorary Staff Members are not required to have malpractice insurance or pay any dues.

(c) Obligations: Honorary Staff Members shall have no assigned duties or responsibilities.

### 4.6 <u>Emergency Room Physicians</u>.

- Qualifications: The Hospital may, from time to time, enter into contracts with Physicians for the provision of services in the Emergency Room of the Hospital, and the Physicians providing such services shall be in the Staff category of Emergency Room Physician. Article III, Section 3.6 of these Bylaws, will govern a Physician covered by this Section 4.6. An Emergency Room Physician must:
  - (i) Provide services to the Hospital under contract; and
  - (ii) Meet the requirements as provided in Article III, Section 3.2.
- (b) Prerogatives: An Emergency Room Physician:
  - (i) May exercise such Clinical Privileges as are granted to the Emergency Room Physician;
  - (ii) Has no right to admit patients to the Hospital independent of the necessity to admit a patient while performing services in the Emergency Room; and
  - (iii) Is not eligible to hold office in the Medical Staff or to vote at meetings of Medical Staff or committees.
- (c) Obligations: Each such Emergency Room Physician:
  - (i) Must satisfy the basic obligations of Staff membership as described in Article III, Section 3.3 of these Bylaws; and
  - (ii) Is encouraged to attend regular and special Staff meetings, and may be asked to serve on committees, but is not required to do so.

### 4.7 Telemedicine Practitioners.

- (a) Qualifications: The Hospital may, from time to time, enter into contracts with Practitioners or other organizations for the provision of services to Hospital patients via telemedicine. A Telemedicine Staff Member must:
  - (i) Provide telemedicine services to the Hospital under a contract with the Hospital or through a third party contracted with the Hospital;

- (ii) Demonstrate active participation on the Active Medical Staff at another hospital requiring performance improvement/quality assurance activities similar to those of this Hospital, or agree to fulfill the obligation of Active Staff membership specified in Subsection 4.2(c) of this Article IV concerning participation in performance improvement/quality assurance and utilization review activities at this Hospital; and
- (iii) Meet the requirements as provided in Article III, Section 3.2.
- (b) Prerogatives: A Telemedicine Practitioner:
  - (i) May exercise such Clinical Privileges as are granted to the Telemedicine Practitioner;
  - (ii) Has no right to admit patients to the Hospital; and
  - (iii) Is not eligible to hold office in the Medical Staff or to vote at meetings of Medical Staff or committees.
- (c) Obligations: A Telemedicine Practitioner:
  - (i) Must meet the requirements of Article III, Section 3.3; and
  - (ii) Is encouraged to attend regular and special Staff meetings and may be asked to serve on committees, but is not required to do so.

### 4.8 *Locum tenens*.

- (a) Qualifications: A *locum tenens* Physician is a Practitioner who has been hired by a member of the Medical Staff to provide coverage for defined periods of time and who has been granted temporary Privileges in accordance with Article VII, Section 7.10.
- (b) Prerogatives: *Locum tenens* members neither vote nor hold any office.
- (c) Obligations: *Locum tenens* Physicians are strongly encouraged to attend general meetings of the Medical Staff and may attend committee meetings, but are not required to do so.

### 4.9 Dentists.

- (a) Qualifications: A Dentist must meet the requirements as provided in Article III, Section 3.2.
- (b) Prerogatives: Dentists admitted to the Medical Staff:
  - (i) Shall have all rights, Privileges and responsibilities as set forth in Section 4.9 of this Article IV applicable to the Dentist's Staff category, except as

- hereinafter limited and as limited in Article VII, Section 7.7, of these Bylaws entitled "Special Conditions for Dentists;" and
- (ii) May admit patients directly to the Hospital subject to the limitations described in Subsection 4.9(b)(i) above and in this Subsection 4.9(b)(ii) of Article IV. Consultation with a qualified Physician who is a member of the Medical Staff shall be required for documentation of the same basic medical appraisal as for patients admitted for other services. Dentists are responsible for documentation of the complete dental history and dental physical examination. Patients for dental care, whether admitted as inpatients or as outpatients, shall be under the care of a Physician Member of the Medical Staff. The Dentist Staff Member must request appropriate consultation when unusual or non-dental complications are encountered.
- (c) Obligations: Dental Members of the Medical Staff:
  - (i) Must meet the requirements of Article III, Section 3.3.
  - (ii) Shall be responsible for the completion of records in their field, and shall record the dental history and dental examination and a simple statement on the patient's general health; and
  - (iii) May write orders within the scope of their license, as limited by applicable law, as consistent with Medical Staff Rules and Regulations and Hospital policies and procedures, and within the scope of their Privileges as granted pursuant to these Medical Staff Bylaws.

**Comment:** Section 4.9 of this Article IV sets forth specific qualifications, Prerogatives and obligations of Dentists. Missouri law and regulations provide for Dentists to be members of the regular Medical Staff and permit them to have admitting Privileges. However, in accordance with Joint Commission Standards and commentary and Medicare Conditions of Participation (42 C.F.R. 482.12(c)(4)), require that the medical care provided to an inpatient be managed by a Physician. Thus, when a patient is admitted by a Dentist, the Hospital should require consultation with or co-admission by a Physician.

### 4.10 Psychologists.

- (a) Qualifications: A Psychologist must meet the requirements as provided in Article III, Section 3.2.
- (b) Prerogatives: Psychologists admitted to the Medical Staff:
  - (i) Shall have all rights, Privileges and responsibilities as set forth in the Section of this Article IV applicable to the Psychologist's Staff category, except as hereinafter limited and as limited in Article VII, Section 7.8, of these Bylaws entitled "Special Conditions for Psychologists;" and

- (ii) May admit patients directly to the Hospital subject to the limitations described in Subsection 4.10(b)(i) above and in this Subsection 4.10(b)(ii) of Article IV. Consultation with a qualified Physician who is a member of the Medical Staff shall be required for documentation of the same basic medical appraisal as for patients admitted for other services. Psychologists are responsible for documentation of the complete psychological history and physical examination. Patients for psychological care, whether admitted as inpatients or as outpatients, shall be under the care of a Physician Member of the Medical Staff. The Psychologist Staff Member must request appropriate consultation when unusual or non-psychological complications are encountered.
- (c) Obligations: Psychologist Members of the Medical Staff:
  - (i) Must meet the requirements of Article III, Section 3.3.
  - (ii) Shall be responsible for the completion of records in the Psychologist's field, and shall record the psychological history, examination and a simple statement on the patient's general health; and
  - (iii) May write orders within the scope of the Psychologist's license, as limited by applicable law, as consistent with Medical Staff Rules and Regulations and Hospital policies and procedures, and within the scope of the Psychologist's Privileges as granted pursuant to these Medical Staff Bylaws.

**Comment:** Section 4.10 of this Article IV sets forth specific qualifications, Prerogatives and obligations of Psychologists. Missouri law and regulations provide for Psychologists to be members of the regular Medical Staff and permit them to have admitting Privileges. However, in accordance with Joint Commission Standards and commentary and Medicare Conditions of Participation (42 C.F.R. 482.12(c)(4)), require that the medical care provided to an inpatient be managed by a Physician. Thus, when a patient is admitted by a Psychologist, the Hospital should require consultation with or co-admission by a Physician.

### 4.11 Podiatrists.

- (a) Qualifications: A Podiatrist must meet the requirements as provided in Article III, Section 3.2.
- (b) Prerogatives: Podiatrists admitted to the Medical Staff:
  - (i) Shall have all rights, Privileges and responsibilities as set forth in the Section of this Article IV applicable to the Podiatrist's Staff category, except as hereinafter limited and as limited in Article VII, Section 7.9, of these Bylaws entitled "Special Conditions for Podiatrists;" and

- (ii) May admit patients directly to the Hospital subject to the limitations described in Subsection 4.11(b)(i) above and in this Subsection 4.11(b)(ii) of Article IV. Consultation with a qualified Physician who is a member of the Medical Staff shall be required for documentation of the same basic medical appraisal as for patients admitted for other services. Podiatrists are responsible for documentation of the complete podiatric history and podiatric physical examination. Patients for podiatric care, whether admitted as inpatients or as outpatients, shall be under the care of a Physician Member of the Medical Staff. The Podiatrist Staff Member must request appropriate consultation when unusual or non-podiatric complications are encountered.
- (c) Obligations: Podiatrist Members of the Medical Staff:
  - (i) Must meet the requirements of Article III, Section 3.3.
  - (ii) Shall be responsible for the completion of records in the Podiatrist's field, and shall record the podiatric history, podiatric examination and a simple statement on the patient's general health; and
  - (iii) May write orders within the scope of the Podiatrist's license, as limited by applicable law, as consistent with Medical Staff Rules and Regulations and Hospital policies and procedures, and within the scope of the Podiatrist's Privileges as granted pursuant to these Medical Staff Bylaws.

**Comment:** Section 4.11 of this Article IV sets forth specific qualifications, Prerogatives and obligations of Podiatrists. Missouri law and regulations provide for Podiatrists to be members of the regular Medical Staff and permit them to have admitting Privileges. However, in accordance with Joint Commission Standards and commentary and Medicare Conditions of Participation (42 C.F.R. 482.12(c)(4)), require that the medical care provided to an inpatient be managed by a Physician. Thus, when a patient is admitted by a Podiatrist, the Hospital should require consultation with or co-admission by a Physician.

### 4.12 Provisional Period. Scope and Duration:

All new Medical Staff appointments and all grants of initial or increased Clinical Privileges to any members are provisional for a period of up to one (1) year ("Provisional Period"). With respect to Applicants who are granted temporary Privileges during the pendency of their Application for Medical Staff membership and Privileges, the Provisional Period shall not run during the period of temporary Privileges, but shall begin at such time as the Applicant's Application is approved. During the Provisional Period, a provisional appointee's performance will be reviewed and evaluated by the Department Director [Physician Liaison], or the Department Director's [Physician Liaison's] designee, for the Department [Service] with which the Practitioner is primarily affiliated, and by the Department Director [Physician Liaison] of any other Department [Service] in which the Applicant has Privileges, or by any Active Staff Member specifically

- assigned the task by the applicable Department Director [Physician Liaison], and the Department Director [Physician Liaison] shall certify successful completion of the Provisional Period.
- (b) No Effect on Membership or Exercise of Privileges: During the Provisional Period, a Practitioner must meet all qualifications, can exercise all of the Prerogatives, and must fulfill all of the obligations of the Practitioner's Staff category; and the Practitioner can utilize all of the Clinical Privileges granted to the Practitioner.
- (c) Review Requirements: The Executive Committee, subject to approval of the Board of Trustees [Directors], shall establish requirements necessary for any review that is intended to conclude the Provisional Period. The provisional Practitioner must arrange for the number and type of cases required for review and observation by the relevant Department Director [Physician Liaison] or other designated person within a time frame that results in the required cases being reviewed/observed before the end of the Provisional Period.

**Comment:** The Joint Commission requires that accredited hospitals have a "focused professional practice evaluation" process to evaluate the Privilege-specific competence of a Practitioner who does not have documented evidence of competency at the organization. This is required to be implemented for all newly requested Privileges. In addition to the Provisional Period in the Bylaws, the Hospital should have specific policies and procedures regarding how it conducts this evaluation and review process.

- (d) Procedure for Concluding or Extending the Provisional Period: Prior to the end of the Provisional Period, the Practitioner must submit to the Credentials Committee a request for declaration that all or any part of the Practitioner's Provisional Period has been successfully concluded or a request for an extension of the Provisional Period. A new appointee's request must be accompanied by one (1) or more signed statements described in Subsection 4.11(d)(i) and (ii) of this Article IV. The request of an existing Staff Member with respect to increased Privileges must be accompanied by one (1) or more signed statements described in Subsection 4.11(d)(ii) of this Article IV. The statements that must be furnished are as follows:
  - (i) From the Department Director [Physician Liaison] of the Department [Service] in which the Practitioner's appointment was made attesting that by observed performance, the Practitioner has demonstrated the Practitioner's qualifications for Staff membership in the Practitioner's Staff category, that the Practitioner has not abused the Practitioner's Prerogatives, and that the Practitioner has discharged the Practitioner's membership obligations; and
  - (ii) From the Department Director [Physician Liaison] of each Department [Service] in which the Practitioner was granted initial or increased Clinical

Privileges, that the Practitioner has satisfactorily demonstrated the Practitioner's ability to exercise those Privileges.

Failure, without good cause, of a Practitioner to act to conclude or extend the Provisional Period is deemed a voluntary relinquishment of Medical Staff membership or the Clinical Privileges provisionally granted, or both, as applicable.

- (e) Procedure for Processing Request: Upon receipt of the request and statement(s) described in Subsection 4.11(d) of this Article IV, the Credentials Committee shall prepare and forward to the Medical Executive Committee a written report with recommendations and supporting documentation. Final processing shall follow the procedures set forth in the Application process as described in Article VI, Subsections 6.7(d) through (i). For purposes of concluding the Provisional Period, an "adverse recommendation" by the Medical Executive Committee or an "adverse action" by the Board [Directors] shall be as defined in Article XIII of these Bylaws.
- (f) Extension: If an initial appointee or a Medical Staff Member is unable to obtain the statement(s) required by Subsection 4.11(d) above because of insufficient patient contacts or because the Practitioner's caseload at the Hospital was inadequate to demonstrate ability to exercise the Privilege in question, the Practitioner may submit to the Credentials Committee a statement to this effect describing the Practitioner's caseload and signed by the Department Director [Physician Liaison] of the applicable Department [Service]. Upon receipt of such statement, the Provisional Period for exercising the Privilege or Privileges involved shall be extended for a period not to exceed one (1) additional year unless the Medical Executive Committee or the Board of Trustees [Directors], after receiving the report of the Credentials Committee, determines such extension is inappropriate.
- (g) Procedural Rights: Whenever any Provisional Period expires without favorable conclusion for the Practitioner, or whenever an initial or additional extension is denied, the Chief Executive Officer will provide the Practitioner with Special Notice of the adverse result and of the Practitioner's entitlement to the procedural rights provided in Article XIII of these Bylaws.

### ARTICLE V - ALLIED HEALTH PROFESSIONALS

**Comment:** Missouri hospital licensing regulations provide that the Hospital Governing Body may authorize in the organization's Bylaws that licensed health care professionals who are not entitled to regular Medical Staff membership can nevertheless be provided Clinical Privileges to function in a clinical capacity on an outpatient basis. These Bylaws list a number of health care providers who are not licensed Practitioners and who may require varying degrees of supervision depending on the scope of their training, experience and education.

**Comment:** The provisions below describe procedures for establishing (i) categories of Allied Health Professionals that may be granted Clinical Privileges, (ii) any sponsorship requirements

for each category of professional, (iii) education, training and experience requirements with respect to each category, (iv) licensure, certification or other credentials required for each category, (v) malpractice insurance required for each category, (vi) Clinical Privileges available for each category, and (vii) any other special requirements.

Comment: Section 5.9 of this Article V provides that the Board [Directors] shall determine what procedural due process and fair hearing rights may apply to any category of Allied Health Professional. If the Board [Directors] establishes [establish] due process rights differently for Allied Health Professionals than those granted for Medical Staff Members, as it is allowed to do, then the Board [Directors] must have in place the detail of the separate due process procedure for Allied Health Professionals. It should be noted that the Health Care Quality Improvement Act of 1986 (HCQIA) allows for (but does not mandate as it does for Physicians and Dentists) reporting with respect to other health care providers. If the Hospital wishes to report with respect to these other health care providers, it should provide the procedural due process required by HCQIA.

5.1 Nature of Health Professionals. An Allied Health Professional ("AHP") is an individual other than a licensed Practitioner who functions in a medical support role to a Practitioner or exercises independent judgment within the area of the AHP's professional competence and who is qualified by licensure, certification or other approval to render medical or surgical care under the supervision of a Practitioner who has been accorded Privileges to provide such care in the Hospital. The following professionals may be AHPs for the purposes of this Section 5.1: Advanced Practice Registered Nurses, Chiropractors, Nurse Anesthetists, Anesthesia Assistants, Optometrists, and Physician's Assistants. Individuals in this category are not members of the Medical Staff and shall have only such limited duties, responsibilities and Prerogatives as may be specifically set forth herein. AHPs will not be eligible to vote or hold office in the Medical Staff. AHPs may be invited to attend Medical Staff meetings. The Board [Directors] shall specify by policy or regulation, the classes of AHP that may be granted Clinical Privileges in the Hospital. In establishing such classes, the Board [Directors] shall consider such factors as needed in the Hospital for the types of services provided by the particular classes of AHP and the availability of Medical Staff Members appropriately trained to oversee the type of services provided by a particular class of AHP. Certain categories of AHP shall require Physician sponsors to perform any services in the Hospital. Guidelines for each AHP category shall set forth whether sponsorship is required, and the procedures for obtaining said sponsorship.

The Medical Staff Executive Committee may make a recommendation to the Board [Directors] regarding the types of AHP to be granted Clinical Privileges in the Hospital, and the Board [Directors] shall consider such recommendation prior to making its decision.

**Comment:** The categories of professionals that are provided Clinical Privileges as AHPs may vary for each Hospital. The above language includes the most frequent categories of AHP, but should be modified to reflect Hospital needs. Some hospitals also include categories such as Audiologists, Bacteriologists, Chemists, Clinical Pharmacologists, Dental Auxiliary, Nuclear

Medicine Technicians, Orthopedic and other Surgical Technicians, Physicists, Physiologists, Psychiatric Social Workers, Scrub Technicians, Speech Pathologists, and Qualified Therapists (e.g., Occupational, Physical, Respiratory). Where these additional categories of professionals only provide services as employees or under contract with the Hospital, it is not necessary to credential them as AHP. However, if the Hospital will allow any professional in a category to provide services without employment or a contractual relationship, the entire category, including employees and contracted professionals, would need to be credentialed.

- 5.2 Qualifications. To be eligible for Clinical Privileges within the Hospital, an AHP must:
  - (a) Be within an AHP category approved for Clinical Privileges by the Board [Directors];
  - (b) Meet the personal qualifications specified in Article III, Section 3.2 other than Subsections 3.2(a) and (b) thereof;
  - (c) Provide evidence of adequate education, training and experience with respect to the services provided and as determined by Board [Directors] policy for each type of AHP;
  - (d) Hold a license, certificate or such other credentials as may be required by applicable state law;
  - (e) Provide proof of malpractice insurance in an amount required by the Board [Directors]; and
  - (f) Meet all other requirements imposed by the Board [Directors] for the individual's category of AHP.
- 5.3 <u>Prerogatives</u>. An AHP's Prerogatives are to:
  - (a) Provide specifically designated patient care services under the direct or ultimate (as determined by Board [Directors] policy or regulation) supervision or direction of a Medical Staff Member;
  - (b) Write orders only to the extent specified in the Medical Staff Rules and Regulations, Hospital policies, or the position description developed for that category of AHPs;
  - (c) Exercise such other Prerogatives as the Medical Executive Committee, with the approval of the Board [Directors], grants any general or specific category of AHPs; and
  - (d) Serve on Medical Staff committees when appointed. AHPs shall not have the privilege of voting or holding office in the Medical Staff. AHPs may, however, participate in committees and vote at committee meetings.

An AHP may not independently admit patients. Each patient's general medical condition and care shall be the ultimate responsibility of a qualified Physician Member of the Medical Staff.

## 5.4 <u>Obligations of AHPs</u>. Each AHP shall:

- (a) Meet the basic responsibilities contained in Article III, Section 3, designed for Medical Staff Members:
- (b) Assume responsibility to the extent applicable under the AHP's scope of practice for the care and supervision of each patient in the Hospital for whom the AHP is providing services;
- (c) Participate as requested in performance improvement/quality assurance program activities and in discharging related performance improvement/quality assurance duties as may be required from time to time;
- (d) Attend clinical and educational meetings of the Hospital and/or Medical Staff as requested as well as meetings of committees of which the AHP is a member; and
- (e) Refrain from any actions that are or may be reasonably interpreted as being beyond, or an attempt to exceed, the AHP's scope of practice under state law and as authorized by the Hospital.
- Application for Privileges. Every AHP who seeks or enjoys Clinical Privileges must make written Application for such Privileges or for any increase in Privileges. Applications for appointment or reappointment of Clinical Privileges of AHPs shall be processed in accordance with the procedures established by the Medical Executive Committee and approved by the Board [Directors]. Included in the Application shall be the name of the Medical Staff Member who will remain the sponsor of the Applicant until a change of sponsor is granted. The Applicant or the Applicant's sponsor shall submit a written statement of the clinical duties and responsibilities for which the Applicant is requesting Clinical Privileges. The Applicant shall agree to abide by these Bylaws to the extent they are applicable to AHPs and other Medical Staff and Hospital Bylaws, Rules, Regulations, policies and procedures (including, but not limited to, Hospital's Corporate Compliance Plan).
- 5.6 Procedure for Specification of Services. Written guidelines for specified services that may be provided by each category of AHP will be developed by the Credentials Committee, subject to approval by the Medical Executive Committee and the Board of Trustees [Directors] and with input, as applicable, from the Department Director [Physician Liaison] of the Clinical Department [Service] involved, the AHP's sponsor, and, as appropriate, other Representatives or groups of the Medical Staff and Hospital administration. For each category of AHP, such guidelines must include at least:
  - (a) Minimum requirements for prior training and experience;

- (b) Specification of the types of patients that may be seen;
- (c) A description of services and procedures to be provided, including any special equipment, procedures or protocol that may be required, and the requirements for recording of Services provided in the patient's medical record;
- (d) Definition of the degree of assistance that an AHP may require from a Medical Staff Member (including the degree of Staff Supervision required for each Service) in the treating of patients on Hospital premises and any limitations applicable to each service; and
- (e) Whether the category of AHP shall require sponsorship and, if so, procedure(s) for the designation of said sponsor.
- 5.7 <u>Department [Service] Assignment.</u> An AHP shall be individually assigned, when appropriate, to the Department [Service] appropriate to the AHP's professional training, and shall be subject to an initial probationary period, formal periodic reviews and disciplinary procedures as determined for the AHP's category.
- 5.8 <u>Review of Credentials</u>. AHP's credentials shall be reviewed on at least an annual basis the first two (2) years of the AHP's appointment and at least every two (2) years thereafter.
- 5.9 <u>Procedural Rights</u>. An AHP is not eligible for the procedural due process rights as provided for Medical Staff Members in Article XIII of these Bylaws unless otherwise determined by the Board [Directors] for the AHP's specific category of AHP. The Board [Directors] shall determine, consistent with applicable law, what procedural due process rights (and commensurate reporting requirements) shall apply to any category of AHP. The Board [Directors] shall establish a policy and procedure delineating criteria for and specific rights to fair hearing and appeal procedures for AHPs.

# ARTICLE VI - APPLICATION, APPOINTMENT AND REAPPOINTMENT PROCEDURES

Comment: This Article VI attempts to balance the needs of the Hospital with fair treatment of the Applicant for Medical Staff Privileges. The general theme is that the burden is on the Applicant, but provisions are made to ensure fairness to the Applicant. The Article is very detailed. The Hospital must be prepared to adhere to the letter as well as the spirit of this Article VI in order to fully protect itself from charges of unfairness or impropriety from Applicants who are denied credentials or Privileges. Missouri courts have granted injunctive relief to Physicians where a Hospital did not adhere to the process described in the Bylaws.

6.1 <u>General</u>. Unless otherwise provided for herein, no person shall exercise Clinical Privileges in the Hospital unless and until the person applies for and receives Medical Staff appointment and/or such Privileges are granted, as set forth in these Bylaws. Appointment to Medical Staff shall confer only those specific Clinical Privileges as have been granted in accordance with these Bylaws.

6.2 Pre-Application. An Applicant desiring appointment to the Medical Staff shall obtain a pre-Application for Medical Staff appointment form from the Chief Executive Officer of the Hospital, complete the pre-Application form in full and return it to the Chief Executive Officer. The specific contents of the pre-Application form shall be as determined by the Board of Trustees [Directors] after consultation with the Executive Committee of the Medical Staff and shall be set forth in the Medical Staff Rules and Regulations. Generally, the form shall require information regarding medical or dental school completion; for Physicians and other Staff Members as applicable, residency training as well as verification of an unrestricted license to practice medicine in the State of Missouri and DEA registration, if applicable. The pre-Application form also shall request that the pre-Applicant indicate generally the Privileges for which the pre-Applicant will be applying. The Chief Executive Officer and the Chief of Staff will review completed pre-Application forms to determine whether the Applicant shall be issued an Application for Medical Staff appointment. When reviewing the completed pre-Application forms, the Chief Executive Officer and the Chief of Staff shall consider whether the pre-Applicant has the requisite training and licensure, whether the Hospital has a need for the type of services the pre-Applicant proposes to perform, and the appropriate facilities and support personnel for the Privileges to be requested. If the pre-Applicant receives an Application for Medical Staff appointment, the Chief Executive Officer shall, in addition to supplying the Application form, supply the Applicant with a copy of the Hospital Bylaws the Hospital Corporate Compliance Plan, and the Bylaws, and Rules and Regulations of the Medical Staff. The Chief Executive Officer also shall provide to the Applicant a brief orientation to the Medical Staff Bylaws, including a description of the mechanism for appointment and reappointment, in such format as determined by the Medical Executive Committee and approved by the Board [Directors].

**Comment:** The pre-Application process allows for an evaluation of the basic threshold requirements for Medical Staff membership as well as a determination of whether the Hospital has appropriate facilities for the Clinical Privileges sought. If the Practitioner does not meet the threshold requirements or seeks Clinical Privileges for services which the Hospital cannot support or does not wish to offer, the Hospital can decline to offer the Practitioner an Application without triggering due process requirements or the reporting requirements of HCQIA because the denial of the Application is not based upon clinical competence considerations.

- 6.3 <u>Application</u>. A written, signed Application for Medical Staff appointment must be submitted to the Chief Executive Officer on the Application form approved by the Board [Directors].
- 6.4 <u>Application Contents</u>. Every Application must include at least the following:
  - (a) A statement that the Applicant has been offered and/or received and read the Hospital Bylaws, the Medical Staff Bylaws and the Medical Staff Rules and Regulations, and the Hospital Corporate Compliance Plan; that the Applicant agrees to be bound by the terms thereof if the Applicant is granted membership and/or Clinical Privileges; and that the Applicant agrees to be bound by the terms thereof in all matters relating to consideration of the Applicant's Application

- without regard to whether the Applicant is granted membership or Clinical Privileges.
- (b) Any post-secondary school training, including the name of the institutions and the dates attended, any degrees granted, course of study or program completed, and for all post-graduate training, names of persons responsible for reviewing the Applicant's performance.
- (c) A copy of all currently valid professional licenses or certifications, DEA registration, and any other controlled substances registration, including the date of issuance and license or provider number.
- (d) Records verifying any specialty or subspecialty board certification, recertification, or eligibility to sit for such board's examination.
- (e) A statement as to whether the Applicant's health status is such that the Applicant is able to perform all the procedures for which the Applicant has requested Privileges, with or without reasonable accommodation, according to accepted standards of professional performance and without posing a direct threat to patients. If the Applicant will require reasonable accommodation, the Application should include a separate sheet to describe the accommodation(s) that will enable the Applicant to perform the clinical activities for which the Applicant has requested Privileges.

**Comment:** This Subsection 6.4 is one of several Sections in the Model Medical Staff Bylaws intended to address the issue of the applicability of the ADA and the Missouri Human Rights Act to Medical Staff membership. There is continuing tension between a Hospital's obligations to provide quality care through qualified, capable Practitioners while at the same time complying with the ADA. Case law suggests that the courts may apply the requirements of the ADA in the Medical Staff context. The recommended course of action is to assume that the ADA does apply.

The ADA requires "employers" to which the law applies to make a "reasonable accommodation" in certain circumstances. The ADA also has certain restrictions on employers with respect to seeking medical information from Applicants and requiring medical examinations of Applicants. Related provisions in these Bylaws include Article III, Subsection 3.2(g), and Article VI, Subsection 6.4(e), Subsection 6.7(i) and Subsection 6.8(b)(v). These provisions require (i) that Medical Staff Members demonstrate freedom from or adequate control over physical or mental impairment that would significantly affect their ability to practice; (ii) that Applicants state whether their health status is such that they are able to perform all procedures for which they request Privileges with or without reasonable accommodation, and to describe any such reasonable accommodation that may be necessary; (iii) that Hospitals make "offers" of Medical Staff membership and Privileges contingent upon review and ascertainment of adequate health status; and (iv) that Hospitals consider and maintain information regarding health status for reappointment.

- (f) Documentation verifying professional liability coverage, including the names of present and past insurance carriers, and any information related to Practitioner's malpractice claims history and experience during the past five (5) years.
- (g) The nature and specifics of any prior actions involving denial, revocation, non-renewal or other challenges or voluntary relinquishment (by resignation or expiration) of any professional license or certificate to practice in Missouri or in any other state or country; any controlled substances registration; membership or fellowship in local, state or national organizations; specialty or sub-specialty board certification or eligibility; faculty membership at any medical or other professional school; Medical Staff membership, Prerogatives or Clinical Privileges at any other health care institution, including any hospital, clinic, skilled nursing facility or managed care organization in this or any other state; participation in a federal or state health care program; or professional liability insurance.
- (h) Location of the Applicant's office(s); names and addresses of other Practitioners with whom the Applicant is or has been associated and the dates of the associations; names and locations of all health care institutions or organizations (including third-party payors) with which the Applicant had or has any association, employment, Privileges or practice and the dates of each affiliation; and status held, general scope of Clinical Privileges or duties and documentation of conformity with applicable Hospital and Medical Staff Bylaws, Rules and Regulations at such other institutions where the Applicant had Privileges.
- (i) The Medical Staff category, Clinical Department [Service] assignment and Clinical Privileges requested.
- (j) The status and, if applicable, resolution of any past or current criminal charges against the Applicant.
- (k) The names of at least three (3) medical or health care professionals in the Applicant's same profession or a member of the local county or regional medical society (for use as professional references) who have known the Applicant for a minimum of [\_\_\_\_\_ (\_\_)] years and who, through observation, have personal knowledge of the Applicant's Clinical ability, ethical character, the effect of the Applicant's health status, if any, on the Privileges sought, ability to work cooperatively with others, and who are willing to provide specific written comments on these matters upon request from the Medical Staff or Hospital. At least one (1) of such individuals must have had organizational responsibility for supervision of the Applicant's performance.

**Comment:** The length of time a reference has known the Applicant is optional but must be reasonable and rationally related to the purpose of this provision. The Hospital may not require references to be associated with itself or with the local medical society because of possible antitrust ramifications and the unreasonable delegation of control over Medical Staff selection to an outside organization. Municipal Hospitals enjoy certain exemptions from antitrust laws.

- (l) A listing and description of any potential conflict(s) of interest with the Hospital or its related entities (including any ownership or contractual interest the Applicant or the Applicant's immediate family members might have with the Hospital or with entities that do business with the Hospital).
- (m) Such other information as the Board [Directors] may require, subject to existing legal requirements.
- (n) The Applicant's signature.
- 6.5 <u>Effect of Application</u>. By signing and submitting an Application for appointment to the Medical Staff, the Applicant:
  - (a) Acknowledges and attests that the Application is correct and complete, and acknowledges that any significant misstatement or omission is grounds for a denial of appointment or for a summary dismissal from the Medical Staff.
  - (b) Agrees to appear for personal interviews, if required, in support of the Application.
  - (c) Consents to the release and review by Hospital Representatives of all documents (including requesting and reviewing information from the National Practitioner Data Bank and any other data bank the Hospital is permitted or required by law to access) that may be necessary to evaluate the Applicant's professional qualifications and ability to carry out the Clinical Privileges the Applicant requests as well as the Applicant's professional ethical qualifications for Staff membership and consents to Hospital Representatives consulting with prior associates or others who may have information bearing on the Applicant's professional or ethical qualifications and competence.
  - (d) Understands and agrees that if Medical Staff membership or requested Clinical Privileges are denied based on the Applicant's professional competence or conduct, the Applicant will be subject to reporting to the National Practitioner Data Bank.
  - (e) Releases from any liability all Hospital Representatives for their acts performed in good faith and without malice in connection with reviewing, evaluating or acting on the Application and the Applicant's credentials.
  - (f) Releases from any liability all individuals and organizations who provide information, including otherwise privileged or confidential information, to Hospital Representatives in good faith and without malice concerning the Applicant's ability, professional ethics, character, physical and mental health, emotional stability, and other qualifications necessary for appointment as discussed herein.

- (g) Agrees that any lawsuit brought by the Applicant against an individual or organization providing information to a Hospital Representative, or against the Hospital or Hospital Representative, shall be brought in a court, federal or state, in the state in which the defendant resides or is located.
- (h) Agrees to practice in an ethical manner and to provide continuous care to patients.
- (i) Agrees to notify the Chief of Staff and the Chief Executive Officer immediately if any information contained in the Application changes. The foregoing obligation shall be a continuing obligation of the Applicant so long as the Applicant is a member of the Medical Staff and/or has Clinical Privileges at the Hospital.
- (j) Agrees to be bound by the terms of and to comply in all respects with these Medical Staff Bylaws, Medical Staff Rules and Regulations, Hospital Bylaws, policies and procedures, and Hospital's Corporate Compliance Plan.
- 6.6 <u>Burden of Providing Information</u>. The Applicant is responsible for producing information adequate to properly evaluate the Applicant's experience, background, training, demonstrated competence, utilization patterns, work habits (which include the ability to work cooperatively with others), and, upon request of the Executive Committee, Chief Executive Officer or Chief Executive Officer's designee, or the Board [Directors], physical and mental health status, and to resolve any doubts or conflicts and to clarify information as requested by appropriate Staff or Board [Directors] authorities.

**Comment:** These provisions make it clear what is expected of an Applicant and that the information requested in the Application must be updated from time to time as appropriate.

**Comment:** Permission to access information from appropriate agencies, such as the National Practitioner Data Bank and state professional licensure agencies, is particularly important because applicable rules generally deny access to information by a third party without consent of the Practitioner. Federal law requires the appointment and reappointment process to include a screening of the National Practitioner Data Bank.

**Comment:** This Section 6.6 of Article VI makes it clear that the Applicant bears the affirmative obligation to provide any and all information necessary to support qualification for Medical Staff membership. The Hospital should avoid the obligation of proving the Applicant is not qualified for membership.

## 6.7 <u>Processing Application</u>.

(a) The completed Application shall be submitted to the Chief Executive Officer who shall immediately forward the same to the Credentials Committee for processing. The Credentials Committee or its designee shall be responsible for collecting and verifying all qualification information received, and for promptly notifying the Applicant of any problems with obtaining required information. Upon notification of any problems or concerns, the Applicant must obtain and furnish the required information. If the Applicant fails to furnish the requested

information within ninety (90) days of a written request, the Application shall be finally and conclusively deemed denied without right to a hearing or appellate review, and the Applicant shall be so informed by the Chief Executive Officer. The Credentials Committee or its designee shall query the National Practitioner Data Bank and any other data bank as permitted or required by law. The Credentials Committee or its designee also shall check the U.S. Department of Health and Human Services Office of Inspector General Cumulative Sanction report, the General Services Administration List of Parties Excluded from Federal Procurement and Non-Procurement Programs, and any other appropriate sources to determine whether the Applicant has been convicted of a health care related offense, or debarred, excluded or otherwise made ineligible for participation in federal health care programs. When the collection and verification process is accomplished, the Credentials Committee shall transmit the Application and all supporting materials to the Department Director(s) [Physician Liaison(s)] of each Clinical Department [Service] in which the Applicant seeks Privileges.

- (b) The Department Director [Physician Liaison] (or the Department Director's [Physician Liaison's] designee) of each Clinical Department [Service] in which the Applicant seeks Privileges is responsible for reviewing the Application and any supporting documentation, and shall prepare a written report evaluating the evidence of the Applicant's training, experience and demonstrated ability and stating how the Applicant's skills are expected to contribute to the clinical and educational activities of the Department [Service]. This report should be forwarded to the Credentials Committee and must state the Department Director's [Physician Liaison's] recommendation as to approval or denial of, and any special limitations on, Medical Staff appointment, category of Medical Staff membership and Prerogatives, Clinical Department [Service] affiliation. and Staff Clinical Privileges. Before submitting a report to the Credentials Committee, the Department Director [Physician Liaison] may, at the Department Director's [Physician Liaison's] discretion, conduct an interview with the Applicant.
- (c) After receipt of the Department Director's [Physician Liaison's] report, the Credentials Committee is responsible for reviewing the report, all Application materials and any other relevant information available to it. The Credentials Committee is then responsible for preparing and submitting a written report and recommendations to the Medical Executive Committee as to approval or denial of, and any special limitations on, Medical Staff appointment, category of Medical Staff membership and Prerogatives, Department affiliation, and Clinical Privileges.
- (d) After receipt of the Credentials Committee report, the Medical Executive Committee, at its next regularly scheduled meeting, shall review the Application, the supporting documentation, the reports and recommendations from the Department Director(s) [Physician Liaison(s)] and Credentials Committee, and any other relevant information available to it. The Medical Executive Committee shall vote on the Application and take one of the actions described in 6.7(e).

- (e) The Medical Executive Committee may take any of the following actions:
  - (i) <u>Deferral Action</u>. A decision by the Medical Executive Committee to defer any action on the Application must be revisited, except for good cause, within thirty (30) days with subsequent recommendations as to approval or denial of, or any special limitations on, Medical Staff Appointment, category of Medical Staff membership and Prerogatives, Department [Service] affiliation, scope of Clinical Privileges, and length of Provisional Period. The Chief Executive Officer of the Hospital shall promptly send the Applicant written notice of a decision to defer action on the Applicant's Application.
  - (ii) <u>Favorable Recommendation</u>. If the Medical Executive Committee makes a favorable recommendation regarding all aspects of the Application, the Medical Executive Committee shall promptly forward its recommendation, together with all supporting documentation, to the Board of Trustees [Directors].
  - (iii) Adverse Recommendation. If the Medical Executive Committee's recommendation is adverse to the Applicant, the Medical Executive Committee must immediately inform the Applicant by Special Notice of the recommendation, and the Applicant shall then be entitled to the procedural rights as provided in Article XIII. No such adverse recommendation shall be required to be forwarded to the Board of Trustees [Directors] until after the Applicant has exercised or has been deemed to have waived the Applicant's right to a hearing as provided in Article XIII of these Bylaws.
- (f) The Board of Trustees [Directors] may take any of the following actions:
  - (i) Favorable Medical Executive Committee Recommendation. The Board of Trustees [Directors] may adopt or reject any portion of the Medical Executive Committee's recommendation that was favorable to an Applicant or refer the recommendation back to the Medical Executive Committee for additional consideration, but must state the reason(s) for the requested reconsideration and set a time limit within which a subsequent recommendation must be made. If the Board's [Directors'] action is favorable, the action shall be effective as its final decision. If the Board of Trustees' [Directors'] decision on receiving a favorable Medical Executive Committee recommendation is adverse to the Applicant, the Board [Directors] shall so notify the Applicant by Special Notice, and the Applicant shall be entitled to the procedural rights provided in Article XIII of these Bylaws.
  - (ii) Without Benefit of Medical Executive Committee Recommendation. If the Medical Executive Committee fails to make a recommendation within the time required, the Board of Trustees [Directors] may, after informing

the Medical Executive Committee of its intent, and allowing a reasonable period of time for response by the Medical Executive Committee, make its own determination using the same type of criteria considered by the Medical Executive Committee. If the Board's [Directors'] decision is adverse to the Applicant, the Board [Directors] shall promptly so inform the Applicant by Special Notice, and the Applicant shall then be entitled to the procedural rights provided in Article XIII of these Bylaws.

(iii) Adverse Medical Executive Committee Recommendation. If, after adverse recommendation by the Medical Executive Committee, the Practitioner waives rights to a hearing under Article XIII of these Bylaws, the recommendation will be submitted to the Board [Directors] for final determination. If the Practitioner requests a hearing under Article XIII in response to the adverse recommendation of the Medical Executive Committee, then the Board [Directors] will review and finalize its determination consistent with the fair hearing and appellate process described in Article XIII.

**Comment:** Missouri law requires the Board of Trustees [Directors] to act upon the recommendations of the Medical Staff to approve or disapprove of appointments and determine privileges extended to each Member of the Medical Staff. Missouri law requires the terms of the Board's involvement in the credentialing process be incorporated in the Board's Bylaws. Care should be taken to ensure consistency between the two (2) documents.

- (g) Any report by an individual or group, including the report of the Board [Directors], required by any portion of this Section 6.7, must state the reasons for each recommendation or action taken, with specific reference to appropriate portions of the Application or other documentation. The reasons shall relate to, but not be limited to, standards of patient care, patient welfare, the objectives of the Hospital, or the conduct or competency of the Applicant. Any dissenting views at any point in the process also must be evidenced in writing, supported by reasons and references, and transmitted with the majority report.
- (h) The Board of Trustees [Directors] shall give notice of its final decision to the Applicant by Special Notice and to the Chief of Staff. The Chief of Staff shall, in turn, transmit the decision to the Department Director [Physician Liaison] of each Clinical Department [Service] concerned. A decision and notice to appoint shall include: the Medical Staff category to which the Applicant is appointed; the Clinical Department [Service] to which the Applicant is assigned; the Clinical Privileges the Applicant may exercise; the length of the provisional period and any special conditions attached to the Appointment.
- (i) With respect to initial Staff appointments, granting of Staff membership and Clinical Privileges shall be contingent upon review and ascertainment of adequate health status. Upon notification of such contingent appointment, the Applicant shall submit to the Medical Executive Committee the following information: any previous or current health problem or disability (including alcohol or drug

dependencies) that affects or that may be expected to progress within the next two (2) years to the point of affecting the Applicant's ability in terms of skill, attitude or judgment to perform professional and Medical Staff duties fully; hospitalizations or other institutionalizations for any such health problem or disability during the past ten (10) year period; if any such health problem or disability in the past is currently controlled by therapy or is resolved but may reoccur, date of last health examination with name and address of performing Physician and findings related to that problem or disability. Any such information shall be maintained in a separate file as a confidential medical record. The Medical Executive Committee will review such information to determine whether the Applicant's health status is such that the Applicant will be able to perform the procedures for which the Applicant has requested Privileges, with or without accommodation, according to accepted standards of professional performance and without posing a direct threat to patients. The Medical Executive Committee shall then make a recommendation to the Board as to whether the contingent offer of membership should be made final. Such recommendation and subsequent Board action shall be in accordance with Subsections 6.7(e) and (f) of this Article VI.

(j) All individuals and groups required to act on an Application under this Section 7 must do so in good faith and, except for good cause, complete their actions within the following time period:

Individual/Group

ilidividual/Oloup	<u>Time</u>
Credentials Committee (collection/verification)	sixty (60) days
Department Director [Physician Liaison]	thirty (30) days
Credentials Committee	thirty (30) days
Medical Executive Committee	Next regular meeting
Board of Trustees [Directors]	Next regular meeting

Time

These time periods are considered guidelines and do not create any rights for a Practitioner to have an Application processed within these precise periods; provided, however, that this provision shall not apply to the time periods contained in the provisions of Article XIII. When Article XIII is activated by an adverse recommendation or action as provided herein, the time requirements set forth therein shall govern the continued processing of the Application.

**Comment:** Under Missouri law, any Applicant for Medical Staff membership whose completed Application is not acted upon in ninety (90) calendar days of completion of verification of credentials data must be provided written notice of the reasons for such lack of action.

(k) An Applicant who has received a final adverse decision regarding, or who has voluntarily resigned or withdrawn an Application for, appointment, reappointment, Medical Staff category, Clinical Department [Service] assignment, or Clinical Privileges may not reapply to the Medical Staff or for the denied

Medical Staff category, Clinical Department [Service], or Clinical Privileges for a period of at least one (1) year. Any reapplication after the one (1)-year period will be processed as an initial Application, and the Applicant must submit such additional information as required by the Medical Staff or the Board [Directors] to show that any basis for the earlier adverse action has been resolved.

(l) If an Applicant's file remains incomplete ninety (90) days after the initial Application for membership, the Applicant will be deemed to have withdrawn the Applicant's Application for membership. The Chief Executive Officer shall notify that Applicant that the Applicant's Application is considered to have been withdrawn, and that the Applicant shall not be entitled to a hearing or any other procedural rights with respect to such Application.

**Comment:** This Section 6.7 of Article VI sets out a clear, structured process for processing Medical Staff Applications. It is based on notions of fundamental fairness. It is important that these processes are carefully adhered to by the Hospital to avoid or mitigate claims that an Applicant was not fairly treated. There may be variations on these provisions appropriate for different hospitals. If a hospital wants to develop a different process, it should do so only with advice of legal counsel to assure it is properly protected.

**Comment:** This Section 6.7 of Article VI clarifies that the Hospital Board [Directors] has [have] the final decision in the Medical Staff appointment process. This is required as a general statement of corporate law and as a specific requirement of the Missouri hospital licensing regulations.

## 6.8 Reappointment.

- (a) No later than three (3) months prior to the date of expiration of the Medical Staff Member's appointment, the Chief Executive Officer shall cause the Medical Staff Member to be notified of the upcoming expiration date. No later than sixty (60) days before the expiration date, the Staff Member must furnish to the Credentials Committee the following reappointment materials in writing and on a form approved by the Board of Trustees [Directors]:
  - (i) All information necessary to bring the Applicant's file current regarding the information required by Section 6.4 of Article VI, including all current licensure and Board [Directors] certification information, controlled substance registration, professional liability insurance coverage, the status of other institutional affiliations, pending or completed disciplinary actions, and health status changes;
  - (ii) A record of continuing medical and/or professional training and education completed outside of the Hospital during the preceding period;
  - (iii) Any requests for additional or reduced Clinical Privileges, with the basis for any requested changes; and

(iv) Any requests for Staff category or Department [Service] assignment changes, with the basis for the requested changes.

If a Staff Member without good cause fails to provide this information, the Staff Member will be deemed to have voluntarily resigned from the Staff and shall have the Staff Member's membership terminated automatically at the end of the current term unless the Staff Member requests, in writing, an extension and the extension is granted by the Medical Executive Committee for a period not to exceed forty-five (45) days. The Practitioner whose membership is so terminated is entitled to the procedural rights provided in Article XIII for the sole purpose of determining the issue of good cause.

The Credentials Committee shall verify the information provided on the reappointment Application and notify the Staff Member of any deficiencies, inadequacies or verification problems. The Staff Member then has the burden of producing adequate information and resolving any doubts about the data.

- (b) The Credentials Committee or its designee shall retain all relevant information regarding the Applicant's professional and collegial activity, performance, and conduct in the Hospital for inclusion in each Staff Member's credentials file. Such information shall include, but is not limited to, the following:
  - (i) Findings of quality assessment and utilization review activities demonstrating patterns of patient care and utilization;
  - (ii) Continuing education activities and participation in other internal training;
  - (iii) Clinical activity at the Hospital;
  - (iv) Previously successful or currently pending challenges to the Staff Member's licensure, sanctions imposed or pending, and other problems related to the Staff Member's practice or professional conduct;
  - (v) Health status, including any reasonable evidence of current health status that may be requested by the Medical Executive Committee (retained in a separate file as a confidential medical record);
  - (vi) Records of attendance at required Medical Staff and Hospital meetings;
  - (vii) Performance as a Staff officer, committee member or chairperson;
  - (viii) Participation in emergency room coverage;
  - (ix) Compliance with requirements related to the preparation of medical records (including requirements regarding timeliness, completeness and accuracy);
  - (x) Ability to work cooperatively with other Practitioners, Hospital personnel and the Board of Trustees [Directors];

- (xi) General character of relationship with patients and the Hospital;
- (xii) Ability to comply with all applicable Medical Staff Bylaws, Medical Staff Rules and Regulations, Hospital Bylaws, policies and procedures, and Hospital's Corporate Compliance Plan;
- (xiii) Voluntary or involuntary termination of Medical Staff membership or voluntary or involuntary limitation, reduction or loss of Clinical Privileges at another hospital;
- (xiv) Ability to practice in an efficient manner, taking into account the patients' medical needs, the facilities, services and resources available and generally recognized utilization standards as identified by the Utilization Review Committee; and
- (xv) Any other relevant information that could affect the Staff Member's status and Privileges at Hospital, including any activities of Staff Member at other hospitals and Staff Member's clinical practice outside the Hospital.

When correction or verification is accomplished, the Credentials Committee shall cause the reapplication form and supporting materials to be forwarded to the Department Director(s) [Physician Liaison(s)] of the Department(s) [Service(s)] in which the Staff Member requests Privileges.

The Department Director [Physician Liaison] (or designee) of each Department (c) [Service] in which a Staff Member has Privileges or has requested Privileges must evaluate the information contained in the Staff Member's file to assess the Applicant's continuing satisfaction of the qualifications contained in these Bylaws, and whether the requested Staff category, Clinical Department [Service] and Clinical Privileges are appropriate. Each applicable Department Director [Physician Liaison] (or designee) shall send to the Medical Executive Committee a written report regarding the Staff Member's professional performance, judgment, and clinical and/or technical skills. This evaluation should be based on the Staff Member's data from organization performance improvement activities and can include procedures performed and their outcomes and can be based on pertinent results of review of operative and other procedures, medication usage, blood usage, medical records mortality rates, utilization management, meeting and committee attendance and risk management data. Any such criteria should be uniformly applied and documented in the credential files. The written report shall include recommendations for, and any special limitations on, reappointment or non-reappointment and Staff Category, Department [Service] and Clinical Privileges.

**Comment:** Joint Commission Standards require that the evaluation be made based on Practitioner-specific data from organization performance improvement activities rather than on aggregate information. The Hospital should ensure that the types of information cited here as

considered for recredentialing are consistent with the Hospital's quality improvement program policies.

- (d) After receipt of the Department Director(s) [Physician Liaison(s)] report, the Credentials Committee is responsible for examining the Staff Member's file, the Department Director [Physician Liaison] reports and all other relevant information available to it, and shall forward a written report to the Medical Executive Committee that contains recommendations for reappointment or non-reappointment and Staff Category, Department [Service] assignment and Clinical Privileges, including any limitations or restrictions.
- (e) The Medical Executive Committee shall review the Staff Member's file, the Department Director(s) [Physician Liaison(s)] and Credentials Committee reports, and any other relevant information available to it, and shall defer action on the reappointment or prepare a written report with recommendations for, and any special limitations on, reappointment or non-reappointment and Staff category, Department [Service] assignment, Clinical Privileges and, if new Privileges or Staff category are requested, length of the Provisional Period.
- (f) The final determinations regarding reappointment Applications shall follow the process set forth in Subsections 6.7(e) through (h) of Article VI. For purposes of reappointment, the terms "Applicant" and "Appointment" as used in those Sections shall be read, as "Staff Member" and "Reappointment," respectively.
- (g) Notice provided to a Staff Member and the Staff Member's provision of updated information shall follow the procedure included in Subsection 6.8(a) of this Article VI. Thereafter, and except for good cause, any other party who is obligated to act under these Bylaws must forward reappointment reports and recommendations to the Medical Executive Committee, and all such reports and recommendations must be returned to the Board of Trustees [Directors] before the expiration of the Staff membership of the renewal Applicant.

The time periods addressed are guidelines for accomplishing the reapplication process. If this process has not been completed due to Hospital's delay by the end of the membership or appointment term, the Staff Member remains a current member and keeps the Staff Member's Clinical Privileges until the time that the process is complete, unless corrective action is taken. If the delay is due to the Staff Member's failure to provide information included in Subsection 6.8(a) of this Article VI, the Staff Member's Staff membership ends on the expiration date as provided in said Subsection 6.8(a) of Article VI unless explicitly extended as provided therein. Any extension of the process is not intended to (and shall not) create a right of automatic reappointment for the current term.

(h) The Staff Member may, either in connection with reappointment or at any other time, request modification of the Staff Member's Staff category, Department [Service] assignment or Clinical Privileges by submitting a written Application to

the Chief Executive Officer on the prescribed form. A modification Application is processed in the same manner as an Application for reappointment.

# 6.9 <u>Telemedicine Applications</u>.

- (a) Where a Practitioner is requesting admission or readmission as a Telemedicine Practitioner on the Medical Staff, the Practitioner may be credentialed in accordance with the other sections of this Article VI, or, at Hospital's discretion, Hospital may rely upon credentialing information (including any supporting documentation required by Hospital) supporting and evidencing the Practitioner's current active Medical Staff membership at the "distant-site hospital" consistent with this Section 6.9.
- (b) If the Hospital chooses to rely upon the credentialing information of another accredited hospital, the Hospital Board [Directors] and Medical Staff must ensure through a written agreement with the distant-site hospital that at least the following provisions have been met:
  - (i) The distant-site hospital's governing body assumes the responsibility for ensuring the requirements of the Medicare Conditions of Participation for oversight of Medical Staff are satisfied;
  - (ii) The distant-site hospital providing the telemedicine services is a Medicare-participating hospital and is a Joint Commission accredited facility;

**Comment:** The Joint Commission requires that the distant-site hospital be Joint Commission accredited. This requirement is only applicable to Joint Commission accredited hospitals.

- (iii) The individual distant-site Practitioner is privileged at the distant-site hospital providing the telemedicine services, and a current list of the distant-site Practitioner's Privileges are provided to Hospital;
- (iv) The individual distant-site Practitioner holds a license issued or recognized by the State of Missouri; and
- (v) For a distant-site Practitioner who holds current Privileges at the Hospital, the Hospital has evidence of an internal review of the distant-site Practitioner's performance of these Privileges and sends such performance information to distant-site hospital for use in the periodic appraisal of the distant-site Practitioner. At minimum, this information must include all adverse events (including sentinel events considered reviewable by The Joint Commission) that result from the telemedicine services provided by the distant-site Practitioner and all complaints the Hospital has received about the distant-site Practitioner.

- (c) If the distant-site entity is not a hospital facility, the Board and Medical Staff must ensure, through written agreement with the distant-site telemedicine entity, that all of the following provisions are met:
  - (i) The distant-site telemedicine entity is a contractor of services to the Hospital and must furnish contracted services in a manner that permits the Hospital to comply with all applicable Medicare Conditions of Participation for the contracted services;
  - (ii) The distant-site telemedicine entity's governing body assumes the responsibility for ensuring that the distant-site telemedicine entity's medical staff credentialing processes and standards at least meet Medicare standards for hospital governing body oversight of Medical Staff;
  - (iii) The distant-site telemedicine entity's Medical Staff credentialing processes and standards at least meet Medicare standards for eligibility and process for appointment to Medical Staff; and
  - (iv) All requirements in Subsection 7.11(a)(iii) through (v) above are satisfied.

**Comment:** The contractual requirements arise out of the Medicare Conditions of Participation and The Joint Commission standards to allow delegated credentialing for telemedicine.

- (d) When the Hospital has decided that the Medical Staff may rely on the decisions of a distant-site hospital or distant-site telemedicine entity, the Application process described in 6.3 is replaced by the application process of the distant-site hospital or distant-site telemedicine entity.
- (e) The Medical Executive Committee must make a recommendation to the Board [Directors] regarding the appointment or reappointment of each Telemedicine Practitioner and must document whether such recommendation is based on i) its own review of a Telemedicine Practitioner under Section 6.7(b)-(e) for appointments or Section 6.8(c)-(e) for reappointments or ii) the decisions of a Hospital approved distant-site hospital or distant-site telemedicine entity.

**Comment:** The Medicare survey guidance instructs surveyors to request documentation of the Medical Staff's recommendation for appointment or reappointment of Telemedicine Practitioners, which should include whether the Medical Staff engaged in its own review or relied on a distant-site hospital or distant-site telemedicine entity.

#### ARTICLE VII - THE DELINEATION OF CLINICAL PRIVILEGES

7.1 <u>Exercise of Privileges</u>. Medical Staff appointment or reappointment shall not confer any Clinical Privileges or right to practice at the Hospital. Each individual who has been given an appointment to the Medical Staff or who otherwise provides Clinical Services at the Hospital may only exercise the Clinical Privileges specifically granted by the Board

[Directors] or temporary Privileges granted in accordance with Section 7.10 of this Article VII. Regardless of the level of Privileges granted, each Practitioner must consult with other Practitioners as required by the Medical Staff Rules and Regulations, other policies of the Medical Staff, any Clinical Departments [Services], or the Hospital.

**Comment:** This Section 7.1 of Article VII establishes the important distinction between appointment or reappointment to the Medical Staff and the awarding of Privileges that describe the specifics of what the Medical Staff Member can do in treating patients in the Hospital. It is important in administering the Medical Staff Bylaws that Hospital administration, as well as Medical Staff leadership, use the terminology appropriately and accurately so as to avoid misunderstanding.

7.2 Basis for Privileges Determination. Clinical Privileges recommended to the Board [Directors] shall be based upon information submitted by Practitioner in accordance with these Bylaws and the criteria identified in Article III, Section 3.2. In determining the Clinical Privileges to be recommended, consideration must, at minimum, be given to: (1) challenges to any licensure or registration; (2) voluntary or involuntary relinquishment of a license or registration, Medical Staff membership, or limitation, reduction or loss of Clinical Privileges; (3) any evidence of an unusual pattern or excessive number of professional liability actions resulting in final judgment against the Applicant; 4) documentation of the Applicant's health status; (5) relevant Practitioner-specific data as compared to aggregate data, when available; and (6) morbidity and mortality data, when available. The following additional factors also may be used in determining Privileges: patient care needs in the area for the type of Privileges requested by the Applicant; the geographic location of the Practitioner; coverage available in the Practitioner's absence; and the adequacy of professional liability insurance. If necessary, review of patient records treated in other hospitals or practice settings may affect Privileges determinations. Privileges determinations for current Staff Members seeking reappointment or a change in Privileges must include observed clinical performance and documented results of Staff quality assessment and utilization review activities, including, but not limited to, the appropriateness of admission and length of stay, necessity of procedures, and indication for ancillary services.

**Comment:** The factors to be utilized in awarding Clinical Privileges contained in this Section 7.2 of Article VII are intended to be as broad as possible, yet still be related to providing quality patient care. The factors included in the "required" Section are consistent with Joint Commission Standards. There may be other factors to be included, but Section 7.2 of Article VII is clear in that the additional listed factors may not be all inclusive. If in doubt about a factor, it should be included. It must be kept in mind, however, that any such factor must be reasonably related to providing quality patient care, which is the mission of the Hospital. Use of any factors that may be interpreted as arbitrary or not related to quality patient care should be avoided.

**Comment:** The importance of this Section 7.2 of Article VII cannot be stressed too much. Hospitals are frequently faced with a Physician who wants to perform procedures that may go beyond the Physician's training, education and experience. Such a situation can cause substantial

tension between a Practitioner and the Hospital. This Section 7.2 of Article VII is the basis for the Hospital to exercise its authority in awarding Privileges in a fair, rational manner.

7.3 Department [Service] Responsibility in Defining Privileges. Each Department [Service] shall assist in defining and recommending in writing the Clinical Privileges for that Department [Service], including the operative, invasive and any special procedures, conditions and problems that fall within its clinical area, including the various levels of severity or complexity and different patient profiles when appropriate, and the requisite training, experience or other qualifications required to perform the procedures or otherwise exercise the Clinical Privileges. The recommendations must be coordinated by the Credentials Committee and approved by the Medical Executive Committee and Board [Directors], must be reviewed and revised on at least a(n) [semi] annual basis, and shall form the basis for delineating Privileges within the Department [Service]. When the delineation of Clinical Privileges is revised, by additions or deletions, all Staff Members holding Clinical Privileges, or adjust their practices to comply with any resulting reduction in Clinical Privileges.

**Comment:** The selection of an annual or semi-annual review of and the making of recommendations for defining Privileges in particular Departments [Services] are individual Hospital decisions. The decisions should be based on an evaluation on the whole of services provided by the Medical Staff and how quickly the various factors described in Section 7.3 of Article VII might change. Historical experience and current literature are the best sources for this information.

**Comment:** If the Hospital is a Joint Commission Primary Care Medical Home, the privileges also must specify which Practitioners or AHPs are permitted to serve in the role of a primary care provider.

7.4 Consultation and Other Conditions. Special requirements for consultation may be required of Practitioners as a condition to the performance of any or all Clinical Privileges, in addition to the consultation requirements in the Bylaws or in the Rules, Regulations and Policies of the Staff, any Department [Service], or Hospital. Each Practitioner requesting Clinical Privileges agrees that in dealing with cases outside the scope of the Practitioner's training or in an unusual area of practice, the Practitioner will seek appropriate consultation with a Practitioner who has expertise in such cases and, if necessary, refer such case to the Practitioner.

**Comment:** The intent of this Section 7.4 of Article VII is obvious. However, it is important because it creates clear authority of the Medical Staff and Hospital to curtail or limit what a Physician is authorized to do in providing patient care.

7.5 Requests. An Application for appointment and reappointment to the Medical Staff must contain a written request for all Clinical Privileges sought by the Applicant or Staff Member. Requests for temporary Privileges and modifications of Privileges in the interim between reappointment also must be submitted in accordance with the procedures contained in these Bylaws.

- 7.6 <u>Procedure</u>. Requests for Clinical Privileges shall be processed in accordance with the procedures outlined in Article VI of these Bylaws, as applicable. Temporary Privileges requests shall be processed according to Section 7.12 of this Article VII.
- 7.7 Special Conditions for Dentists. Clinical Privileges requests received from Dentists shall be reviewed in accordance with the procedures contained in this Article VII. Surgical procedures shall be reviewed and supervised by the Surgery Department Director [Physician Liaison]. Patients of Dentists must receive a basic medical evaluation by a Physician Member of the Medical Staff. A Physician Member of the Medical Staff also shall be responsible for the medical care of any patient during the patient's hospitalization and will advise on the risk and effect of any procedure on the patient's total health status. The Physician consultant and the Dentist must agree on the performance of any surgical procedure if a significant medical abnormality is present. The Surgery Department Director [Physician Liaison] will decide the issue in case of dispute. The Dentist is responsible for the dental history and physical and all appropriate elements of the patient's record. A Dentist specialist may write orders within the scope of the Dentist specialist's license as limited by law and as consistent with the Medical Staff Rules and Regulations. The Dentist specialist shall agree to comply with all applicable Medical Staff Bylaws, Medical Staff Rules and Regulations, Hospital Bylaws, and policies and procedures.
- 7.8 Special Conditions for Psychologists. Clinical Privileges requests received from Psychologists shall be reviewed in accordance with the procedures contained in this Article VII. Patients of Psychologists must receive a basic medical evaluation by a Physician Member of the Medical Staff. A Physician Member of the Medical Staff also shall be responsible for the medical care of any patient during the patient's hospitalization and will advise on the risk and effect of any procedure on the patient's total health status. The Psychologist is responsible for the psychological history and physical and all appropriate elements of the patient's record. A Psychologist may write orders within the scope of the Psychologist's license as limited by law and as consistent with the Medical Staff Rules and Regulations. The Psychologist shall agree to comply with all applicable Medical Staff Bylaws, Medical Staff Rules and Regulations, Hospital Bylaws, and policies and procedures.
- 7.9 Special Conditions for Podiatrists. Clinical Privileges requests received from Podiatrists shall be reviewed in accordance with the procedures contained in this Article VII. Surgical procedures shall be reviewed and supervised by the Surgery Department Director [Physician Liaison]. Patients of Podiatrists must receive a basic medical evaluation by a Physician Member of the Medical Staff. A Physician Member of the Medical Staff also shall be responsible for the medical care of any patient during the patient's hospitalization and will advise on the risk and effect of any procedure on the patient's total health status. The Physician consultant and the Podiatrist must agree on the performance of any surgical procedure if a significant medical abnormality is present. The Surgery Department Director [Physician Liaison] will decide the issue in case of dispute. The Podiatrist is responsible for the podiatric history and physical and all appropriate elements of the patient's record. A Podiatrist may write orders within the

scope of the Podiatrist's license as limited by law and as consistent with the Medical Staff Rules and Regulations. The Podiatrist shall agree to comply with all applicable Medical Staff Bylaws, Medical Staff Rules and Regulations, Hospital Bylaws, and policies and procedures.

- 7.10 Special Conditions for AHPs. Requests for Clinical Privileges by AHPs shall be processed in the manner specified in Article V, as applicable. An AHP may, subject to any licensure requirements or other limitations, exercise independent judgment within the areas of the AHP's professional competence and participate directly in the medical management of patients under the supervision of a Practitioner who has been accorded Privileges to provide such care. Surgical procedures performed by an AHP, if any, are under the overall supervision of the Surgery Department Director [Physician Liaison]. A Physician Member of the Medical Staff must perform a basic medical appraisal for each AHP patient, be responsible for the care of any medical problem that may be present at admission or that may arise during hospitalization, and determine the risk and effect of any proposed surgical or special procedure on the total health status of the patient.
- 7.11 Emergency Privileges. In case of an emergency as defined in this Section 7.11 of Article VII, any Medical Staff Member is authorized and shall be assisted to render medical treatment to attempt to save the patient's life or to save the patient from serious harm, as allowed within the Staff Member's scope of practice, and notwithstanding the Medical Staff Member's Department [Service] affiliation, Staff category or level of Clinical Privileges. A Practitioner exercising emergency Privileges must obtain all consultative assistance deemed necessary and arrange for appropriate post-emergency care. For purposes of this Section 7.11 of Article VII, "emergency" is defined as a situation where serious permanent harm is imminent or in which the life of a patient is in immediate danger and delay in administering treatment could increase the danger to the patient.

**Comment:** Hospitals should also consider a category for disaster Privileges. A disaster is an emergency that, due to its complexity, scope or duration, threatens the organization's capabilities and requires outside assistance to sustain patient care, safety or security functions. Under Joint Commission Standards, a hospital may grant disaster Privileges to volunteer licensed independent Practitioners when its emergency operations plan has been activated. Medical Staff Bylaws must identify those individuals responsible for granting disaster Privileges to volunteer licensed independent Practitioners.

### 7.12 Temporary Privileges.

- (a) Temporary Privileges may be granted as described in Subsection 7.12(b) of this Article VII, and are available only to a Practitioner, when documentation submitted substantially supports that the requesting Practitioner has the requisite qualifications, ability and judgment to perform the Privileges requested, and only after the Practitioner shows that the professional liability insurance requirements of these Bylaws have been met. Temporary Privileges may be granted as described in Subsection 7.12(b) of this Article VII only in the following events:
  - (i) To fulfill an important patient care need; or

(ii) When an Applicant with a complete, clean Application is awaiting review and approval of the Medical Staff Executive Committee and the Governing Body.

Comment: Section 7.12 of Article VII deals with the issue of temporary Privileges and describes the situations in which the Hospital will grant such Privileges. Although Medical Staff Bylaws often routinely provide for temporary Privileges for Applicants to the Medical Staff, the Model Medical Staff Bylaws (in Subsection 7.12(b)(i) of this Article VII) create a presumption that the temporary Privileges shall not be granted to Medical Staff Applicants. This Section 7.12 of Article VII allows for exceptions under circumstances necessary for patient care otherwise in the best interests of the Hospital as determined by the Chief of Staff and the applicable Department Director [Physician Liaison]. The rationale for this presumption against granting of temporary Privileges to Applicants is that the purpose of the Application process is to determine the qualifications of the Physician for treating patients in the Hospital. Permitting the Physician to practice in the Hospital prior to final determination regarding such qualifications could place the Hospital at a high degree of risk that should not be permitted absent special circumstances.

The Chief of Staff may impose a requirement that the relevant Department Director [Physician Liaison] or the Department Director's [Physician Liaison's] designee shall consult and report regarding the Practitioner's activity in the Hospital during the temporary Privileges period. The Practitioner must agree in writing to comply with these Medical Staff Bylaws, the Medical Staff Rules and Regulations, Hospital Bylaws, Hospital Corporate Compliance Plan, and Hospital policies and procedures in matters relating to the Practitioner's activities in the Hospital.

- (b) After reviewing the favorable recommendation of the Chief of Staff, the Chief Executive Officer may grant temporary Privileges under the following circumstances:
  - (i) Applicants: Temporary Privileges shall <u>not</u> be granted to Medical Staff Applicants except under circumstances as the Chief of Staff determines are necessary for patient care or otherwise in the best interest of the Hospital. Upon recommendation of the Chief of Staff, the Chief Executive Officer or the Chief Executive Officer's designee may grant temporary Privileges to an Applicant for Medical Staff membership only after receipt of the following:
    - a. Written verification of current licensure.
    - b. Written verification of relevant training or experience.
    - c. Written verification of current competence.
    - d. Written verification of the ability to perform the Privileges requested.

- e. Written verification of the results of the National Practitioner Data Bank query.
- f. The Applicant has a complete Application, and the Applicant is waiting for review and recommendation by the Medical Staff Executive Committee and approval by the Governing Body.
- g. The Applicant has no current challenge or previously successful challenge to licensure or registration.
- h. The Applicant has not been subject to involuntary termination of Medical Staff membership at another organization.
- i. The Applicant has not been subject to involuntary limitation, reduction or denial of Clinical Privileges.

Temporary Privileges to Applicants may be awarded for an initial period of up to [\_\_\_\_\_\_(\_\_)] days with subsequent renewals not to go beyond a total of one hundred twenty (120) days. Any such renewal of temporary Privileges can be made only upon the written recommendation of the Chief of Staff, applicable Department Director [Physician Liaison] and the concurrence of the Chief Executive Officer, and is contingent upon the receipt of information that continues to support the granting of the Practitioner's Application for membership and Privileges. Temporary Privileges cannot be extended if the Application is still pending due to the Applicant's failure to respond in a satisfactory manner to a request for more information or clarification.

**Comment:** The Hospital should set the time frame or duration for the initial period of temporary Privileges. Joint Commission requirements limit the total period for temporary Privileges to one hundred twenty (120) days. A study of historical needs of the Hospital should provide a guide. On the other hand, caution should be exercised to avoid the granting of temporary Privileges becoming a means to circumvent the normal process for granting regular Privileges.

#### (ii) Patient Care Need:

- a. Locum tenens. Upon receipt of a written request, temporary Privileges may be granted to a Practitioner who is providing coverage for a member of the Medical Staff. Locum tenens Privileges may be granted initially for a maximum period of sixty (60) days and may be renewed for two (2) consecutive periods of thirty (30) days.
- b. Care of Specific Patients: Upon receipt of a written request, temporary Privileges may be granted to a Practitioner who possesses skills necessary to provide care to a patient that no other Practitioner currently holding Privileges at the Hospital possesses.

Such Privileges may not be granted in more than four (4) instances in any twelve (12)-month period, after which the Practitioner must apply for Staff membership.

- c. Verification Required: *Locum tenens* Privileges or Privileges for care of a specific patient may be granted only if the Hospital receives the following:
  - A completed Application for appointment to the position sought, including a delineation of the specific Privileges requested;
  - A copy of or verbal confirmation from the appropriate authority of adequate licensure;
  - DEA/controlled substance registration;
  - Evidence of sufficient professional liability insurance;
  - Verification that the Practitioner is not on the U.S.

    Department of Health and Human Services Office of
    Inspector General Exclusion list or any other exclusion list
    affecting the Practitioner's ability to participate in state
    health care programs;
  - Written or verbal reference affirming current competence from a Medical Staff authority or appropriate administrative personnel at the Practitioner's current principal hospital affiliation; and
  - The recommendation of either the applicable Department Director [Physician Liaison] or Chief of Staff.
- (c) Temporary Privileges shall not be routinely used for other administrative purposes, including, but not limited to, the licensed Practitioner's failure to provide all information necessary to process the Practitioner's appointment or reappointment in a timely manner and/or the failure of the Medical Staff to verify performance data and information in a timely fashion.
- (d) Upon receipt of information that raises doubts or concerns about a Practitioner's qualifications or specific temporary Privileges granted, and after discussion with Practitioner's supervising Department Director [Physician Liaison], the Chief Executive Officer or the Chief of Staff must determine whether to suspend or terminate a portion or all of a Practitioner's temporary Privileges. If the life or wellbeing of a patient is thought to be endangered, any person entitled to impose summary suspensions under these Bylaws may affect the termination. In the event of any such termination, the Practitioner's patients then in the Hospital shall

- be assigned to another Practitioner by the Department Director [Physician Liaison] responsible for supervision of the Practitioner. The wishes of the patient shall be considered, where feasible, in choosing a substitute Practitioner.
- (e) The granting of temporary Privileges is a courtesy on the part of the Hospital. Neither the granting, denial, suspension nor the revocation of such Privileges shall entitle the Practitioner concerned to any of the procedural rights provided in these Bylaws with respect to hearings or appeal.

#### 7.13 Telemedicine.

- (a) The Clinical Privileges at Hospital related to the provision of telemedicine will include both the types of services that may be provided through telemedicine and the types of Practitioners and AHPs that are permitted to request Privileges to perform those services.
- (b) At the time of granting, renewal or revision to a Practitioner or AHP's Privileges to perform telemedicine services, whether to Hospital's patients or from Hospital to the patients of another entity, the Medical Staff will evaluate performance of those services as a part of the Privileging process and determine:
  - (i) The Clinical Services performed are appropriately delivered by Practitioners or AHPs through this medium;
  - (ii) The Clinical Services offered are consistent with commonly accepted quality standards; and
  - (iii) If the Practitioner is a Telemedicine Practitioner on the Medical Staff, whether the services are within the scope the Telemedicine Practitioner is to perform for the Hospital under the applicable contract.
- (c) If a Practitioner or AHP desires to provide services via telemedicine, the request for Clinical Privileges must specify telemedicine. Receipt of Clinical Privileges to provide a specific service inperson does not automatically allow the Practitioner or AHP to provide the same service through telemedicine.

#### **ARTICLE VIII - OFFICERS**

- 8.1 <u>Officers of the Medical Staff</u>. The officers of the Staff shall consist of a Chief of Staff, a Vice Chief of Staff and a Secretary-Treasurer.
- 8.2 Qualifications of Officers. Each officer shall:
  - (a) Be a member in Good Standing of the Active Medical Staff at the time of nomination and election, and remain in Good Standing throughout the member's term of office. Any officer who fails to maintain such status shall immediately be removed from office.

- (b) Have been recognized for a high level of clinical competence in the member's field and have demonstrated executive and administrative ability through active participation in Staff activities and other experience.
- (c) Have demonstrated a high level of interest in and support of the Medical Staff and Hospital by the member's Staff tenure and the member's level of Clinical activity at Hospital.
- (d) Willingly and faithfully exercise the duties and authority of the office held and cooperate and work with the other officers, Department Directors [Physician Liaisons] of the Staff, the Chief Executive Officer, the Board of Trustees [Directors], and their respective committees.

#### 8.3 Election of Officers.

- (a) Officers shall be elected, subject to Board [Directors] approval, by a majority vote of those Active Staff Members in Good Standing who are present at the Medical Staff's annual meeting or by mail or electronic ballot. The decision to use mail or electronic ballot shall be at the discretion of the Medical Executive Committee. If the elections are held by mail or electronic ballot, officers shall be elected by a majority vote of all Active Staff Members' ballots received within such time period as is established by the Medical Executive Committee. Officers so elected and approved by the Board shall take office on the January 1 following said meeting. If there are three (3) or more candidates for any office, and no candidate receives a majority of the votes cast, the name of the candidate who receives the fewest votes will be omitted from successive ballots until a majority vote is obtained by one (1) candidate.
- (b) At the discretion of the Medical Executive Committee, a nominating committee may be appointed. Such committee, if appointed, shall consist of three (3) members, and shall include the immediate past and current Chiefs of Staff and one (1) other Active Staff Member appointed by the Chief of Staff. This committee shall offer one (1) or more nominees for each office. The voting members of the Medical Staff shall be notified of the nominations in such manner as is determined by the Medical Executive Committee at least \_\_\_\_\_ (\_\_\_) calendar days prior to elections.
- (c) Nominations also may be made from the floor at the time of election.
- 8.4 <u>Term of Office</u>. Officers shall serve a one (1)-year term, but may be re-elected for two (2) additional terms up to a maximum of three (3) consecutive full-year terms. Each officer shall serve until the end of the officer's term and until a successor is elected, unless the officer sooner resigns or is removed from office.

**Comment:** This provision on term of office, which allows for an officer to serve for up to three (3) consecutive terms, is reasonable. The purpose of limiting the number of successive terms is to provide for continuity on one hand, and "new blood" on the other. Each hospital must determine

the most appropriate balance of those factors in establishing its own limits on the number of successive terms.

- 8.5 <u>Vacancies in Office</u>. If there is a vacancy in the office of the Chief of Staff, the Vice Chief of Staff shall serve out the remaining term. Vacancies in the other offices shall be filled by appointment by the Medical Executive Committee subject to Board [Directors] approval.
- 8.6 Removal of Officers. Grounds for removal of any officer of the Medical Staff shall include, but are not limited to, (i) conduct that is detrimental to or reflects adversely on the Medical Staff or the Hospital, (ii) inability or failure to perform the necessary functions of the office held, or (iii) any action of conduct that would form the basis for corrective action pursuant to Article XIII, even if corrective action is not taken. Removal of a Medical Staff officer may be initiated by the Board [Directors], the Medical Executive Committee or by a petition signed by at least one-third (1/3) of the members of the Active Medical Staff. Removal shall be considered at a special meeting called for that purpose. Removal shall require a two-thirds (2/3) vote of the Active Medical Staff or a majority vote by the Board [Directors]. Voting may take place at any regular or special meeting of the Medical Staff or the Board [Directors] or by mail or electronic ballots, as determined by the Medical Executive Committee or the Board [Directors]. Written notice of any meeting at which removal of an officer or voting by mail or electronic ballots are to be considered, shall be delivered to all Medical Staff Members or Board Members [Directors] entitled to vote at least \_\_\_\_\_ (\_\_\_) calendar days before the date of the voting.

#### 8.7 Duties of Officers.

- (a) The Chief of Staff shall:
  - (i) Serve as the Chief Administrative Officer of the Medical Staff;
  - (ii) Act in coordination and cooperation with the Chief Executive Officer in all matters of mutual concern within the Hospital;
  - (iii) Aid in coordinating the activities and concerns of the Hospital administration and of the nursing and other patient care services with those of the Medical Staff;
  - (iv) Call, preside at and oversee the preparation of the agenda for all general meetings of the Medical Staff;
  - (v) Serve as chairperson of the Medical Executive Committee;
  - (vi) Serve as ex-officio member without vote of all other Medical Staff committees:

- (vii) Be responsible for the following as they relate to the Medical Staff: enforcement of the Medical Staff Bylaws, Medical Staff Rules and Regulations, the Hospital Bylaws, policies and procedures, and Hospital Corporate Compliance Plan; implementation of sanctions where indicated; and compliance with appropriate procedure as set forth in these Bylaws in all instances where corrective action has been requested against a Practitioner;
- (viii) Except with respect to the Medical Executive Committee, appoint committee members to all standing, special and disciplinary Medical Staff committees;
- (ix) Meet with the Board of Trustees [Directors] and the Chief Executive Officer at least annually to report the views, needs, policies and grievances of the Medical Staff;
- (x) Receive and interpret the policies of the Board of Trustees [Directors] to the Medical Staff, and report to the Board of Trustees [Directors] on the performance and maintenance of quality with respect to the Medical Staff's delegated responsibility to provide medical care;
- (xi) Be the spokesperson for the Medical Staff in its professional and public relations;
- (xii) Be responsible for the educational activities of the Medical Staff;
- (xiii) Direct the development, implementation, and day-to-day functioning and organization of the Medical Staff components of the quality review, risk management and utilization management programs; oversee that the programs are clinically and professionally sound in accomplishing program objectives and are in compliance with regulatory and accrediting agency requirements; and report to the Board regarding such programs and activities;
- (xiv) Perform such other duties, and exercise such authority commensurate with the office as are set forth in the Medical Staff Bylaws and other Hospital and Medical Staff Rules and Policies, and as may, from time to time, be reasonably requested by the Medical Executive Committee, the Board [Directors] or Chief Executive Officer; and
- (xv) Attend meetings of the Board [Directors].

**Comment:** The requirement that the Chief of Staff attend Hospital Board meetings does not require that the Chief of Staff be a Board [Directors] Member. Such status would be a function of the Hospital Bylaws. The purpose of this requirement is to provide for the necessary liaison between the Medical Staff (to which clinical functions have been delegated) and the governance and management of the Hospital.

## (b) The Vice Chief of Staff shall:

Perform all the duties and assume all the responsibilities of the Chief of Staff in the Chief of Staff's absence be a member of the Medical Executive Committee and succeed the Chief of Staff when the Chief of Staff fails to serve for any reason.

### (c) The Secretary-Treasurer shall:

Call Medical Staff meetings at the request of the Chief of Staff, keep minutes of all Medical Staff meetings and Medical Executive Committee meetings, respond to all correspondence, and perform such other duties as ordinarily pertain to the office of Secretary. The Secretary-Treasurer shall keep and maintain, or cause to be kept and maintained, adequate and correct accounts of the funds, if any, and business transactions of the Medical Staff and perform such other duties as ordinarily pertain to the office of Treasurer. The Secretary-Treasurer shall be a member of the Medical Executive Committee of the Medical Staff.

**Comment:** This Article VIII presents a formal approach to Medical Staff leadership selection and duties. The language indicates the importance of Medical Staff leadership. The Model Provisions may be difficult to apply to small hospital situations with small Medical Staffs. The language can be adjusted to fit the situation; however, the spirit of this Article VIII should be retained.

#### **ARTICLE IX - CLINICAL DEPARTMENTS**

Comment: This version of Article IX provides for a Departmental Staff, including Departmental Directors (as opposed to Physician Liaisons in a non-departmental setting). The language describing the duties of a Department Director is stronger than that for Physician Liaisons. For example, a Physician Liaison "oversees" professional and administrative activities of a given service, while a Department Director is "responsible" for Clinical and administrative activities of the Department Director's Department. There are a number of other specific responsibilities of the Department Director described in this Article IX that do not appear in the list of responsibilities for Physician Liaisons found in the non-departmental model.

9.1	Current Departments, Affiliation	n. The current Clinical Departments are [(1) <b>Family</b>
	Practice/Medicine (including _	), (2) <b>Surgery</b> , (3)
	Pathology (including	), (4) <b>Radiology</b> (including but not
	limited to	and (5) <b>Internal Medicine</b> (including
	)].	Every Staff Member must have a primary affiliation with
	the Department which most clear	arly reflects the Staff Member's professional training and
	experience in the Clinical area in	n which the Staff Member's practice is concentrated. A
	Practitioner may be granted Clir	nical Privileges in one (1) or more Departments, and the
	Staff Member's exercise of Privi	leges within the jurisdiction of any Department is always

subject to the Rules and Regulations of that Department and the authority of the Department Director.

**Comment:** Section 9.1 of Article IX provides for listing and describing the Departments available in the Hospital. The bolded Departments are provided as examples. Each hospital will list those it offers and will decide what is to be included in each Department.

# 9.2 <u>Department Director; Election Qualifications and Appointment.</u>

- (a) Each Department shall have one (1) or more Department Directors who must be members of the Active Medical Staff and of the applicable Department, remain in Good Standing throughout the Department Director's term, and must be willing and able to faithfully discharge the functions of the Department Director's office. The Department Directors shall be board certified by an appropriate specialty board or affirmatively demonstrate, through the Privilege delineation process, competence in the appropriate area of practice.
- (b) The Chief of Staff, subject to Board [Directors] approval, shall appoint each Department Director. The Department Director will serve a one (1)-year term commencing upon appointment and continuing until the Department Director's successor is chosen, unless the Department Director sooner resigns or is removed from office. The Department Director may be eligible for reappointment.

# 9.3 <u>Department Director; Responsibilities and Authority</u>. Each Department Director shall:

- (a) Be responsible for all clinically-related activities and (unless otherwise provided for by the Hospital) all administratively-related activities of the Department and report on such activities as requested by the Chief Executive Officer, the Medical Executive Committee or the Board of Trustees [Directors];
- (b) Be responsible for continuing surveillance of the professional performance of all individuals in the Department who have delineated Clinical Privileges, including but not limited to monitoring adherence to Staff, Hospital and Department policies and procedures for obtaining consultation, alternate coverage, unexpected patient care management events, patient safety and adherence to sound principles of Clinical practice generally;
- (c) Recommend to the Medical Staff the criteria for Clinical Privileges that are relevant to the care provided in the Department;
- (d) Be responsible for recommending Clinical Privileges for each member of the Department;
- (e) Assess and recommend to the relevant Hospital authority off-site sources for needed patient care services not provided by the Department or the Hospital;

- (f) Be responsible for the integration of the Department into the primary functions of the Hospital;
- (g) Be responsible for coordination and integration of inter-departmental and intradepartmental services, including coordination of personnel, supplies, special regulations, standing orders and techniques;
- (h) Develop (as necessary) and implement policies and procedures that guide and support the provision of services in the Department;
- (i) Make recommendations for a sufficient number of qualified and competent persons to provide care in the Department;
- (j) Determine the qualifications and competence of Department personnel who are not Practitioners or AHPs and who provide patient care services in the Department;
- (k) Provide for continuous assessment and improvement of the quality of care and services provided in the Department;
- (l) Maintain quality control programs, as appropriate, in the Department;
- (m) Provide for orientation and continuing education of all persons in the Department; and
- (n) Make recommendations for space and other resources needed by the Department.
- 9.4 <u>Department Functions</u>. Each Department shall perform the following functions:
  - (a) Review and make recommendations regarding criteria for the granting of Clinical Privileges in the Department consistent with the policies of the Medical Staff and the Board of Trustees [Directors];
  - (b) Cooperate with the Performance Improvement/Quality Assurance Committee and the Utilization Review Committee in their retrospective review of completed records of discharged patients and others for the purposes of contributing to continuing education and to the process of developing criteria to help assure quality patient care and efficient and effective usage of health care services;
  - (c) Meet, as necessary, to review and analyze the clinical work of the Department;
  - (d) Receive and relay reports regarding performance improvement/quality assurance issues in the applicable Hospital Department for referral to the Performance Improvement/Quality Assurance Committee;
  - (e) Develop and update, as necessary, policies and procedures for the operation of the Department and coordinate such policies and procedures with those of associated Departments;

- (f) Provide input to the relevant Hospital Department's organizational plan to define the Department's role within the particular Hospital Department in which it operates; and
- (g) Be responsible for making available to its members scientific or other educational programs as deemed necessary by the Medical Executive Committee.
- 9.5 <u>Assignment</u>. The Medical Executive Committee shall, after consideration of the recommendation of the Credentials Committee, recommend Department assignments for all Medical Staff Members and for all approved AHPs with Clinical Privileges in accordance with the guidelines of Section 9.1 of this Article IX.
- 9.6 Resignation and Removal. A Department Director may resign at any time by giving written notice to the Medical Executive Committee. Such resignation shall take effect on the date of receipt or at any later time as specified in the notice. Removal may be effected by the Board of Trustees [Directors] acting upon its own initiative; by a two-thirds (2/3) majority vote of the Medical Executive Committee and subject to approval of the Board; or by a two-thirds (2/3) majority vote of the applicable constituent group of the respective Department and subject to approval of the Board [Directors]. An unexpected vacancy will be filled by the Medical Executive Committee through appointment of an acting officer subject to Board [Directors] approval.

## ARTICLE IX - CLINICAL SERVICES [ALTERNATE]

Comment: This version of the Model Medical Staff Bylaws provides for a non-departmental Staff and uses "Services" to perform many of the functions that would be performed by Departments under a departmental model. Similarly, the Model Bylaws provide for "Physician Liaisons" who perform many of the functions that would be performed by Department heads under the departmental model. This model attempts to minimize and streamline the responsibilities and obligations of the Services and the Physician Liaison to discharge efficiently the administrative burden of small hospital medical staffs. Nevertheless, these provisions may need additional modification to suit medical staffs with only a few members. However, the notion of Services is important, and the spirit of this provision should be retained in some form.

Non-Departmental Staff. The Medical Staff shall constitute a unified Staff, without
clinical Departments. The Medical Executive Committee will periodically reexamine the
structure and recommend to the Board [Directors] desirable or necessary actions in
creating a new service or in establishing or reorganizing the Staff in order to promote
improved efficiency and patient care.

9.2	<u>Current Services</u> ; <u>Affiliation</u> . The current Clinical Services are [(1) <b>Family</b>
	Practice/Medicine (including), (2) Surgery (including), (3)
	Pathology (including), (4) Radiology (including but not limited to
	) and (5) <b>Internal Medicine</b> (including)]. Every Staff Member
	must have a primary affiliation with the Service which most clearly reflects the Staff
	Member's professional training and experience in the Clinical area in which the Staff
	Member's practice is concentrated. A Practitioner may be granted Clinical Privileges in

one (1) or more Services, and the Practitioner's exercise of Privileges within the jurisdiction of any Service is always subject to the Rules and Regulations of that Service and the authority of the Physician Liaison.

**Comment:** Section 9.2 of Article IX provides for listing and describing the Services available in the Hospital. The bolded Services are provided as examples. Each Hospital will list those it offers and will decide what is to be included in each service.

## 9.3 <u>Physician Liaison; Election Qualifications and Appointment.</u>

- (a) Each Service shall have at least one (1) Physician Liaison who must be a member of the Active Medical Staff and of the applicable Service and remain in Good Standing throughout the Physician Liaison's term and must be willing and able to faithfully discharge the functions of the Physician Liaison's office. The Physician Liaison shall demonstrate, through the Privilege delineation process, competence in the appropriate area of practice.
- (b) The Chief of Staff, subject to Board approval, shall appoint the Physician Liaisons. Each Physician Liaison will serve a one (1)-year term commencing upon appointment and continuing until the Physician Liaison's successor is chosen, unless the Physician Liaison sooner resigns or is removed from office. The Physician Liaison may be eligible for reappointment.

# 9.4 <u>Physician Liaison; Responsibilities and Authority</u>. Each Physician Liaison shall:

- (a) Oversee professional and administrative activities within the Physician Liaison's Service and report on such activities as requested by the Medical Executive Committee, the Chief Executive Officer or the Board of Trustees [Directors];
- (b) Upon request, transmit to the Credentials Committee the Physician Liaison's recommendations concerning appointment or reappointment, Staff category, and delineation of Clinical Privileges of all Practitioners in the Physician Liaison's Service:
- (c) Assist in monitoring, on a continuing and concurrent basis, the performance of those with Clinical Privileges in the Service for adherence to Staff, Hospital and Service policies and procedures, including, but not limited to, requirements for alternate coverage and for obtaining consultation, for adherence to sound principles of Clinical practice generally, for unexpected patient care management events, and for patient safety;
- (d) Coordinate with Hospital Services and Hospital administration in matters affecting patient care in the Physician Liaison's Service, including personnel, supplies, special regulations, standing orders and techniques; and
- (e) Perform such other duties as may be determined by the Medical Executive Committee.

- 9.5 <u>Service Functions</u>. Each Service shall perform the following functions:
  - (a) Upon request, review and make recommendations regarding criteria for the granting of Clinical Privileges in the Service consistent with the policies of the Medical Staff and the Board of Trustees [Directors];
  - (b) Cooperate with the Performance Improvement/Quality Assurance Committee and the Utilization Review Committee in their retrospective review of completed records of discharged patients and others for the purposes of contributing to continuing education and to the process of developing criteria to help assure quality patient care and efficient and effective usage of health care services;
  - (c) Meet, as necessary, to review and analyze the clinical work of the Service;
  - (d) Receive and relay reports regarding performance improvement/quality assurance issues in the applicable Hospital Department for referral to the Performance Improvement/Quality Assurance Committee;
  - (e) Provide input to the relevant Hospital Department's organizational plan to define the Service's role within the particular Hospital Department in which it operates; and
  - (f) Be responsible for making available to the Service Members scientific or other educational programs as deemed necessary by the Medical Executive Committee.
- 9.6 <u>Assignment</u>. The Medical Executive Committee shall, after consideration of the recommendation of the Credentials Committee, recommend Service assignments for all Medical Staff Members and for all approved AHPs with Clinical Privileges in accordance with the guidelines of Section 9.2 of this Article IX.
- 9.7 <u>Resignation and Removal</u>. A Physician Liaison may resign at any time by giving written notice to the Medical Executive Committee. Such resignation shall take effect on the date of receipt or at any later time as specified in the notice. Removal may be effected by the Board of Trustees [Directors] acting upon its own initiative, by a two-thirds (2/3) majority vote of the Medical Executive Committee and subject to approval of the Board [Directors], or by a two-thirds (2/3) majority vote of the applicable constituent group of the respective Service and subject to approval of the Board [Directors]. An unexpected vacancy will be filled by the Medical Executive Committee through appointment of an acting officer subject to Board [Directors] approval.

#### **ARTICLE X - COMMITTEES**

**Comment:** The purpose of designating Peer Review Committees in the sweeping fashion described in this Article X is to take advantage of the Privilege afforded by Missouri statute (the peer review statute) prohibiting the discovery of peer review materials and information and providing immunity from suit to peer review participants. However, these privileges and immunities attach to the review of patient care in a narrow sense. The courts will look to the

function of the committee, not just the name of the committee, in determining whether the Privileges and immunities are applicable. Further, Missouri law defines a "Peer Review Committee" to be a committee of "health care professionals," which includes Missouri licensed Physicians, Dentists, Podiatrists, Optometrists, Pharmacists, Chiropractors, Psychologists, Nurses, Social Workers, Professional Counselors, or Mental Health Professionals. In order to follow the statutory definition of "Peer Review Committee," hospitals may wish to consider provisions that allow a non-health care professional to attend a Peer Review Committee meeting without designating that non-health care professional as a "member" of the committee.

**Comment:** Hospitals with small medical staffs may face a challenge in maintaining all of the separate committees described in these Model Bylaws. In many cases, committee functions can be combined to fit the situation. In situations with very small medical staffs, the entire staff could act as a committee and carry out all of the essential functions. The Bylaws should be adjusted to reflect these realities. As the need for additional committees arises, the Medical Staff Bylaws can be amended.

#### 10.1 Peer Review Committees.

- (a) The Medical Staff as a whole and each committee provided for by these Bylaws is hereby designated as a Peer Review Committee in accordance with 537.035 RSMo, as it may be amended. Such committees shall be responsible for evaluating, maintaining and/or monitoring the quality and utilization of health care Services. Each committee shall have a Representative from the Medical Records Department acting as support Staff for the committee.
- (b) In carrying out the Staff Member's duties under these Bylaws, whether as a committee member, Department Director [Physician Liaison], Staff officer or otherwise, each Staff Member shall be acting in the Staff Member's capacity as a peer review officer.
- (c) Such Peer Review Committees may, from time to time and/or as specifically provided herein, appoint the Chief Executive Officer or other administrative personnel as the committee's agent in carrying out the committee's peer review duties.

#### 10.2 Medical Executive Committee.

(b) Process for Selecting and/or Electing and Removing Medical Executive
Committee Members: Members of the Medical Executive Committee shall be
elected/appointed by [\_\_\_\_\_\_]. Once selected by the Medical Staff,
the Medical Executive Committee Members shall be submitted to the Board of
Trustees [Directors] for approval. Any committee member, including members of
the Medical Executive Committee, may be removed by the individual or entity
which elected or appointed the committee member. At large members of the
Medical Executive Committee may be removed from the Executive Committee in
the same manner as officers of the Medical Staff.

**Comment:** Joint Commission Standards require that the Bylaws include the manner in which the Medical Executive Committee Members are selected or elected and removed. Other than requiring approval of the Governing Body, there is no restriction on the process. Medical Executive Committee Members may be selected directly or indirectly.

- (c) Duties: The duties of the Medical Executive Committee shall be:
  - (i) To represent and act on behalf of the Medical Staff, subject to any limitations as may be imposed by these Bylaws and those of the Hospital;
  - (ii) To coordinate the activities and general policies of the various Clinical Departments [Services];
  - (iii) To receive and act upon committee reports and recommendations;
  - (iv) To implement policies of the Medical Staff, including, but not limited to, enforcement of the Medical Staff Bylaws, the Medical Staff Rules and Regulations, the Hospital's Bylaws, and the Hospital Corporate Compliance Plan;
  - (v) To provide liaison among Medical Staff, the Chief Executive Officer and the Board of Trustees [Directors];
  - (vi) To ensure that the Medical Staff is kept abreast of the Hospital's accreditation program and informed of the accreditation status of the Hospital;
  - (vii) To review the credentials of all Applicants and to make recommendations to the Board of Trustees [Directors] for Staff membership assignments to Departments [Services] and delineation of Clinical Privileges;
  - (viii) To review at least every two (2) years all information available regarding the performance and clinical competence of Staff Members and Allied Health Professionals with Clinical Privileges, and as a result of such reviews, to make recommendations for reappointments and renewal or changes in Clinical Privileges;

- (ix) To request evaluations of Practitioner Privileges through the Medical Staff process in instances where there is doubt about the Applicant's ability to perform Privileges requested;
- (x) To make appropriate effort to ensure professional, ethical conduct and competent clinical performance by all Staff Members, including the initiation of and/or participation in Medical Staff corrective action or review procedures when warranted and implementation of any actions taken as a result thereof:
- (xi) Investigate, review and report on any matters related to the conduct or clinical practice of any Practitioner in accordance with these Bylaws;
- (xii) Oversee and direct medical education activities and programs for members of the Medical Staff;
- (xiii) Participate in identifying community health needs and setting Hospital goals and establishing plans and programs to meet those needs;
- (xiv) Recommend Clinical Services to be provided by telemedicine;
- (xv) To report at general Staff meetings regarding the proceedings of all meetings and decisions made regarding Staff policy in the interim between Staff meetings;
- (xvi) To make recommendations on Hospital management matters (e.g., long-range planning) to the Board of Trustees [Directors] through the Chief Executive Officer;
- (xvii) To make recommendations to the Board of Trustees [Directors] regarding Medical Staff structure; participation of the Staff in performance improvement/quality assurance and utilization review activities; and mechanisms for Clinical Privileges delineation, credentials review, termination of Staff membership and fair hearing procedures;
- (xviii) To act on behalf of the Medical Staff, if appropriate and subject to ratification by the Medical Staff, with respect to matters which require action prior to the next scheduled or special meeting of the Staff; and
- (xix) To organize the Medical Staff's performance improvement/quality assurance, quality review and utilization management activities and establish a mechanism to conduct, evaluate and revise such activities.

**Comment:** Joint Commission Standards permit the organized Medical Staff, with the approval of the Hospital's Governing Body, to delegate authority over Medical Staff Rules, Regulations or policies to the Medical Executive Committee. If such authority is delegated, then the above duties

also should describe the Medical Executive Committee's duties to make decisions about Medical Staff Rules, Regulations or policies.

(d) Meetings: The Medical Executive Committee shall meet monthly and otherwise at the call of the Chief of Staff.

#### 10.3 Credentials Committee.

- (a) Composition: The Credentials Committee shall be a standing committee, appointed by the Chief of Staff. A Representative of the Medical Records Department shall act as support Staff to the committee. The committee shall appoint a chairperson.
- (b) Duties: The duties of the Credentials Committee shall be:
  - (i) To review the credentials of all Applicants and to make recommendations to the Medical Executive Committee for membership and delineation of Clinical Privileges in compliance with Article VI and VII of these Bylaws;
  - (ii) To make a report to the Medical Executive Committee on each Applicant for Medical Staff membership and Clinical Privileges, including specific consideration of the recommendations from the Department [Service] in which the Applicant requests Privileges;
  - (iii) To review at least every two (2) years all information available regarding the qualifications and competence of Staff Members and, as a result of such review, to make recommendations for granting of Privileges, reappointments and assignment to the various Departments [Services] as provided in these Bylaws. This will be reported in writing to the Medical Executive Committee at its December meeting;
  - (iv) To investigate any breach of ethics that is reported to it; and
  - (v) To review reports referred by the Staff Committees and by the Chief of Staff.
- (c) Meetings: The Credentials Committee shall meet at the call of the Chief of Staff or chairperson of the Committee.

#### 10.4 Joint Conference Committee.

(a) Composition: The Joint Conference Committee is a standing committee composed of the Chief of Staff, the immediate past Chief of Staff, two (2) members of the Board of Trustees [Directors] of the Hospital appointed by the Chairperson of the Board of Trustees [Directors], and the Chief Executive Officer. [A Representative of the Medical Records Department shall act as support Staff.] All members of the committee shall be entitled to vote except the Chief Executive Officer of the Hospital. The chairperson shall be the chairperson

of the Board of Trustees [Directors] or the Board [Directors] chairperson's appointee.

- (b) Duties: The duties of the Joint Conference Committee shall be to:
  - (i) Conduct itself as a forum for the discussion of administrative and medical policies and procedures requiring agreement among the Board of Trustees [Directors], Medical Staff and Administration;
  - (ii) Oversee Hospital compliance with laws and regulations of federal, state and local government and with standards of the various accrediting and approval bodies;
  - (iii) Consider the formulation and implementation of standards and activities for the review and maintenance of the quality and efficiency of patient care;
  - (iv) Review and recommend updates to Hospital's disaster preparedness activities; and
  - (v) Provide medico-administrative liaison between the Board of Trustees [Directors] and Medical Staff, for discussion and interaction between the Board and Medical Staff on matters referred by the Medical Executive Committee or the Board [Directors].
- (c) Meetings: The Joint Conference Committee shall meet, as necessary, at the call of the chairperson of the Board [Directors] or the Chief of Staff.
- 10.5 Performance Improvement/Quality Assurance Committee.
  - (a) Composition: This committee shall consist of \_\_\_\_\_ (\_\_\_) Physicians appointed by the Chief of Staff for a one (1)-year term, the Director of Nursing, the Medical Records Director and the Chief Executive Officer. The Committee shall elect its own chairperson. Other Hospital Representatives may be invited to participate as indicated by various audit topics.

**Comment:** Because Peer Review Committees are committees of "health care providers" under Missouri law, hospitals may wish to provide that non-health care providers (such as a Chief Executive Officer or Medical Records Director) may attend a committee meeting, as opposed to including such non-health care professional as a "member" of the committee.

- (b) Duties:
  - (i) Medical Records and Review: The committee shall be responsible for assuring that the medical records reflect realistic documentation of medical events. The committee shall conduct a periodic review at least quarterly of selected records of current Hospital patients to assure that they reflect the diagnosis, results of diagnostic tests, therapy ordered,

condition, in-hospital progress and condition of the patient at discharge, and that they are sufficiently complete at all times, so that in the event of transfer of patient responsibility, complete medical comprehension of the case is represented. The committee shall see that the medical reviews are conducted as required and that the results of these reviews are forwarded to the Medical Executive and Credentials Committees, which shall jointly prepare recommendations to the Medical Staff, Chief Executive Officer of the Hospital or other committees by way of the Chief of Staff. The committee shall maintain a permanent record of its proceedings and activities.

(ii) Ongoing Professional Practice Evaluation: The committee shall be responsible for assuring ongoing practice evaluation of Practitioners and AHPs, including, but not limited to, review of operative and other clinical procedures performed, patterns of blood and pharmaceutical usage, requests for tests and procedures, length of stay patterns, morbidity and mortality data, and Practitioner's use of consultants. Information obtained on each Practitioner or AHP through the ongoing professional practice evaluation will be incorporated into the Practitioner or AHP's Medical Staff file for inclusion in decisions regarding Clinical Privileges.

**Comment:** The Joint Commission requires Medical Staff participation in ongoing review of the clinical performance of Practitioners and AHPs. This is not a reactive review based on complaint, as described in Article XII, but is proactive monitoring specific to each Practitioner or AHP. In developing the specific process for this evaluation, the Hospital should ensure any policies are consistent with or integrated into the Hospital's risk management and quality improvement processes.

- (iii) Performance Improvement/Quality Assurance Activities: The committee shall be responsible for assuring the implementation of a planned and systematic process for monitoring and evaluating the quality of the care and treatment of patients served and the Clinical performance of all individuals with Clinical Privileges. The committee shall perform reviews of surgical procedures to help assure that surgery performed in the Hospital is justified and of high quality. The committee shall review and evaluate the quality of clinical practice of the Staff throughout the Hospital at least once each quarter. Review and evaluation shall include selected deaths, unimproved cases, tissue, infections, complications, error in diagnosis and results of treatment. The committee shall report in writing its conclusions, recommendations, action taken and the results of actions.
- (c) Meetings: The Performance Improvement/Quality Assurance Committee shall meet at least quarterly and otherwise at the call of the Chief of Staff, Chairperson of the Committee, Chief Executive Officer or Chairperson of the Board of Trustees [Directors].

#### 10.6 Utilization Review Committee.

(a) Composition: A Utilization Review Committee consisting of two (2) or more Practitioners must carry out the utilization review function. At least two (2) of the members of the committee must be doctors of medicine or osteopathy. As otherwise stated above, this committee shall consist of \_\_\_\_\_\_ (\_\_\_\_) Physicians appointed by the Chief of Staff for a one (1)-year term, the Director of Nursing, the Medical Records Director, Social Worker and Business Office Manager. Other Hospital Representatives may be invited to participate as needed.

**Comment:** Depending on the size of the hospital, the Utilization Review Committee may be either of the following: (i) a Staff Committee of the institution, as provided in this provision, or (ii) a group outside the institution -- (A) established by the local medical society and some or all of the Hospitals in the locality or (B) established in a manner approved by CMS. The committee's or group's reviews may not be conducted by any individual who (i) has a direct financial interest (for example, an ownership interest) in that hospital; or (ii) was professionally involved in the care of the patient whose case is being reviewed.

#### (b) Duties:

- Utilization Review Studies: The committee shall conduct utilization (i) review studies designed to assess and evaluate the appropriateness and/or efficiency of admissions to the Hospital, length of stay, discharge practices, use of medical and Hospital Services, and other factors that may contribute to the effective utilization of Hospital and Physician Services. Specifically, the committee shall analyze how under-utilization and overutilization of each of the Hospital's Services affects the quality of patient care provided at the Hospital, study patterns of care and obtain criteria relating to average or normal (usual) lengths of stay by specific disease categories, and evaluate systems of utilization review employing such criteria. It also shall work toward the assurance of proper continuity of care upon discharge through, among other things, the accumulation of appropriate data on the availability of other suitable health care facilities and services outside the Hospital. The committee shall communicate the results of its studies and other pertinent data to the entire Medical Staff and shall make recommendations for the optimum utilization of Hospital resources and facilities in accordance with quality of patient care and safety.
- (ii) Other duties include: (1) the development of a written utilization review plan, which shall be approved by the Staff and the Board of Trustees [Directors], which must be in effect at all times; (2) the completion of medical care evaluation studies, under criteria developed by the federal government's review organizations; and (3) the review of extended duration stays.

(c) Meetings: The Utilization Review Committee shall meet, as needed, at the call of the Chief of Staff, Chairperson of the Committee, Chief Executive Officer or Chairperson of the Board of Trustees [Directors].

## 10.7 Pharmacy and Therapeutics Committee.

- (a) Composition: Membership shall consist of Representatives of the Medical Staff appointed by the Chief of Staff, the Hospital Pharmacist, a Representative of Nursing Services, the Infection Control Manager, the Performance Improvement/Quality Assurance Coordinator and the Director of the Laboratory. A Medical Records Representative and the Chief Executive Officer shall serve as advisory Staff to the committee and shall attend all meetings.
- (b) Duties: This committee shall be responsible for the development and surveillance of all drug utilization policies and practices within the Hospital. The committee shall assist in the formulation of professional policy statements regarding evaluation, appraisal, selection, procurement, storage, distribution, use, safety procedures and other matters relating to obtaining, retaining and usage of drugs in the Hospital. It also shall perform the following specific functions:
  - (i) Serve in an advisory capacity to the Medical Staff and the Pharmacist with respect to the choice of available drugs;
  - (ii) Make recommendations regarding drugs to be stocked;
  - (iii) Develop and periodically review a formulary for use in the Hospital;
  - (iv) Prevent unnecessary duplication in stocking drugs and drugs-incombination that have identical amounts of the same therapeutic ingredients;
  - (v) Evaluate clinical data concerning new drugs or preparations requested for use in the Hospital; and
  - (vi) Establish standards concerning the use and control of investigational drugs and/or research in the use of recognized drugs.
- (c) Meetings: This committee shall meet at least quarterly and send reports to the Medical Executive Committee regarding its activities.

# 10.8 <u>Infection Control Committee</u>.

(a) Composition: Membership shall consist of Representatives of the Medical Staff appointed by the Chief of Staff, the Hospital Pharmacist, a Representative of Nursing Services, the Infection Control Manager, the Performance Improvement/Quality Assurance Coordinator and the Director of the Laboratory. A Medical Records Representative and the Chief Executive Officer shall serve as advisory Staff to the committee and shall attend all meetings.

- (b) Duties: The Infection Control Committee must develop a system for identifying, reporting, investigating and controlling infections and communicable diseases of patients and personnel. The committee shall be responsible for the surveillance of inadvertent Hospital infection potentials, the review and analysis of actual infections, the promotion of a preventative and corrective program, and the supervision of infection control in all phases of the Hospital's activities, including the operating rooms, recovery room and special care units. It shall study and supervise sterilization procedures, isolation procedures, disposal of infectious material, testing of Hospital personnel for carrier status and other situations as requested by the Medical Executive Committee. It shall be responsible for the overall cleanliness of the Hospital and shall communicate, through the Hospital management Representatives, with the Board of Trustees [Directors] on emergency problems.
- (c) Meetings: This committee shall meet quarterly, and shall maintain a record of its proceedings and activity and shall report in writing through its chairperson to the full Medical Staff.

#### 10.9 Physician Health Committee.

- (a) Composition: The Physician Health Committee shall be a standing committee and shall consist of \_\_\_\_\_\_ (\_\_\_) members of the Active Medical Staff appointed by the Chief of Staff. Except for initial appointments, each member shall be appointed for a term of \_\_\_\_\_ (\_\_\_) years. The terms shall be staggered to achieve reasonable continuity in the committee membership. To the extent possible, members of this committee shall not serve as active participants on other peer review or quality assessment and improvement committees while serving on the Physician Health Committee.
- (b) Duties: The Physician Health Committee shall arrange for education of the Medical Staff and other Hospital Staff about illness and impairment recognition issues specific to Physicians; receive referrals from Physicians and other Hospital Staff regarding issues of illness or impairment of Medical Staff Members; refer affected Physicians to appropriate professional internal or external resources for diagnosis and treatment of illness or impairment; evaluate the credibility of referrals, complaints, allegations or concerns which are referred to the committee; develop a procedure for monitoring the affected Physician and the safety of patients until any rehabilitation or disciplinary process is complete; and report to the Medical Executive Committee instances in which a Physician is providing unsafe treatment. The Physician Health Committee is intended to constitute a Peer Review Committee as defined in 537.035 RSMo, and the proceedings, findings, deliberations, reports and minutes of the Physician Health Committee shall be privileged to the fullest extent allowed under applicable law.
- (c) Meetings: The committee shall meet as often as necessary, but at least quarterly. It shall maintain only such record of its meetings as it deems advisable; however,

it shall report on its activities to the Medical Executive Committee at least quarterly.

**Comment:** This committee is intended to provide a mechanism to implement The Joint Commission Standard requiring that Hospitals implement a Physician health policy. This requirement is discussed in more detail in the Comment to Article XV, Section 15.3.

## 10.10 Special Committees.

(a) Other committees, whether standing or ad hoc, may be established at the discretion of the Chief of Staff and/or the Medical Executive Committee.

# ARTICLE XI - MEDICAL STAFF COMMITTEE AND DEPARTMENT [SERVICE] MEETINGS

**Comment:** Blanks have been left in various provisions of this Article XI to provide hospitals with the flexibility of determining the frequency of regular Medical Staff meetings and the timing of such meetings. Blanks also have been left to allow flexibility for quorum and attendance requirements. A quorum is the number of members of the group that must be present in order to conduct a meeting. Typically, half or more than half of the members of the group constitutes a quorum. However, in hospitals with large medical staffs, the quorum might be lower as a matter of convenience.

**Comment:** Missouri law requires that the organized Medical Staff meet at intervals necessary to accomplish the Medical Staff's required functions. The frequency of meetings or other mechanisms must allow for monthly decision-making by or on behalf of the Medical Staff.

#### 11.1 Medical Staff.

(a)	Regular Meetings. There shall be [()] meetings of the Medical
	Staff to be held at the time and place as determined in accordance with the
	Medical Staff Rules and Regulations. The annual meeting of the Medical Staff
	will be the last [] meeting before the end of the calendar year. At
	the annual meeting, the retiring officers and committees shall make reports
	reviewing activities and achievements of the past year. The agenda also shall
	include the election of officers for the following year and recommendations for
	reappointment to the Staff. The regular meetings shall include, but not be limited
	to, a review and analysis of clinical work done in the Hospital as presented by the
	Performance Improvement/Quality Assurance Committee, the Utilization Review
	Committee or members of the Medical Staff.

## (b) Special Meetings:

(i) The Chief of Staff or the Medical Executive Committee may call a special meeting of the Staff at any time, including monthly meetings of the Medical Staff, if a decision is required by all of the Medical Staff. The Chief of Staff shall call a special meeting within [ ( )] days

of receipt of a request of such meeting signed by [()]	
members of the Active Medical Staff and stating the purpose of suc	h
meeting.	

- (ii) Notice in writing or by phone stating the place, day and hour of any special meeting shall be delivered to each member of the Active Medical Staff not less than one (1) day before the date of such meeting. No business shall be transacted at any special meeting except that stated in the notice calling the meeting.
- (c) Quorum. The presence of [\_\_\_\_\_\_\_ percent (\_\_\_\_%)] of the membership of the Active Medical Staff at any regular or special meeting, including a meeting for the purpose of amendment of these Bylaws, shall constitute a quorum.
- Attendance Requirements. All members of the Active Medical Staff shall be (d) required to attend all meetings of the Staff, regular and special. Absence from \_\_ (\_\_\_\_) consecutive meetings of the Staff, or from more than \_\_ (\_\_\_\_) scheduled meetings in any calendar year without acceptable excuse shall result in reduction of Privileges to Courtesy or Provisional Active Staff for the next year with Active Staff Privileges pending the following year upon the fulfillment of the meeting requirement. The Medical Executive Committee shall decide if an excuse is acceptable or not. The Secretary-Treasurer or another person designated by the Chief of Staff shall notify any Medical Staff Member whose meeting attendance requirement has not been fulfilled of the reduction in Privileges and the Staff Member's procedural rights according to the Bylaws. Upon receipt of appropriate written notice from the affected Practitioner by the Chief of Staff, the Staff Member will be entitled to a hearing or other procedural rights in accordance with Article XIII. Reinstatement of members to Active Staff following this may be made on Application as in the case of original appointment.
- (e) Agenda:
  - (i) The agenda at any regular Medical Staff meeting may be as follows:
    - Call to order:
    - Acceptance of the minutes of the last regular and of all special meetings;
    - Unfinished business:
    - Communications;
    - Report of the Chief Executive Officer of the Hospital;
    - New business;
    - Presentation of interesting or pertinent findings stemming from utilization review and/or patient care evaluation studies;
    - Reports of standing and of special Medical Staff Committees;
    - Discussion and recommendations for improvement of professional work at the Hospital;
    - Educational programs; and

- Adjournment.
- (ii) The agenda at special meetings shall include:
  - The reading of the notice calling the meeting;
  - Transaction of business for which the meeting was called; and
  - Adjournment.

**Comment:** This is a model agenda. It can and should be modified to fit the historical manner in which Medical Staff meetings are conducted in a given hospital. However, as a model, this provides a good measure against which to compare current procedures.

(f) Written minutes of Medical Staff meetings shall be signed by the Secretary/Treasurer and permanently filed on a confidential basis in the Hospital.

## 11.2 Department [Service] and Committee Meetings.

- (a) Regular Meetings: Departments [Services] shall hold meetings as called by the Department Director [Physician Liaison] for that Department [Service] and shall meet to review and evaluate the clinical work of Practitioners with Department [Service] Privileges. All committees shall meet as specified in these Bylaws and may establish their own schedules in accordance with these Bylaws.
- (b) Special Meetings: A special committee meeting may be called by the Chief of Staff, the committee chairperson or by one-third (1/3) of the members of the committee.
- (c) Notice: Written or oral notice stating the place, day and hour of any committee meeting shall be given to each member not less than twenty-four (24) hours before the time of the meeting. The attendance of a member at a meeting shall constitute a waiver of notice of such meeting.
- (d) Quorum: \_\_\_\_\_\_ percent (\_\_\_\_ %) of the Active Medical Staff Members of a committee, but not less than \_\_\_\_\_ (\_\_\_\_) members, shall constitute a quorum at any meeting.
- (e) Manner of Action: The action of a majority of the members present at a meeting at which a quorum is present shall be the action of the committee or Department [Service]. Action may be taken without a meeting by unanimous consent in writing signed by each member entitled to vote thereon.
- (f) Minutes: Minutes of each regular and special meeting of a committee or Department [Service] shall be prepared and shall include a record of the attendance of members and the vote taken on each matter. The minutes shall then be approved, signed by the Secretary and forwarded to the Medical Executive Committee. Each committee shall maintain a permanent file of the minutes of each meeting.

(g)	Attendance Requirements: All members of the Active Staff shall be required to
	attend all meetings of committees and Departments [Services] of which they may
	be members. Absence from () consecutive meetings or from
	more than () of the meetings of any committee or Department
	[Service] for any twelve (12)-month period without acceptable cause shall have
	the same result and be handled in the same manner as described in Subsection
	11.1(d) of this Article XI regarding failure to attend Medical Staff meetings.

#### ARTICLE XII - INVESTIGATION AND CORRECTIVE ACTION

**Comment:** The purpose of this Article XII along with Article XIII is to provide for due process and fair hearing procedures required by state regulations, federal certification requirements and private accreditation standards. The focus is to ensure fundamental fairness to the Practitioner while providing the Hospital with the tools necessary to fulfill its obligation to obtain and maintain competent Practitioners on its Medical Staff.

**Comment:** Careful adherence to the provisions of this Article XII and Article XIII should provide the basis for immunity from liability pursuant to HCQIA. A full explanation of this federal statute is beyond the scope of the Comments to these Bylaws. In addition to HCQIA immunity, adherence to the provisions of the Bylaws is required under Missouri law. In Egan v. St. Anthony's Medical Center, the court granted injunctive relief to a physician to compel the Hospital to comply with the terms of its Bylaws. 244 S.W.3d 169 (2008). The Hospital should consult with its legal counsel for a detailed explanation of the requirements of and protections offered under this law.

**Comment:** It should be noted that these Bylaws do not contain procedures for reporting to the National Practitioner Data Bank, relevant state agencies or other entities requiring such reporting. Instead, the Bylaws in Article XIII, Section 13.12 require that the Chief Executive Officer of the Hospital report any final actions taken by the governing Board [Directors] of the Hospital in accordance with applicable Hospital procedures.

- 12.1 <u>Grounds for Corrective Action</u>. Corrective action against a Practitioner with Staff membership or Clinical Privileges may be initiated whenever the Practitioner engages in or exhibits actions, statements, demeanor or conduct, either within or outside the Hospital, that is, or is reasonably likely to be:
  - (a) Contrary to these Bylaws, the Medical Staff Rules and Regulations, the Hospital's Bylaws, policies or procedures, or the Hospital's Corporate Compliance Plan;
  - (b) Detrimental to patient safety or to the quality or efficiency of patient care in the Hospital;
  - (c) Disruptive to Hospital operations;
  - (d) Damaging to the Medical Staff's or the Hospital's reputation;
  - (e) Below the applicable standard of care; or

- (f) In violation of any law or regulation relating to federal or state reimbursement programs.
- 12.2 <u>Authorization to Initiate</u>. Any of the following may request that corrective action be taken or initiated:
  - (a) Any officer of the Medical Staff;
  - (b) A Department Director [Physician Liaison] of any Department [Service] in which the Practitioner holds membership or exercises Privileges;
  - (c) Any standing committee or subcommittee of the Medical Staff (including the Medical Executive Committee) or a chairperson thereof;
  - (d) The Chief Executive Officer; or
  - (e) The Board of Trustees [Directors].
- 12.3 Discretionary Interview; Resolution Prior to Investigation or Request for Corrective Action. When a party authorized to initiate a corrective action under Section 12.2 receives information that may provide grounds for corrective action under Section 12.1, the party considering initiating an investigation or request for corrective action may, but is not obligated to, afford the Practitioner an interview at which the circumstances prompting the potential investigation are discussed and the Practitioner is permitted to present relevant information on the Practitioner's own behalf. The interview provided in this Section 12.3 of Article XII is not a procedural right of the Practitioner and need not be conducted according to the procedural rules provided in Article XIII of these Bylaws. The party initiating the interview shall prepare a dated, written record of the interview indicating the type of problem, what was discussed with the Practitioner and any proposal as to the type of intervention that will be undertaken to address the problem. This written report shall be forwarded to the Medical Executive Committee and shall be retained in the Practitioner's confidential peer review file. If the interview does not fully resolve the potential situation giving rise for corrective action, the individual may initiate formal investigation by requesting a corrective action under Section 12.4.

**Comment:** Section 12.3 of this Article XII is intended to provide opportunities for resolution of potential corrective actions prior to formal investigation or other action by the Medical Executive Committee that may trigger National Practitioner Data Bank Reporting. Once an "investigation" has begun, even voluntary withdrawal of Privileges may be reportable. The National Practitioner Data Bank is very broad and, depending on the specific facts involved, even very early evaluation of a Practitioner may trigger reporting obligations.

12.4 <u>Initiation, Requests, Notices</u>. All requests for corrective action shall be submitted to the Medical Executive Committee in writing, and shall be supported by a statement of the specific activities or conduct that constitutes the grounds for the request. The Chief of Staff shall promptly notify the Chief Executive Officer in writing of all such requests.

- 12.5 <u>Informal Interview</u>. Upon receipt of a request for corrective action, the Medical Executive Committee or its designee may, at the Committee's option, conduct an informal interview with the Practitioner against whom corrective action has been requested. At such interview, the Practitioner will be informed of the general nature of the charges against the Practitioner and will be invited to discuss, explain or refute them. This interview shall not constitute an "investigation" or a "hearing," will be preliminary in nature, and will not be subject to any of the procedural rules provided in these Bylaws with respect to hearings. A summary of such interview shall be included with the report from the Medical Executive Committee to the Board of Trustees [Directors] of the Hospital and shall be placed in the Practitioner's confidential peer review file.
- 12.6 Investigation. Upon receipt of the request for corrective action, and following any informal interview with the affected Practitioner as described in the preceding Section 12.5 or Section 12.3 above, the Medical Executive Committee shall either act on the request or direct, by written resolution, that a formal investigation concerning the grounds for the corrective action request be undertaken. The Medical Executive Committee may conduct such investigation itself or may assign the task to a Medical Staff officer, a Department Director [Physician Liaison], a standing or ad hoc committee, an individual or group who is not affiliated with the Hospital, or any other Medical Staff component. This investigative process is not a "hearing" as that term is used in Article XIII and shall not entitle the Practitioner to the procedural rights provided in said Article XIII. The investigative process may include, without limitation, a consultation with the Practitioner involved, with the individual or group who made the request and with other individuals who may have knowledge of or information relevant to the events involved. If the investigation is conducted by a group or individual other than the Medical Executive Committee, that group or individual shall submit a written report of the investigation to the Medical Executive Committee as soon as is practicable after the group's or individual's receipt of the assignment to investigate. The Medical Executive Committee may, at any time in its discretion, and shall at the request of the Board, terminate the investigative process and proceed with action as provided below. If the investigating group or individual has reason to believe that Practitioner's conduct giving rise to the request for corrective action was the result of a physical or mental disability, the Medical Executive Committee may require the Practitioner involved to submit to an impartial physical or mental evaluation within a specified time and pursuant to guidelines set forth below. Failure by the Practitioner to comply, without good cause, shall result in immediate suspension of the Practitioner's Medical Staff membership and all Clinical Privileges until such time as the evaluation is completed, the results are reported to the Executive Committee and the Board takes [Directors take] final action. The Executive Committee shall name the Practitioner(s) who will conduct the examination. The Hospital shall pay for the examination. All reports and other information resulting from the mental or physical evaluation shall be maintained in a separate file as a confidential medical record.

**Comment:** The above Section allows the Hospital to engage an individual or group not affiliated with the Hospital to assist in an investigation. Engaging of an outside individual to review records or participate in an investigation may be beneficial in many situations, such as

when other Practitioners have a conflict of interest. However, caution should be exercised when engaging outside individuals as Missouri courts have declined to apply the peer review privilege to these reports. In State ex rel. <u>Kirksville Missouri Hospital v. Jaynes</u>, the court found that an outside physician's report was not a peer review committee report and, therefore, not privileged, despite the fact that the review and report were requested by the peer review committee. 328 S.W.3d 418 (2010). For this reason, Hospital counsel should be consulted prior to engaging outside individuals in the peer review process.

- 12.7 <u>Medical Executive Committee Action</u>. As soon as practicable after the conclusion of the investigative process, if any, the Medical Executive Committee shall act upon the request for corrective action. Its action may include recommending, without limitation, the following:
  - (a) Rejection of the request for corrective action;
  - (b) Verbal warning or a letter of reprimand;
  - (c) Education and/or training;
  - (d) Medical or psychiatric treatment or referral of the Practitioner to the Missouri Physicians Health Program or similar impaired provider program;
  - (e) A probationary period with retrospective review of cases and/or other review of professional behavior but without requirement of prior or concurrent consultation or direct supervision;
  - (f) A requirement of prior or concurrent consultation or direct supervision;
  - (g) A limitation of the right to admit patients;
  - (h) Reduction, suspension or revocation of all or any part of the Practitioner's Clinical Privileges; or
  - (i) Suspension or revocation of the Practitioner's Medical Staff membership.

**Comment:** Section 12.7 of Article XII includes a list of examples of actions which may be taken by the Medical Executive Committee. This list is intended to include some actions that would not require reporting to the National Practitioner Data Bank (at least Subsections 12.7(a) through (e)).

## 12.8 Effect of Medical Executive Committee Action.

(a) When the Medical Executive Committee's recommendation is adverse (as defined in Article XIII of these Bylaws) to the Practitioner, the Chief Executive Officer of the Hospital shall immediately inform the Practitioner by Special Notice, and the Practitioner shall be entitled, upon timely and proper request, to the procedural rights contained in Article XIII.

- (b) When the Medical Executive Committee's recommendation is favorable to the Practitioner, the Chief Executive Officer of the Hospital shall promptly forward it, together with all supporting documentation, to the Board of Trustees [Directors]. Thereafter, the procedure set forth in Article VI, Subsection 6.7(f)(i), is applicable.
- (c) If the Medical Executive Committee fails to act in processing and recommending action on a request for corrective action within an appropriate time as determined by the Board [Directors], the procedure set forth in Article VI, Subsection 6.7(f)(ii), shall be applicable.
- 12.9 Other Action. The commencement of corrective action procedures against a Practitioner shall not preclude the summary suspension of all or any portion of any of said Practitioner's Clinical Privileges in accordance with the procedure set forth in Section 12.10 of this Article XII.

## 12.10 Summary Suspension.

- (a) Whenever a Practitioner's conduct is of such a nature as to require immediate action to protect the life of any patient(s) or to reduce the substantial likelihood of injury or damage to the health or safety of any patient, employee or other person present in the Hospital or to preserve the continued effective operation of the Hospital, any of the following has the authority to suspend summarily the Medical Staff membership or all or any portion of the Clinical Privileges of such Practitioner:
  - (i) The Chief of Staff;
  - (ii) The applicable Department Director [Physician Liaison];
  - (iii) The Chief Executive Officer, after conferring when possible with either the Chief of Staff, the immediate Past Chief of Staff, the appropriate Department Director [Physician Liaison] or the Credentials Committee chairperson;
  - (iv) The Medical Executive Committee; or
  - (v) The Board of Trustees [Directors].
- (b) A summary suspension is effective immediately. The person or group imposing the suspension shall immediately inform the Chief Executive Officer of the suspension, and the Chief Executive Officer shall promptly give Special Notice thereof to the Practitioner. The applicable Department Director [Physician Liaison] shall assign a suspended Practitioner's patients then in the Hospital to another Practitioner, considering the wishes of the patient, where feasible, in selecting a substitute Practitioner.

(c) As soon as possible, [but in no event later than fourteen (14) days] after a summary suspension is imposed, the Medical Executive Committee shall convene to review and consider the need, if any, for a professional review action. Such a meeting of the Medical Executive Committee shall in no way be considered a "hearing" as contemplated in Article XIII (even if the Practitioner involved attends the meeting), and no procedural requirements shall apply. The Medical Executive Committee may recommend modification, continuation or termination of the terms of the summary suspension.

**Comment:** The suggestion of fourteen (14) days as the period in which a hearing should be held is based on the HCQIA. The Act allows a period of fourteen (14) days to investigate a suspension without providing for notice of a hearing. The period of time can be reduced, but should not be extended. In any event, the time period must be reasonable.

- (d) The effect of the Medical Executive Committee's action shall be as set forth in Section 12.8 of this Article XII. The terms of the summary suspension as originally imposed shall remain in effect pending a final decision of the Board of Trustees [Directors].
- 12.11 <u>Automatic Suspension or Revocation</u>. A Practitioner's Medical Staff membership and/or Clinical Privileges may be immediately suspended or revoked in any of the following situations:
  - (a) Occurrences Affecting Licensure:
    - (i) Revocation: When a Practitioner's license to practice in Missouri is revoked, the Practitioner's Medical Staff membership and Clinical Privileges shall be immediately and automatically revoked as of the date of license revocation.
    - (ii) Restriction: When a Practitioner's license to practice in Missouri is limited or restricted, those Clinical Privileges that the Practitioner has been granted that are within the scope of the limitation or restriction are similarly automatically limited or restricted as of the date of license limitation or restriction.
    - (iii) <u>Suspension</u>: When a Practitioner's license to practice in Missouri is suspended, the Practitioner's Medical Staff membership and Clinical Privileges shall be automatically suspended effective upon and for the term of the suspension.
    - (iv) <u>Probation</u>: When a Practitioner is placed on probation by the Practitioner's licensing authority, the Practitioner's voting and office-holding Privileges are automatically suspended effective upon and for at least the term of the licensure probation.
  - (b) Occurrences Affecting Controlled Substances Regulation:

- (i) Revocation: Whenever a Practitioner's Drug Enforcement Administration (DEA) or other controlled substances number is revoked, the Practitioner shall be immediately and automatically divested of the Practitioner's right to prescribe medications covered by the number.
- (ii) Suspension or Restrictions: When a Practitioner's DEA or other controlled substances number is suspended or restricted in any manner, the Practitioner's right to prescribe medications covered by the number is similarly suspended or restricted during the term of the suspension or restriction.
- (c) Medical Records Completion: After written warning (and failure to cure by proscribed date) by the Medical Executive Committee of delinquency or failure timely to prepare or complete medical records, a Practitioner's Clinical Privileges (except with respect to the Practitioner's patients already in the Hospital, already reserved for admission or surgery, and emergency situations), admitting rights, right to consult with respect to new patients, and voting and office-holding Prerogatives shall be automatically suspended. The suspension shall be effective on the date specified in the warning and shall continue until the delinquent records are prepared or completed.
- (d) Professional Liability Insurance: Failure to maintain the minimum required type and amount of professional liability insurance with an approved insurer shall result in immediate and automatic suspension of Practitioner's Medical Staff membership and Clinical Privileges until such time as a certificate of appropriate insurance coverage is furnished.
- (e) Exclusion from State or Federal Health Care Reimbursement Programs: Upon exclusion, debarment or other prohibition from participation in any state or federal health care reimbursement program, the Practitioner's Medical Staff membership and Clinical Privileges shall be immediately and automatically suspended until such time as the exclusion, debarment or prohibition is lifted.

**Comment:** Section 12.11 of this Article XII sets forth various reasons for automatic suspension. Subsection 12.11(e) of Article XII provides that automatic suspension will result from a Practitioner's exclusion from state or federal health care reimbursement programs.

**Comment:** To protect the Hospital's own ability to bill state and federal health care reimbursement programs, the Hospital should ensure that Practitioners are not excluded from participating in any programs with which the Hospital participates. Under Section 6501 of the Affordable Care Act, a state Medicaid agency must deny enrollment or terminate the enrollment of any provider that is terminated on or after January 1, 2011, under Title XVIII of the Social Security Act or under the Medicaid program or Children's Health Insurance Program of any other state.

(f) Conviction of a Crime: Upon conviction of a felony or a crime involving moral turpitude in any court of the United States, either federal or state, the

Practitioner's Medical Staff membership and Clinical Privileges shall be automatically revoked.

## (g) Procedure:

- (i) An automatic suspension is effective immediately. The person or group imposing the suspension shall immediately inform the Chief Executive Officer of the suspension, and the Chief Executive Officer shall promptly give Special Notice thereof to the Practitioner. The applicable Department Director [Physician Liaison] shall assign a suspended Practitioner's patients then in the Hospital to another Practitioner, considering the wishes of the patient where feasible, in selecting a substitute Practitioner.
- (ii) The Practitioner will be given thirty (30) calendar days to produce clear and convincing evidence that the facts relied upon by the Hospital in imposing the automatic suspension are incorrect. In the absence of clear and convincing evidence to the contrary, the individual's Medical Staff membership and Clinical Privileges shall automatically terminate.
- (iii) In the event the Practitioner produces clear and convincing evidence disputing the facts relied upon, the Chief Executive Officer may either reinstate the Practitioner or provide notice of a hearing under Article XIII.
- (h) Procedural Rights and Additional Corrective Action: No Practitioner shall be entitled to the procedural rights set forth in Article XIII as a result of the sanctions automatically imposed pursuant to the preceding Subsections 12.11(a) through (f) of this Article XII. Any of the persons entitled to initiate corrective actions under Section 12.2 of this Article XII may, however, initiate such action on the basis of any of the occurrences specified in Subsections 12.11(a) through (f) of this Article XII, and if, as a result thereof, an adverse recommendation or decision is made which exceeds the scope of the sanctions automatically imposed under Subsections 12.11(a) through (f) of this Article XII, then the Practitioner shall be entitled to the same procedural rights to which the Practitioner would be entitled under Section 12.8 of this Article XII, but only with respect to the additional sanctions recommended or imposed.
- 12.12 <u>Reporting</u>. The Chief Executive Officer shall report any corrective action taken pursuant to these Bylaws to the appropriate authorities as required by law and in accordance with applicable Hospital procedures regarding the same.

#### ARTICLE XIII - HEARING AND APPELLATE REVIEW PROCEDURE

**Comment:** The importance of this Article XIII is noted in the Comments to Article XII. These provisions may be viewed as legalistic and complicated; however, they are important not only to protect the rights of the Practitioner, but also to protect the Hospital and the Medical Staff in carrying out responsible resolution of issues requiring implementation of these procedures. Adherence to the details of this process is very important to demonstrate that the Physicians under

scrutiny received fair treatment. In addition, careful adherence to the requirements of this Article XIII will provide a basis for raising the favorable presumption of fairness to obtain the immunities provided in HCQIA.

- 13.1 <u>Purpose</u>. The purpose of this Article XIII shall be to provide a mechanism for intraprofessional resolution of matters bearing on the professional competency and conduct of Practitioners who have Medical Staff membership and Clinical Privileges at Hospital.
- Right to Hearing. Except as otherwise specifically provided in these Bylaws, the recommendations set forth in Subsection 13.2(a) of this Article XIII shall, if deemed adverse pursuant to Subsection 13.2(b) of this Article XIII, entitle the Practitioner thereby affected to a hearing.
  - (a) Recommendations or Actions:
    - (i) Denial of initial appointment or subsequent reappointment to the Medical Staff;
    - (ii) Suspension or revocation of Medical Staff membership;
    - (iii) Denial of requested appointment to or advancement in Medical Staff category;
    - (iv) Involuntary reduction in Medical Staff category;
    - (v) Limitation of admitting Privileges not related to standard administrative or Medical Staff Policies;
    - (vi) Denial of requested affiliation with a Department [Service];
    - (vii) Denial or restriction of requested Clinical Privileges other than a denial or restriction of temporary Privileges; or
    - (viii) Reduction, suspension or revocation of Clinical Privileges other than reduction, suspension or revocation of temporary Privileges.
  - (b) When Deemed Adverse: A recommendation or action listed in Subsection 13.2(a) of this Article XIII shall be deemed adverse only when it has been:
    - (i) Recommended by the Medical Executive Committee;
    - (ii) Taken by the Board [Directors] contrary to a favorable recommendation by the Medical Executive Committee; or
    - (iii) Taken by the Board [Directors] on its own initiative without benefit of a prior recommendation by the Medical Staff.

- (c) Actions Which Do Not Give Right to Hearing: Notwithstanding the provisions of Subsections 13.2(a) and (b) of Article XIII, no action described in this Subsection 13.2(c) of Article XIII shall constitute grounds for or entitle the Practitioner to request a hearing.
  - (i) An oral or written reprimand or warning;
  - (ii) Imposition of any general consultation requirement or any requirement that the Practitioner must be supervised while performing certain procedures;
  - (iii) Imposition of a probationary period with retrospective or prospective review of cases;
  - (iv) Denial of requested Privileges because the Practitioner failed to satisfy the basic qualifications or criteria of training, education or experience established for the granting of Privileges for a specific procedure or procedures;
  - (v) Ineligibility for Medical Staff appointment or reappointment or the Clinical Privileges requested because a Department [Service] has been closed or there exists an exclusive contract limiting the performance of Privileges within the specialty with which the Practitioner is associated or the Clinical Privileges which the Practitioner has requested to one (1) or more Physicians;
  - (vi) Termination or revocation of Medical Staff appointment or Clinical Privileges either in whole or in part because the Hospital has determined to close a Department [Service] or grant an exclusive contract limiting the performance of Privileges within the specialty in which the Practitioner practices to one (1) or more Physicians;

**Comment:** Some Medical Staff Bylaws provide for a hearing conducted by the Board of Trustees [Directors] or a committee thereof prior to entering into an exclusive contract or closing a Department [Service]. If a Hospital chooses this approach, then the above provision should be removed from the Medical Staff Bylaws. If the Hospital does not intend to provide a hearing when closing a department, it should assure that the above provision is included in the Medical Staff Bylaws.

(vii) Termination of the Practitioner's employment or other contract for Services unless the employment contract or Services contract provides otherwise;

**Comment:** Terms have been incorporated into these Medical Staff Bylaws to allow automatic termination of Medical Staff Membership upon termination of employment with the Hospital. It must also be clear that this action is not an adverse action under the Bylaws so that the Hospital is not required to provide fair hearing. This makes the termination of a Practitioner employment

agreement more straightforward. It also can avoid triggering reporting obligations under the corrective action process.

- (viii) Ineligibility for Medical Staff appointment or requested Clinical Privileges because of lack of facilities or equipment or because Hospital has elected not to perform, or does not provide, the Service which the Practitioner intends to provide or the procedure for which Clinical Privileges are sought;
- (ix) Revocation, suspension or restriction of Clinical Privileges or Medical Staff appointment or denial of Medical Staff reappointment because of the failure of the Practitioner to comply with requirements of the Medical Staff Bylaws or Medical Staff Rules and Regulations pertaining to any required attendance at committee, Department [Service] or general Medical Staff meetings, or payment of required dues or any other requirement not based on professional competence or conduct;
- (x) Reduction, suspension or revocation of Medical Staff appointment or Clinical Privileges as provided in Section 12.11 of Article XII regarding automatic suspension or revocation;
- (xi) Voluntary suspension or relinquishment of Clinical Privileges or Medical Staff membership when professional competence or conduct is not under investigation;
- (xii) Voluntary suspension or relinquishment of Clinical Privileges or Medical Staff membership, which is not in return for the Medical Staff refraining from conducting an investigation of professional competence or conduct;
- (xiii) The imposition of a requirement for retraining, additional training or continuing education; or
- (xiv) Suspension of Privileges, either in whole or in part, or Medical Staff membership for less than thirty (30) days and during which an investigation is being conducted to determine the need for further action.
- 13.3 <u>Notice of Adverse Recommendation or Action</u>. A Practitioner against whom an adverse action has been taken or recommended pursuant to Section 13.1 of this Article XIII shall be given Special Notice of such action. The notice shall:
  - (a) Advise the Practitioner of the nature of and reasons for the proposed action;
  - (b) Advise the Practitioner that the Practitioner has thirty (30) days after receiving the Notice within which to submit a request for hearing on the proposed action;
  - (c) Include a summary of the Practitioner's rights in the hearing, which at a minimum includes the rights described in Subsection 13.8(d) of this Article XIII;

- (d) State that if the Practitioner fails to request a hearing in the manner and within the time period prescribed, such failure shall constitute a waiver of the right to a hearing and to an appellate review on the issue that is the subject of the Notice; and
- (e) State that if the Practitioner properly requests a hearing, the Practitioner shall be provided with written notice of the date, time and place of the hearing. Such notice also shall state the grounds upon which the adverse recommendation or action is based.
- 13.4 Request for Hearing. A Practitioner shall have thirty (30) days after the Practitioner's receipt of a Notice pursuant to Section 13.3 of this Article XIII to file a written request for a hearing. Such request shall be delivered to the Chief Executive Officer either in person or by certified mail.
- 13.5 <u>Waiver by Failure to Request Hearing</u>. A Practitioner, who fails to request a hearing within the time and in the manner specified in Section 13.4 of this Article XIII, waives any right to such hearing and to any appellate review to which the Practitioner might otherwise have been entitled. Such waiver shall constitute acceptance of the adverse action or recommendation, and the adverse action or recommendation shall become a final action.
- 13.6 Right to One (1) Hearing and Appellate Review. Notwithstanding any other provision of this Article XIII to the contrary, no Practitioner shall be entitled as a matter of right to more than one (1) hearing and one (1) appellate review on any matter for which there is a hearing right. Adverse recommendations or actions on more than one (1) matter may be consolidated and considered together or separately, as the Board [Directors] shall designate in its sole discretion.

**Comment:** The foregoing may seem self-evident. However, Practitioners undergoing this process often complain they have not had a hearing or are entitled to additional hearings. This provision makes clear the Practitioner's rights in this regard.

#### 13.7 Hearing Requirements.

(a) Notice of Time and Place for Hearing: Upon receipt from a Practitioner of a timely and proper request for hearing, the Chief Executive Officer shall deliver the same to: (i) the Chief of Staff if the request for hearing was prompted by an adverse recommendation of the Medical Executive Committee under Section 13.2(b)(i), or (ii) to the Chairperson of the Board of Trustees [Directors] if the request for hearing was prompted by an adverse recommendation or action of the Board of Trustees [Directors] under Section 13.2(b)(ii) or (iii). Within ten (10) days after receipt of such request, the Chief of Staff or the Chairperson of the Board [Directors], as applicable, shall schedule and arrange for a hearing. At least thirty (30) days prior to the hearing, the Chief Executive Officer on behalf of the Executive Committee or the Board [Directors], as applicable, shall send the Practitioner Special Notice of the time, place and date of the hearing, which the

date shall not be less than thirty (30) nor more than sixty (60) days after the date of the Notice. Provided, however, that subject to the foregoing, a hearing for a Practitioner who is under summary suspension shall be held as soon as the arrangements may be reasonably made. The Special Notice of the hearing provided to the Practitioner shall include a list of witnesses (if any) expected to testify at the hearing in support of the proposed action and a summary of the Practitioner's rights according to these Bylaws.

- (b) Statement of Issues and Events: The Special Notice of hearing shall contain a concise statement of the Practitioner's alleged acts or omissions, a list of the specific or representative patient records in question, and/or a concise statement of any other reasons or subject matter forming the basis for the adverse action, which is the subject of the hearing.
- (c) Conduct of Hearing: If the adverse action that is the subject of the hearing was recommended by the Medical Staff Executive Committee, the hearing shall be held before a hearing officer or Hearing Committee as determined by the Chief of Staff. If the Board [Directors] took the adverse action, the chairperson of the Board [Directors] shall determine whether the hearing shall be held before a hearing officer or a Hearing Committee. The hearing officer shall be appointed by either the Chief of Staff or the chairperson of the Board [Directors] pursuant to Subsection 13.7(d) of this Article XIII. A Hearing Committee shall be appointed by either the Chief of Staff or the chairperson of the Board [Directors] pursuant to Subsection 13.7(e) of this Article XIII.
- (d) Appointment of Hearing Officer: The hearing officer may be a Physician, Dentist, attorney or other individual qualified to conduct the hearing. The hearing officer is not required to be a member of the Medical Staff. The hearing officer shall not be in direct economic competition with the Practitioner involved in the hearing.
- (e) Appointment of Hearing Committee:
  - (i) By Chief of Staff: A Hearing Committee appointed by the Chief of Staff shall consist of at least three (3) members of any Staff category of the Hospital, one (1) of whom shall be designated as chairperson by the Chief of Staff. Notwithstanding the foregoing, if, because of the requirements of Subsection 13.7(e)(iii) of this Article XIII, or for any other reason, it is not possible to include members of the Medical Staff of Hospital on the Hearing Committee, the Hearing Committee may include members of the active Medical Staff of any hospital. The Chief of Staff shall designate one (1) of the appointees as chairperson of the Hearing Committee.
  - (ii) By Chairperson of the Board: A Hearing Committee appointed by the Chairperson of the Board shall consist of at least three (3) persons. One of the members shall be designated as chairperson by the chairperson of the Board [Directors]. Subject to the provisions of Subsection 13.7(e)(iii) of this Article XIII, at least one (1) member of the committee shall be a

member of the Medical Staff when possible. Other members of the committee are not required to be members of the Medical Staff, and if no member of the Medical Staff is available because of the provisions of Subsection 13.7(e)(iii) of this Article XIII, the Hearing Committee may include a member of the active Medical Staff of any hospital.

(iii) Service on Hearing Committee: A Practitioner or Board [Directors]

Member shall not be disqualified from serving on a Hearing Committee
merely because the Practitioner or Board [Directors] Member participated
in initiating or investigating the underlying matter at issue or because the
Practitioner or Board [Directors] Member has heard of the case. However,
no member of the Hearing Committee may be in direct economic
competition with the Practitioner involved in the hearing. All members of
the Hearing Committee shall be required objectively to consider and
decide the case with good faith.

**Comment:** The prohibition against having anyone in direct economic competition with the subject of the hearing is important for two (2) reasons. First, this is a requirement of the HCQIA as a condition of obtaining immunity from damages if the Practitioner files suit, and, second, it avoids raising the issue of a group boycott that would bring the Hospital and Medical Staff under possible antitrust scrutiny. The authority granted to utilize Physicians from other hospitals helps facilitate this situation from both a legal and practical perspective.

- (iv) Presiding Officer: An individual qualified to conduct hearings may be designated as the presiding officer for a hearing to be heard by the Hearing Committee. Such individual need not be a member of the Hearing Committee.
- (f) List of Witnesses: In addition to the list of witnesses required in the Special Notice of hearing, at least five (5) days prior to the scheduled date for commencement of the hearing, each party shall provide the other with a list of names of the individuals who, as far as then reasonably known, will give testimony or evidence in support of that party at the hearing. Admissibility of testimony to be presented by a witness not so listed shall be at the discretion of the presiding officer.

## 13.8 Hearing Procedure.

- (a) Forfeiture of Hearing: A Practitioner who requests a hearing pursuant to this Article XIII but fails to appear at the hearing without good cause, as determined by the Hearing Committee or hearing officer, shall forfeit the Practitioner's rights to such hearing and to any appellate review to which the Practitioner might otherwise have been entitled.
- (b) Presiding Officer: The hearing officer, the chairperson or the individual designated pursuant to Subsection 13.7(e)(iv) of Article XIII, shall be the presiding officer. The presiding officer shall act to maintain decorum and to

assure that all participants in the hearing process are provided a reasonable opportunity to present relevant oral and documentary evidence. The presiding officer shall be entitled to determine the order of procedure during the hearing and shall make all rulings on matters of law, procedure and the admissibility of evidence.

- (c) Representation: The Practitioner who requested the hearing shall be entitled to be accompanied and represented at the hearing by a member of the Medical Staff in Good Standing, by a member of the Practitioner's professional society and/or by an attorney. The Medical Staff or the Board [Directors] may appoint a member of the Medical Staff in Good Standing and/or an attorney to represent the Medical Staff or the Board [Directors] at the hearing to present the facts in support of its adverse recommendation or action, and to examine witnesses.
- (d) Rights of Parties: During the hearing, each party may:
  - (i) Call, examine and cross-examine witnesses;
  - (ii) Introduce any relevant evidence, including exhibits;
  - (iii) Question any witness on any matter relevant to the issues that are the subject of the hearing;
  - (iv) Impeach any witness;
  - (v) Offer rebuttal of any evidence;
  - (vi) Have a record made of the hearing in accordance with Subsection 13.8(h) of this Article XIII; and
  - (vii) Submit a written statement at the close of the hearing.
- (e) Procedure and Evidence: At the hearing, the rules of law relating to examination of witnesses or presentation of evidence need not be strictly enforced, except that oral evidence shall be taken only on oath or affirmation. If the Practitioner who requested the hearing does not testify in the Practitioner's own behalf, the Practitioner may be called and examined as if under cross examination. The hearing officer or hearing panel may consider any relevant matter upon which responsible persons customarily rely in the conduct of serious affairs regardless of whether such evidence would be admissible in a court of law. Prior to or during the hearing, any party may submit memoranda concerning any procedural or factual issue, and such memoranda shall be included in the hearing record.
- (f) Information Pertinent to Hearing: In reaching a decision, the Hearing Committee or hearing officer shall be entitled to consider any pertinent material contained on file in the Hospital, and all other information that can be considered, pursuant to the Medical Staff Bylaws, in connection with Applications for appointment or

reappointment to the Medical Staff and for Clinical Privileges. The Hearing Committee or hearing officer may at any time take official notice of any generally accepted technical or scientific principles relating to the matter at hand and of any facts that may be judicially noticed by the Missouri courts. The parties to the hearing shall be informed of the principles or facts to be noticed, and the same shall be noted in the hearing record. Any party shall be given the opportunity, upon timely request, to request that a principle or fact be officially noticed or to refute any officially noticed principle or fact by evidence or by written or oral presentation of authority in such manner as determined by the hearing officer or committee.

- (g) Burden of Proof: When a hearing relates to denial, limitation or other restriction of a Practitioner's request for new status or Privileges, including an initial Application for appointment, the Applicant shall have the burden of producing evidence to demonstrate that the adverse decision or recommendation lacks any substantial factual basis or that the basis or the conclusions drawn therefrom are arbitrary, unreasonable or capricious. Otherwise, the body whose adverse decision or recommendation is under consideration at the hearing shall have the initial obligation to present evidence in support thereof, but the Practitioner thereafter is responsible for supporting the Practitioner's challenge that the adverse decision or recommendation lacks any substantial factual basis or that the basis or the conclusions drawn therefrom are arbitrary, unreasonable or capricious.
- (h) Record of Hearing: A record of the hearing shall be kept of sufficient accuracy that an informed and valid judgment can be made by any group that may later be called upon to review the record and render a recommendation or decision in the matter. The Hearing Committee or hearing officer may select the method to be used for making the record, such as a court reporter, electronic recording unit, detailed transcription or minutes of the proceedings. Upon request, the Practitioner shall be entitled to obtain a copy of the record or use an alternative recording method, at the Practitioner's own expense.
- (i) Postponement: Requests for postponement of a hearing may be granted by the chairperson of the Hearing Committee or the hearing officer upon a showing of good cause and only if the request is made as soon as is reasonably practical.
- (j) Presence of Hearing Committee Members and Vote: A majority of the Hearing Committee shall be present at all times during the hearing and deliberations. If a committee member is absent from any part of the proceedings, the presiding officer in the presiding officer's discretion may rule that such member be excluded from further participation in the proceedings or recommendations of the committee.
- (k) Recesses and Adjournment: The Hearing Committee or hearing officer may recess the hearing and reconvene it without additional notice if the committee or officer deems such recess necessary for the convenience of the participants, to

obtain new or additional evidence, or if consultation is required for resolution of the matter. When presentation of oral and written evidence is complete, the hearing shall be closed. The Hearing Committee shall deliberate outside the presence of the parties and at such time and in such location as is convenient to the committee. Upon conclusion of the Hearing Committee's deliberations, the hearing shall be adjourned.

## 13.9 Report and Further Action.

- (a) Within fifteen (15) days after final adjournment of the hearing, the Hearing Committee or hearing officer shall report in writing all findings and recommendations with specific references to the hearing record and other documentation considered and shall forward the report along with the record and other documentation to the body whose adverse recommendation or decision occasioned the hearing.
- (b) Within twenty (20) days after receipt of the report, the Medical Executive Committee or the Board [Directors], as the case may be, shall consider the same and affirm, modify or reverse its recommendation or action in the matter. The decision shall be transmitted, together with the hearing record, the report of the Hearing Committee or hearing officer and all other documentation considered to the Chief Executive Officer.
- If the Medical Executive Committee's decision pursuant to 13.9(b) is adverse to (c) the Practitioner, it shall have the notice and effect provided in Section 13.9(d) of this Article XIII. If the Medical Executive Committee's decision pursuant to Subsection 13.9(b) of this Article XIII is favorable to the Practitioner, the Committee shall promptly forward it, together with all supporting documentation, to the Board [Directors] for its final action. The Board [Directors] shall take action thereon by adopting or rejecting the Medical Executive Committee's decision in whole or in part or by referring the matter back to the Medical Staff Executive Committee for further consideration. Any such referral shall state the reasons therefor, set a time limit within which a subsequent recommendation to the Board [Directors] must be made, and may include a directive that an additional hearing be conducted to clarify issues that are in doubt. After receipt of such subsequent recommendation and any new evidence in the matter, the Board [Directors] shall take action. The Chief Executive Officer shall, on behalf of the Board [Directors] or Medical Executive Committee, as applicable, promptly send notice to the affected Practitioner informing the Practitioner of each action taken pursuant to this Subsection 13.9(c) of Article XIII. A favorable determination shall become the final action of the Board [Directors], and the matter shall be considered closed.
- (d) Notice and Effect of Result.
  - (i) Notice: The Chief Executive Officer, on behalf of the body that made the decision, shall promptly send Special Notice, including a copy of the

- decision, to the affected Practitioner, to the Chief of Staff and to the Board [Directors].
- (ii) Effect of Favorable Result: If the Board's decision pursuant to Subsection 13.9(b) or (c) of this Article XIII, as applicable, is favorable to the Practitioner, such result shall become the final decision of the Board [Directors], and the matter shall be considered closed.
- (iii) Effect of Adverse Result: If the decision of the Medical Executive Committee or of the Board [Directors] continues to be adverse to the affected Practitioner, the Special Notice required by Subsection 13.9(d)(i) of this Article XIII shall inform the affected Practitioner of the Practitioner's right to request an appellate review by the Board [Directors] as provided in Section 13.10 of this Article XIII.

# 13.10 <u>Initiation and Prerequisites of Appellate Review</u>.

- (a) Request for Appellate Review: A Practitioner shall have fifteen (15) days after receiving Special Notice of the Practitioner's right to request an appellant review to submit a written request for such review. Such request shall be directed to the Board [Directors] in care of the Chief Executive Officer in person or by certified mail and may include a request for a copy of the Hearing Committee report and record of all other material, favorable or unfavorable, if not previously forwarded, that was considered in taking the adverse action. If the Practitioner wishes an attorney to represent the Practitioner at any appellate review appearance permitted in this Article XIII, the Practitioner's request for appellate review shall so state.
- (b) Waiver by Failure to Request Appellate Review: A Practitioner who fails to request an appellate review in accordance with Subsection 13.10(a) of this Article XIII waives any right to such review. Such waiver shall have the same force and effect as provided in Section 13.5 of this Article XIII regarding waiver by failure to request a hearing.
- (c) Notice of Time and Place for Appellate Review: Upon receipt of a timely request for appellate review, the Chief Executive Officer shall deliver such request to the Board [Directors]. As soon as practicable, the Board [Directors] shall schedule and arrange for an appellate review, which shall not be less than fifteen (15) days nor more than thirty (30) days after receipt by the Chief Executive Officer of the request for review; provided, however, that an appellate review for a Practitioner who is under summary suspension shall be held as soon as the arrangements for the appellate review may be reasonably made, but not later than fifteen (15) days after the Chief Executive Officer's receipt of the request. At least ten (10) days prior to the date of the appellate review, the Chief Executive Officer shall send the Practitioner Special Notice setting forth the time, place and date of the review. The Appellate Review Body may extend the time for the appellate review for good cause, and if the request is made, as soon as is reasonably practicable.

(d) Appellate Review Body: The Board [Directors] shall determine whether the appellate review shall be conducted by the Board [Directors] as a whole or by an appellate review committee composed of three (3) or more members of the Board appointed by the chairperson of the Board [Directors]. If a committee is appointed, one (1) of its members shall be designated as chairperson by the Board [Directors] chairperson. The Appellate Review Body shall, in all circumstances, include a member of the Medical Staff appointed by the chairperson of the Board [Directors].

#### 13.11 Appellate Review Procedure.

- (a) Nature of Proceedings: The proceedings by the review body shall be in the nature of an appellate review based upon the record of the hearing before the Hearing Committee or hearing officer, the Hearing Committee or hearing officer's report, and all subsequent results and actions thereof. The Appellate Review Body also shall consider any written statements submitted pursuant to Subsection 13.11(b) of this Article XIII and such other materials as may be presented and accepted under Subsections 13.11(c)(ii) and (iii) of this Article XIII.
- (b) Written Statements: The Practitioner seeking the review shall, at least five (5) days prior to the scheduled date of the appellate review, submit to the Appellate Review Body (through the Chief Executive Officer), a written statement that describes the findings of fact, conclusions and procedural matters with which the Practitioner disagrees, and the reasons for such disagreement. The group whose adverse action occasioned the review may submit a written statement in support of the group's action and/or in reply to the Practitioner's statement. Any such statement by the group whose action occasioned the review shall be furnished to the Appellate Review Body at least two (2) days prior to the scheduled date of the appellate review. The Chief Executive Officer, on behalf of the Appellate Review Body, shall immediately forward a copy of the statement to the affected Practitioner. The Appellate Review Body in its sole discretion may waive the time limits provided in this Subsection 13.11(b) of Article XIII.

# (c) Conduct of the Appellate Review:

- (i) The chairperson of the Appellate Review Body shall preside over the appellate review, including determining the order of procedure, making all required rulings and maintaining decorum during all proceedings.
- (ii) The Appellate Review Body may, at its discretion, allow the parties or their Representatives to appear and make statements. Parties or their Representatives appearing before the review body must answer questions posed to them by the review body.
- (iii) If a party wishes to introduce new matters or evidence not raised or presented during the original hearing and not otherwise reflected in the record, the party may introduce such information at the appellate review

only if expressly permitted by the review body in its sole discretion and only upon a showing by the party requesting consideration of the information that it could not have been discovered in time for the initial hearing. Prior to introduction of such information at the review, the requesting party shall provide, to the Appellate Review Body and the other party, a written, substantive description of the information.

- (d) Presence of Members and Vote: A majority of the review body shall be present at all times during the review and deliberations. If a review body member is absent from any part of the review or deliberations, the chairperson of the review body, in the chairperson's discretion, may rule that such member be excluded from further participation in the review or deliberations or in the recommendation of the review body.
- (e) Recesses and Adjournments: The Appellate Review Body may recess the review proceeding and reconvene the same without additional notice if it deems such recess necessary for the convenience of the participants or to obtain new or additional evidence or consultation required for resolution of the matter. When oral statements (if allowed) are complete, the appellate review shall be closed. The review body shall then deliberate outside the presence of the parties at such time and in such location as is convenient to the review body. The appellate review shall be adjourned at the conclusion of those deliberations.
- (f) Action Taken: The Appellate Review Body may recommend that the Board affirm, modify or reverse the adverse result or action or, in its discretion, may refer the matter back to the Hearing Committee or hearing officer for further review and recommendation to be returned to it within ten (10) days and in accordance with its instructions. Within five (5) days after receipt of such recommendation after referral, the review body shall make its recommendation to the Board [Directors] as provided in this Subsection 13.11(f) of Article XIII.
- 13.12 <u>Final Decision of the Board</u>. Within forty-five (45) days after receipt of the Appellate Review Body's recommendation, the Board [Directors] shall render its final decision in the matter in writing and shall send notice thereof to the affected Practitioner and to the Chief of Staff. The Board's [Directors'] final decision shall be immediately effective, and the matter shall not be subject to any further referral or review.
- 13.13 <u>Reporting</u>. The Chief Executive Officer shall report any final action taken by the Board [Directors] pursuant to these Bylaws to the appropriate authorities as required by law and in accordance with applicable Hospital procedures regarding the same.

## 13.14 General Provisions.

(a) Waiver: If at any time after receipt of notice of an adverse recommendation, action or result, the affected Practitioner fails to make a required request or appearance or otherwise fails to comply with this Article XIII, the Practitioner shall be deemed to have consented to such adverse recommendation, action or

- result and to have voluntarily waived all rights to which the Practitioner might otherwise have been entitled with respect to the matter involved and the adverse action or recommendation shall become a final action.
- (b) Exhaustion of Remedies: Any Applicant or member of the Medical Staff must exhaust the remedies afforded by this Article XIII before resorting to any form of legal action.
- (c) Release: By requesting a hearing or appellate review under these Bylaws, a Practitioner or Applicant agrees to be bound by the provisions of these Bylaws relating to immunity from liability.

## ARTICLE XIV - CONFIDENTIALITY, REPORTING IMMUNITY AND RELEASES

**Comment:** This Article XIV providing for confidentiality, reporting immunity and releases is self-implementing without additional actions. However, Sections 14.5 and 14.6 of Article XIV require Applicants for appointment and reappointment to the Medical Staff to execute general and specific releases. In spite of the self-implementing nature of this Article XIV, such releases should be obtained in order to provide maximum protection to the Hospital.

- 14.1 <u>Special Definitions</u>. The following are special definitions applicable only to this Article XIV:
  - (a) "Information" means all communications, regardless of form, relating to the subject matter of Section 14.5 of this Article XIV: Examples of such Information shall include, but not be limited to, the following: data, reports, records, minutes or other records of proceedings, memoranda, findings, recommendations, opinions, conclusions, actions, or forms whether written, oral, electronic, digital or in any other form or format. Such Information may include but is not limited to any matter that directly or indirectly affects the quality, appropriateness or efficiency of patient care by a Provider, including, but not limited to, a Provider's professional qualifications and judgment, clinical ability, character and professional ethics, mental and physical health, and utilization patterns.
  - (b) "Malice" means actual knowledge of the falsity of statements or information or reckless disregard for the truth in making statements or disseminating Information.
  - (c) "Provider" means any Practitioner or AHP who has, or has applied for, Medical Staff membership or Clinical Privileges at the Hospital.
  - (d) "Third Parties" means any person (including organizations as well as individuals) who provides Information to any Representative.
- 14.2 <u>Authorizations and Conditions</u>. Submission of an Application for Staff membership or for the exercise of Clinical Privileges or the provision of patient care services in this Hospital constitutes a Provider's express:

- (a) Authorization for Hospital Representatives and the Medical Staff to request, provide and/or act upon Information bearing on the Provider's professional competency, conduct and qualifications;
- (b) Agreement to be bound by the provisions of these Bylaws (including this Article XIV), Medical Staff Rules and Regulations, Hospital's Bylaws, Hospital's policies and procedures, and Hospital's Corporate Compliance Plan, and to waive all legal claims against any Hospital Representative who acts in accordance with such provisions, including the provisions of this Article XIV; and
- (c) Acknowledgment that the provisions of this Article XIV are express conditions to Provider's Application for, acceptance of, and continuation of Staff membership, and to Provider's exercise of Clinical Privileges at Hospital.
- 14.3 Confidentiality of Information. The following Information shall, to the fullest extent permitted by law, be confidential and shall not be disclosed or disseminated except to a Representative or used in any way except as permitted in these Bylaws and as allowed by applicable law. Information relating to a Provider that is collected, prepared or submitted by a Representative of the Hospital or any other hospital or health care organization or facility or Medical Staff for the purpose of monitoring, evaluating or improving the quality of health services; determining that health services rendered were professionally indicated or were performed in compliance with the applicable standard of care; evaluating the qualifications, competence and performance of health care Providers or acting upon matters relating to the discipline of a health care Provider; reducing morbidity or mortality; evaluating the quantity, quality and timeliness of health care Services rendered; and conducting research, or teaching, establishing or enforcing guidelines designed to keep costs of health care within reasonable bounds. Similarly, such Information provided by Third Parties shall be confidential to the extent permitted by law. No such Information shall be considered a part of or be included in any patient record. Each Provider acknowledges that violation of the confidentiality provided for in this Article XIV shall be grounds for revocation of Staff membership and Clinical Privileges.

**Comment:** Section 14.3 of this Article XIV purports to provide for broad confidentiality of clinical and quality Information. This is qualified by the limited confidentiality authorized by law. The principle law governing this in Missouri is contained in the peer review statutes found primarily at 537.035 RSMo. It is not improper to draft the Bylaws with a broad confidentiality provision. However, the drafter should be aware that recent case law may somewhat limit the protection provided by these statutes.

## 14.4 Immunity from Liability.

(a) For Action Taken: Representatives of the Hospital or Medical Staff shall be immune from liability to a Provider for damages or other relief for such Representatives' actions, statements, recommendations, opinions, decisions or other conduct performed within the scope of their duties as a Representative, if such Representatives act in good faith and without malice after reasonable effort

- under the circumstances to ascertain the truthfulness of the facts and in the reasonable belief that the decision, opinion, action, statement or recommendation is warranted by such facts.
- (b) For Providing Information: Representatives of the Hospital or Medical Staff and Third Parties shall be immune from liability to a Provider for damages or other relief for providing Information, even if such Information would otherwise be privileged or confidential, to a Representative of this Hospital or Medical Staff or to any other health care facility or organization concerning a Provider, provided that such Information is related to the scope of the Representative's Peer Review Committee, related to the performance of the duties and functions of the recipient, and is reported in a factual manner; and further provided that the Representative or Third Party acts in good faith and without Malice.

**Comment:** The courts generally construe immunity from suit, like confidentiality provisions, narrowly. The suggested language is appropriate in order to take full advantage of whatever the law and courts will give. However, the drafters are encouraged to discuss this issue with legal counsel as part of the drafting process so the Hospital and Medical Staff personnel have a full understanding what protection the law provides.

- 14.5 Activities Covered. The confidentiality and immunity provided by this Article XIV apply to all acts, communications, proceedings, interviews, reports, records, minutes, forms, memoranda, statements, recommendations, findings, evaluations, opinions, conclusions or disclosures performed or made in connection with this or any other health care facility's or organization's credentialing, peer review, performance improvement/quality assurance and utilization review activities including, but not limited to:
  - (a) Applications for appointment or Clinical Privileges;
  - (b) Applications for reappointment or addition or renewal of Clinical Privileges;
  - (c) Corrective, supervisory or disciplinary action;
  - (d) Hearings and appellate reviews;
  - (e) Performance improvement/quality assurance activities;
  - (f) Utilization review and utilization management activities;
  - (g) Profiles and profile analysis;
  - (h) Risk management activities; and
  - (i) Any other Hospital, committee, Department [Service] or Staff activities related to evaluating, monitoring and maintaining quality and efficient patient care and professional conduct.

14.6 <u>Releases</u>. Upon request of the Hospital, each Provider shall execute general and specific releases in accordance with this Article XIV. Such releases will operate in addition to the provisions of this Article XIV and execution of such releases shall not be a prerequisite to the effectiveness of this Article XIV.

**Comment:** This Section 14.6 of Article XIV provides that separate releases obtained from Applicants for appointment and reappointment to the Medical Staff or other documents relating to authorizations, confidentiality and immunity from liability are to be interpreted to be in addition to, and not in limitation of, any provisions of applicable law.

14.7 <u>Cumulative Effect</u>. Any provisions in these Bylaws and in the Application or other Hospital or Staff forms relating to authorizations, confidentiality of information and immunities from liability are in addition to, and not in limitation of, other protections provided by applicable law.

#### **ARTICLE XV - GENERAL PROVISIONS**

15.1 <u>Staff Rules and Regulations</u>. The Medical Staff shall adopt such Rules and Regulations as necessary to implement the general principles set forth in these Bylaws. Any such Rules and Regulations shall be subject to Board [Directors] approval. The Medical Staff shall follow the procedures outlined in Article XVI of these Bylaws in the adoption and amendment of the Rules and Regulations, except that Medical Staff action may occur by a majority vote at any regular meeting at which a quorum is present and without previous notice, or at any special meeting on notice, by majority vote of those present who are eligible and qualified to vote.

**Comment:** Unlike with the Bylaws, the Medical Staff is permitted to delegate the authority to adopt Rules and Regulations and policies of the Medical Staff to the Medical Executive Committee. If such delegation is elected, the following additional provisions are required by The Joint Commission. Subsection 15.1(d) of Article XV below is intended to address The Joint Commission requirement that the organized Medical Staff has the ability to propose Medical Staff Rules and Regulations and policies directly to the Board [Directors].

- (a) If the Medical Executive Committee proposes to adopt a Rule and Regulation, or an amendment thereto, it must first distribute the proposal to the Medical Staff for review and comment.
- (b) When the Medical Executive Committee adopts a policy or amendment, it must communicate such action to the Medical Staff in writing.
- (c) A policy of the Medical Staff may be proposed to the Board [Directors] without prior approval of the Medical Staff.
- (d) Rules and Regulations and policies of the Medical Staff also may be proposed by a signed petition of \_\_\_\_\_\_ percent (\_\_\_%) of the Medical Staff or approved by the majority vote of the Medical Staff. Such proposed Rules and Regulations and policies of the Medical Staff shall be submitted to the Medical

Executive Committee for review and opinion prior to submission to the Board of Trustees [Directors] for approval. The comments of the Medical Executive Committee may be submitted to the Board [Directors] with the approved rule, regulation or policy.

15.2 <u>Urgent Amendment to Rules and Regulations</u>. An urgent amendment of the Rules and Regulations may be provisionally adopted where the Medical Executive Committee has a documented belief that the amendment is necessary to comply with a law or regulation. Following adoption by the Medical Executive Committee, the urgent amendment must be provisionally approved by the Board of Trustees [Directors]. The urgent amendment shall be provided immediately to the Medical Staff. If objection from the Medical Staff is not received by the Medical Executive Committee within \_\_\_\_\_ (\_\_\_) days, such amendment shall become final. If, however, the Medical Staff objects to the urgent amendment within the time specified above, such amendment shall not be final and may be submitted to the conflict resolution process described in Section 15.3 of this Article XV.

**Comment:** The authority of the Medical Executive Committee to provisionally adopt urgent amendments to Medical Staff Rules and Regulations must be delegated by the voting members of the organized Medical Staff. The Joint Commission requires the Medical Staff be afforded the opportunity to object to the provisional amendment. In order to avoid allowing a single member's disagreement to trigger the conflict resolution process, a hospital can insert a threshold percentage of the Medical Staff required to constitute an objection.

- 15.3 <u>Management of Conflicts Between the Medical Staff and the Medical Executive</u>

  <u>Committee</u>. A Conflict Resolution Committee shall be formed to resolve any conflict arising between the Medical Executive Committee and a group comprised of at least \_\_\_\_\_ percent (\_\_\_%) of the members of the Medical Staff entitled to vote.
  - (a) The Conflict Resolution Committee shall be an ad hoc committee and shall consist of at least \_\_\_\_\_ (\_\_) members, including equal representation from the Medical Executive Committee and the group of Medical Staff members. The Committee may include a third-party facilitator as determined necessary by the Committee members or the Chief Executive Officer.
  - (b) The authority of the Conflict Resolution Committee shall be limited to resolution of disputes related to Medical Staff rules, regulations and policies.
  - (c) If the Conflict Resolution Committee is able to come to an agreement with regards to the disputed rule, regulation or policy, such agreement shall be submitted to the Board of Trustees [Directors] for approval.
  - (d) If, after good faith efforts to resolve the dispute have been exhausted, the Committee is unable to reach agreement, the unresolved portions of the rule, regulation or policy shall be submitted to the Board of Trustees [Directors] for final decision.

(e) Nothing in this conflict resolution procedure is intended to prevent the Medical Staff from proposing a rule, regulation or policy directly to the Board of Trustees [Directors].

**Comment:** The above Section 15.3 of Article XV is intended to address The Joint Commission requirement that the organized Medical Staff implement a process to manage conflict between the Medical Staff and the Medical Executive Committee on issues including, but not limited to, proposals to adopt a rule, regulation or policy or an amendment thereto.

15.4 Physician Health. The Medical Staff Rules and Regulations shall include policies and procedures to identify and manage matters of individual Physician health. The policies and procedures shall be designed to (either through internal processes or by referral to an impaired provider program approved by the Hospital Board of Trustees [Directors]) provide education about Physician health; address prevention of physical, psychiatric or emotional illness; and facilitate confidential diagnosis, treatment and rehabilitation of Physicians who suffer from a potentially impairing condition. If at any time a Physician's health renders the Physician unable to safely perform the Physician's Privileges, the matter shall be forwarded to the Medical Executive Committee for appropriate corrective action.

Comment: This provision is designed to comply with requirements of The Joint Commission on Accreditation of Health Care Organizations Standards. The standards also recommend that the required policy include mechanisms for education of Medical Staff and other Staff about illness and impairment recognition issues; self-referral by a Physician and organization Staff; referrals to internal or external resources for treatment; maintenance of confidentiality; evaluation of creditability of complaints and concerns; monitoring of affected Physicians and safety of patients until completion of rehabilitation and/or discipline; and reporting to Medical Staff leadership of instances of Physicians providing unsafe treatment.

15.5 <u>Departments [Services]</u>. Subject to the approval of the Medical Executive Committee and the Board [Directors], each Department [Service] will formulate its own written policies, if any, as needed for the conduct of its affairs and the discharge of its responsibilities.

**Comment:** Note that Section 15.5 of this Article XV identifies the Hospital's Corporate Compliance Plan. Such a plan is strongly recommended based upon federal law. Section 15.5 provides that the Medical Staff acknowledges and agrees to abide by the Plan and report violations as required by the Plan.

15.6 Corporate Compliance. To ensure that its business practices are conducted with the highest of ethical standards, Hospital has adopted a Corporate Compliance Plan (the "Plan"). This Plan generally requires the conduct of business in compliance with all applicable laws, regulations and standards and provides that any possible violations be reported to the Compliance Officer or \_\_\_\_\_\_\_. The Practitioners and AHPs acknowledge and agree that, in performance of their duties, they are expected to follow these same standards and to report to the above-designated persons any possible violations of laws, regulations, standards or acceptable business practices.

- 15.7 <u>Conflicts of Interest</u>. Each Medical Staff Member shall disclose in writing on an annual basis any potential conflicts of interest (including any ownership or contractual interest the Practitioner or Practitioner's immediate family members may have with the Hospital or its related entities, suppliers, vendors or contractors) that the Staff Member may have with the Hospital or its related entities, consistent with the Hospital's conflict of interest policy.
- 15.8 <u>Staff Dues</u>. The Medical Executive Committee, with the approval of the Active Staff, will establish the amount and manner of disposition of annual dues, if any. Dues are payable at the beginning of each new Medical Staff year. Failure, unless excused by the Medical Executive Committee for good cause, to render payment within two (2) months of the start of the new Staff year shall, after Special Notice of the delinquency, result in automatic suspension of Staff membership (including all Prerogatives) and Clinical Privileges until the delinquency is remedied. The Medical Executive Committee, applicable Department Directors [Physician Liaisons] and organizational components on which the delinquent Practitioner holds membership will be notified of the suspension.

#### ARTICLE XVI - ADOPTION AND AMENDMENT OF BYLAWS

**Comment:** The Joint Commission requires the organized Medical Staff to adopt and amend Medical Staff Bylaws. The responsibility to adopt or amend Medical Staff Bylaws cannot be delegated. Once adopted or amended by the Medical Staff, the Bylaws become effective only upon approval of the Board [Directors]. Under Joint Commission Standards, neither the organized Medical Staff nor the Board [Directors] may unilaterally amend the Medical Staff Bylaws or the Rules and Regulations.

16.1 Medical Staff Authority and Responsibility. Because the Board [Directors] has delegated to the Medical Staff the authority and responsibility to initiate and recommend to the Board [Directors] the Bylaws and related protocols establishing the Staff's organizational structure, and governing its processes and manner of acting, subject only to certain limitations detailed in the Board's [Directors'] Bylaws, the adoption and amendment of these Bylaws require the actions specified in Sections 16.2 and 16.3 of this Article XVI.

**Comment:** While the Medical Executive Committee may be permitted to propose amendments to the Bylaws, the adoption or amendment of the Bylaws cannot be delegated by the Medical Staff, and any proposed amendment from the Medical Executive Committee must be voted on and approved by a majority of the members of the Active Staff. The Medical Executive Committee may be permitted to make non-substantive changes to the Bylaws, such as renumbering or correction of grammar or typographical errors.

16.2	Medical Staff Action. Adoption or amendment of these Bylaws shall require the
	affirmative vote of percent (%) of the Staff Members eligible and qualified
	to vote on Bylaws, cast at a regular or special Staff meeting. A copy of the proposed
	documents or amendments must be given to each Staff Member entitled to vote thereon
	with the Notice of the meeting. The Medical Staff's action shall be forwarded to the
	Board [Directors] for its action.

**Comment:** Joint Commission Standards require that the organized Medical Staff has the ability to adopt Medical Staff Bylaws, Rules and Regulations, and policies, and amendments thereto, and to propose them directly to the Board [Directors].

- 16.3 <u>Board Approval</u>. Medical Staff recommendations regarding adoption or amendment of Bylaws are effective upon the affirmative vote of a majority of the members of the Board of Trustees [Directors].
- 16.4 <u>Conflict Resolution</u>. When the Board disapproves a Bylaw or Bylaw amendment presented by the Medical Staff or proposes to adopt or amend Medical Staff Bylaws and such Bylaw or Bylaw amendments are contrary to Medical Staff recommendations, the Board shall submit the Bylaw or Bylaw amendment to the Medical Staff for consideration.
  - (a) If the Bylaw or Bylaw amendment is approved by the affirmative vote of the majority of the Medical Staff, the Bylaw or Bylaw amendment shall be effective upon such date.
  - (b) If the Medical Staff does not approve the proposed Bylaw or Bylaw amendment, the approved Bylaw or Bylaw amendment shall be returned to the Board of Trustees [Directors] for consideration. The Board of Trustees [Directors] may either approve the Bylaw or Bylaw amendment, refer the Bylaw or Bylaw amendment to a Joint Conference Committee comprised of members of the Board and the Medical Executive Committee, or follow the conflict resolution process as provided in the Board of Trustees [Directors] Bylaws. If submitted to a Joint Conference Committee, the Committee's recommendation must be presented to the Board of Trustees [Directors] and Medical Staff for approval.

**Comment:** Section 16.4 of Article XVI addresses the situation where there is disagreement between the Governing Board [Directors] and Medical Staff concerning an amendment to the Medical Staff Bylaws.

ADOPTED by the Medical Staff on	, 2
Chief of Staff	
APPROVED by the Board of Trustees [Directors] on	, 2
Chairperson, Board of Trustees [Directors]	

# **MODEL**

# **MEDICAL STAFF**

# **RULES & REGULATIONS**

	OF	
		_
Adopted:		
Revised:		

© 2017 Missouri Hospital Association and Lathrop Gage LLP All Rights Reserved

#### **General Comment**

These Rules and Regulations are intended to be a Model which can be tailored to meet the needs of individual hospitals regardless of size. Because they are a Model, the Rules and Regulations may have more detail than is needed for a given hospital, or a hospital may determine that additional Rules and Regulations on specialized subjects are necessary. Each section of the Model Rules and Regulations should be customized to address specific procedural requirements for the particular facility. Comments to a number of sections are intended to focus discussion as to whether particular language is needed and where the language may be customized for a hospital's specific situation.

Procedural specifics for application of Medical Staff Bylaws may be described in Medical Staff Rules and Regulations or Medical Staff policies. These Rules and Regulations incorporate all of the common provisions into a single document for the purposes of simplicity and understanding. Specific provisions may be easily reorganized into separate policies for additional elaboration.

Where a regulation is triggered by a federal or state requirement, we provide citations to the applicable law. The C.F.R. is the Code of Federal Regulations. RSMo is the Missouri Revised Statutes. The C.S.R. is the Missouri Code of State Regulations. Both citations will be found throughout the document. Where a comment refers to the law generally, authority may be found in a statute, regulation or case law.

In evaluating which provisions should be retained or excluded from a hospital's Medical Staff Rules and Regulations, a long view should be taken concerning potential community needs. Development of Medical Staff Rules and Regulations is a time-consuming process, as it should be. Thoroughness and completeness at the outset may prevent the need for frequent amendments. An important aspect of accomplishing this is to include Medical Staff leadership in the development process from the beginning.

The Joint Commission requires that any key issue of the Medical Staff be addressed in the Medical Staff Bylaws and not the Rules and Regulations.

#### **RULES AND REGULATIONS**

#### OF THE MEDICAL STAFF OF

\_\_\_\_\_

## ARTICLE 1 GENERAL

- 1.1 Capitalized terms used in these Rules and Regulations shall have the same meaning as such terms in the Medical Staff Bylaws.
- 1.2 Each Physician who is an Active Member of the Medical Staff has a continuous duty to designate another Active Member of the Medical Staff who has equivalent Privileges to be the Physician's official alternate. In case of unavailability of such alternate, the Chief of the Medical Staff shall have the authority to call any Member of the Medical Staff should it be considered necessary.
- 1.3 The meetings of the Medical Staff shall be held at least quarterly or as otherwise provided in Article XI of the Medical Staff Bylaws. One (1) meeting shall be considered an annual meeting for election of officers.
- 1.4 Regular meetings of the Medical Staff will be held to review the work, reports and recommendations of the Medical Staff and its committees and to complete Medical Staff administrative duties. The Medical Staff also will encourage educational programs.

**Comment:** Any additional requirements for Medical Staff meetings may be inserted in this Section 1.4 of Article 1. These additional requirements may include frequency of review for certain committee activity, frequency for educational programs or other functional requirements. Missouri law requires the Medical Staff meet at intervals to accomplish required functions and establish a mechanism to make monthly decisions.

1.5 A Physician shall be on-duty or on-call at all times. The Medical Staff shall participate in on-call coverage according to policies established by the Medical Staff and the Hospital Administration. Physicians shall comply with the EMTALA Rules and Regulations as outlined herein.

**Comment:** A Physician is required to be on-call or on-duty at all times, 42 C.F.R. 482.12(c)(3); 19 C.S.R. 30-20.092(4). If the Hospital has Physician ownership, there are additional requirements for disclosures related to times when a Physician is not present in the facility. These additional requirements are not included here, but may be found at 42 C.F.R. 411.362.

1.6 The Hospital maintains Disaster Programs for both the handling of mass casualties arising from external disasters and for internal disasters such as fire. All Medical Staff members shall participate in these Disaster Programs as assigned by the Chief of Staff

and as otherwise specified in the program documents. These programs shall be reviewed at least once a year by key Hospital personnel.

**Comment:** Hospitals are required to have internal and external disaster plans 19 C.S.R. 30-20.108(4). The Joint Commission also requires that hospitals have an Emergency Operations Plan in accordance with requirements set forth at EM.02.01.01.

1.7 The Medical Staff shall participate in Hospital performance improvement activities to improve the quality of care, treatment and Services, and patient safety. Medical Staff Members shall participate in the development and implementation of these activities as required in related Hospital policies.

Medicare Conditions of Participation and Missouri law require Medical Staff **Comment:** responsibility and accountability for the quality improvement and patient safety programs. 42 C.F.R. 482.21(e) and 19 C.S.R. 30-20.086(12). The quality improvement plan must include assurance of ongoing communication, reporting and documentation of patient care issues, and quality improvement activities and their effectiveness to the Governing Body and Medical Staff at least quarterly. 19 C.S.R. 30-20.112(3)(D).

1.8 The Medical Staff shall abide by Hospital's infection control and safety policies to enable the administration to protect patients from contagious disease or to protect the patients from self-harm.

# **ARTICLE 2** PROVISION OF PATIENT CARE

#### **Admission to Care** 2.1

- Authority to admit and supervise treatment of patients is exclusively delegated to (a) Practitioners with status and Privileges required by the Medical Staff Bylaws for admission and treatment of patients. The admitting Practitioner must be in Good Standing under the Medical Staff Bylaws.
- (b) If a patient is admitted by a Practitioner other than a Physician, the Practitioner must designate a Physician who has agreed to oversee the medical care provided to the patient.

**Comment:** Patients may only be admitted to the Hospital upon the recommendation of a licensed Practitioner permitted under State law to admit patients. Missouri law, 19 C.S.R. 30-20.082(10) requires that each patient's general condition be the primary responsibility of a Physician Member of the Medical Staff.

A provisional diagnosis, justification of admission and continued hospitalization, (c) patient's progress, and response to medications and Services must be recorded on all medical records. Where provision of diagnosis is delayed due to emergency, the provisional diagnosis shall be given as soon after admission as possible.

**Comment:** A provisional diagnosis is a required element of medical record documentation. 42 *C.F.R.* 482.24(c); Joint Commission Standard RC.02.01.01. Missouri law requires that all medical records justify admission and continued hospitalization, support the diagnosis, and describe the patient's progress and response to medications and services. 19 C.S.R. 30-20.094(5).

## 2.2 Medical Necessity and Utilization Review

- (a) The admitting Practitioner is responsible for deciding whether the patient should be admitted as an inpatient or receive Services as an outpatient. If the admitting Practitioner expects that the patient will require hospital care for at least two (2) midnights, inpatient admission may be appropriate. The decision to admit a patient is a complex medical judgment, which can be made only after the Practitioner has considered a number of factors, including the patient's medical history and current medical needs, the types of facilities available to inpatients and to outpatients, the Hospital's Bylaws and admissions policies, and the relative appropriateness of treatment in each setting. Factors to be considered when making the decision to admit include such things as: the severity of the signs and symptoms exhibited by the patient, the medical predictability of something adverse happening to the patient, the need for diagnostic studies that appropriately are outpatient Services (i.e., their performance does not ordinarily require the patient to remain at the Hospital for twenty-four (24) hours or more) to assist in assessing whether the patient should be admitted, and the availability of diagnostic procedures at the time when, and at the location where, the patient presents.
- (b) The admitting Practitioner is responsible for completing documentation sufficient to support the medical necessity for an inpatient admission. Documentation also should support the clinical indications for diagnostic tests and treatments ordered by the Practitioner.
- (c) The Hospital has a Utilization Review Committee responsible for reviewing the medical necessity of admissions to the Hospital, duration of stays and professional Services furnished, including drugs and biologicals. Practitioners are expected to cooperate with the Utilization Review Committee to ensure medically appropriate treatment and efficient use of Hospital resources in the provision of patient care.

Comment: The above provisions are not required to be in Medical Staff Rules and Regulations by federal or state law. However, Medicare and other third party payors are scrutinizing the appropriateness of reimbursement for Hospital Services where the decisions regarding level of care, Services performed and treatment plan are in the control of the Medical Staff. As such, the above provisions may be added to the Medical Staff Rules and Regulations to guide Medical Staff Members in their decisions and ensure their involvement in the process. Most of this information may be found in the Medicare Benefit Policy Manual, Pub. 100-02. Utilization review requirements are found at 42 C.F.R. 482.30.

#### 2.3 Coordination of Care

A patient's general medical condition shall be managed and coordinated by a (a) Physician, provided that a Dentist, Psychologist, Podiatrist or AHP with appropriate Clinical Privileges may manage and coordinate such care to the extent permitted under the Medical Staff Bylaws, relating to Prerogatives of Practitioners and AHPs and to the extent to which such care is within the scope of practice of the Practitioner or AHP, as permitted by Missouri law.

Medicare Conditions of Participation allow Dentists, Podiatrists, Optometrists, **Comment:** Chiropractors and clinical Psychologists to supervise care of patients in the Hospital. 42 C.F.R. 482.12(c)(1). However, Medicare Conditions of Participation and Missouri law require that each patient's general condition be the primary responsibility of a Physician Member of the Medical Staff. 42 C.F.R. 482.12(c)(4); 19 C.S.R. 30-20.082(10).

(b) Every patient must be assessed and reassessed at an interval appropriate to address the medical needs of the particular patient. Patients shall be seen at least once a day by the attending Practitioner or a qualified designee, and documentation of such assessment will be entered in the medical record.

**Comment:** The Joint Commission requires that a hospital ensure that each patient is reassessed as necessary based on the plan of care. Joint Commission Standard PC.01.02.01, *PC.01.02.03. Most hospitals require a daily evaluation of inpatients by the admitting Practitioner.* Additional sections specifying more or less frequent visits for a specific department (eg. ICU, Skilled Nursing, Rehab, etc.) may be inserted as applicable for the Hospital based on clinical preferences or reimbursement requirements.

Where the admitting Practitioner transfers care to the control and supervision of (c) another Practitioner, such transfer must be clearly documented in the patient's medical record.

#### 2.4 **Radiologic Services**

The Radiology Department and all radiology patient Services shall be under the (a) supervision of a designated Medical Staff Physician who is a Radiologist. Whenever possible, this Physician shall be an attending or consulting Radiologist.

**Comment:** A qualified full-time, part-time or consulting Radiologist must supervise ionizing radiology Services and must interpret only those radiologic tests that are determined by the Medical Staff to require a Radiologist's specialized knowledge. 42 C.F.R. 482.26(c)(1). Missouri law requires that the Physician responsible for the patient Services of the Radiology Department be responsible for overseeing the development of radiology policies and procedures. 19 C.S.R. 30-20.102(2).

Use of radiology equipment and administration of radiology procedures is limited (b) to personnel considered qualified per requirements established with the approval of the Medical Staff.

**Comment:** The Medical Staff is responsible for designating personnel as qualified to use radiology equipment and administer procedures. 42 C.F.R. 482.26(c)(2). This may be maintained in a separate document approved by the Medical Staff or incorporated into this Section 2.4 of Article 2.

(c) The interpretation of all radiologic examinations shall be made by Physicians qualified by education and experience in radiology. A written report of the findings and evaluation of each radiological examination performed, or course of treatment conducted, shall be signed by the Physician responsible for the procedure and shall be made a part of the patient's permanent medical record. Fluoroscopy shall be conducted by, or under the direct supervision of, a Physician.

**Comment:** The Medical Staff should determine which radiologic tests require the specialized knowledge of a Radiologist and which tests may be interpreted by Physicians who are not Radiologists.

#### 2.5 **Consultation**

- (a) A consultant must be a Member of the Medical Staff with appropriate Clinical Privileges and qualified to give an opinion in the field in which the consultant's opinion is sought.
- (b) A consultation request may be appropriate in the following circumstances:
  - (i) Any time the nature of the outcome is not clear or diagnosis is obscure;
  - (ii) The problem is outside the normal scope of practice of the Practitioner;
  - (iii) Anticipated treatment requires a Practitioner with Privileges in the specialty to properly manage or treat the patient;
  - (iv) In cases in which the patient exhibits severe psychiatric symptoms, including drug overdose and attempted suicide, except when the attending is a Psychiatrist;
  - (v) When the patient does not respond to conventional treatment or is a poor risk for operation or treatment; or
  - (vi) When requested by the patient or the patient's family.
- (c) The order for a consultation must be documented in the patient's medical record.
- (d) The consultant must respond to a consultation request within a reasonable amount of time, taking into account whether the consultation request is urgent or routine.
- (e) The consultant shall communicate all findings to the referring Practitioner and document the consultation in the patient's medical record. Such documentation must evidence a review of the patient's record by the consultant, describe pertinent findings on examination of the patient, and provide the consultant's

opinion and recommendations. A limited statement such as "I concur" is not an acceptable report of consultation. When operative procedures are involved, the consultation notes shall, except in emergency situations, be recorded prior to the operation.

#### 2.6 **Surgical and Anesthesia Services**

A roster of Medical Staff Members with surgical Privileges shall be maintained in (a) the surgical suite and available to the surgical nurse supervisor.

**Comment:** Such delineation of Privileges must be determined per the Medical Staff Bylaw provisions related to Clinical Privileges. 42 C.F.R. 482.51(a)(4).

(b) Except in cases of extreme emergency, a study of the patient shall be completed and recorded before inpatient surgery. This study shall include: complete history, physical examination and recording of the preoperative diagnosis, as well as appropriate laboratory work. If not recorded, the surgery will be deferred. Patient studies for outpatient surgery will be handled according to Hospital's policies and procedures, as amended from time to time.

**Comment:** A history and physical must be documented or updated (if performed in the thirty (30) days prior to admission) in a patient's medical record prior to performance of any surgery or procedure requiring anesthesia. 42 C.F.R. 482.24(c)(4)(i) & 482.51(b).

(c) Surgical assistants may be used at the discretion of the operating surgeon. All surgical assistants must either have applicable Clinical Privileges at Hospital or, if not within a category of AHP-provided Clinical Privileges, be approved otherwise by the Hospital to provide patient care.

**Comment:** Surgical Privileges for each Practitioner must be determined in accordance with the Medical Staff Bylaws, and a roster of surgical Privileges must be maintained in the surgical suite. Licensed practical nurses and surgical technologists (operating room technicians) may serve as "scrub nurses" under the supervision of an R.N. 42 C.F.R. 482.51(a).

(d) Written, signed, informed consent shall be obtained by the operating surgeon prior to the operative procedure, except in those situations wherein the patient's life is in jeopardy, and suitable signatures cannot be obtained due to the condition of the patient. In emergencies involving a minor or unconscious patient in which consent for surgery cannot be immediately obtained from parents, guardian or next of kin, these circumstances should be fully explained in the medical record. A consultation in such instances may be desirable before the emergency operative procedure is undertaken, if time permits.

**Comment:** While the Practitioner is typically responsible for describing the risks of a procedure and obtaining informed consent from the patient, the Hospital is responsible for ensuring evidence of the informed consent is obtained in writing, except in emergency situations. 42 C.F.R. 482.51(b)(2); 19 C.S.R. 30-20.140(6).

(e) All tissues removed during an operation shall be the property of the Hospital and shall be examined by a competent Physician whose report shall form a part of the patient's medical record.

**Comment:** The Hospital must have policies and procedures regarding the handling of tissues post-operatively to ensure that the tissues are examined by a Physician and a report is included in the medical record. 42 C.F.R. 482.27(a)(3); 19 C.S.R. 30-20.098(11).

(f) All tissues removed shall be macroscopically examined. If deemed necessary by written Hospital policies and procedures, tissues shall then be microscopically examined.

**Comment:** The Medical Staff and a Pathologist must determine which tissue specimens require a macroscopic (gross) examination and which require both macroscopic and microscopic examinations. A list of all tissues which routinely do not require microscopic examination must be developed in writing by a Pathologist and approved by the Medical Staff of each hospital. 42 C.F.R. 482.27(a)(4) and 19 C.S.R. 30-20.098(11).

- (g) There shall be a Department or director of anesthesia that shall be responsible for all anesthetics administered. Anesthesia shall be administered only by a qualified individual with appropriate Clinical Privileges at Hospital and licensure from the state of Missouri.
- (h) Each patient requiring anesthesia shall have a pre-anesthesia evaluation by a Physician regarding the choice of anesthesia. This evaluation must be performed within forty-eight (48) hours prior to surgery or a procedure requiring anesthesia Service.

**Comment:** Administration of anesthesia is limited to a qualified Anesthesiologist, a doctor of medicine or osteopathy (other than an Anesthesiologist), a Dentist, on oral surgeon, a Podiatrist qualified to administer anesthesia, a certified registered nurse anesthetist ("CRNA"), or an anesthesiologist's assistant. 42 C.F.R. 482.52(a); 19 C.S.R. 30-20.120(3); and 19 C.S.R. 30-20.001(2). Under Missouri law, pre-anesthesia evaluation must be performed by a Physician. 19 C.S.R. 30-20.120(5).

- (i) Each patient's condition shall be reviewed immediately prior to induction. This shall include a review of the patient's medical record with regard to completeness of pertinent laboratory data and an appraisal of any changes in the condition of the patient as compared with that noted on the patient's medical record.
- (j) Following the procedure for which anesthesia was administered, the anesthetist or a designee shall remain with the patient as long as required by the patient's condition relative to the patient's anesthesia status and until responsibility for proper patient care has been assumed by other qualified individuals. A post anesthesia evaluation, completed and documented by the individual qualified to administer anesthesia, must be completed no later than forty-eight (48) hours after surgery or a procedure requiring anesthesia Services.

**Comment:** Documentation and supervision of anesthesia Services must be performed by someone qualified to administer anesthesia. It must include pre-anesthesia evaluation, an interoperative anesthesia record and post-anesthesia evaluation. The anesthesia provider is responsible for monitoring the care of the patient until such care can be transferred to qualified individuals. Specifics regarding anesthesia record information, monitoring frequency or qualified individuals may be included in separate Hospital policy and may vary depending on the procedure performed or type of anesthesia administered. 42 C.F.R. 482.52(b)(1); 19 C.S.R. 30-20.120.

#### 2.7 **Dental Services**

- (a) In order to be granted Clinical Privileges in dentistry and/or oral and maxillofacial surgery, a Dentist shall complete the initial Application process as outlined in the Medical Staff Bylaws.
- (b) Oral and maxillofacial surgeons may perform the medical history and physical examination, to the extent that they have such Clinical Privileges in accordance with the Medical Staff Bylaws, in order to assess the medical, surgical and anesthetic risks of the proposed procedures.
- (c) Dentists may write orders and prescribe medications within the limits of their licensure and Clinical Privileges granted by the Medical Staff.

# 2.8 **Podiatry Services**

- (a) Podiatric Privileges shall be delineated pursuant to the Medical Staff Bylaws. Podiatrists with Clinical Privileges may provide consultation on the request of a Member of the Medical Staff.
- (b) Podiatric patients will be admitted by a Physician Member of the Medical Staff who shall be responsible for the pre-operative medical evaluation of the patient, care of any pre-existing or inter-current medical problems, and completion of the medical record planning discharge instructions. The Podiatrist may write orders directly related to the operative procedure.
- (c) The podiatric history and physical may be performed by the Podiatrist along with the operative report and post-operative progress record as permitted by the Medical Staff Bylaws. The Physician Member of the Medical Staff involved in the patient's care must provide appropriate documentation to supplement any record created by the Podiatrist. The Podiatrist shall complete all patient medical records in compliance with the Hospital's medical records, Rules and Regulations. The justification for the operative procedure shall be clearly delineated in the report.

**Comment:** Where a Podiatrist provides Services to a patient, there should be clear documentation of the involvement of a Physician in the patient's care, including admission of the patient. 42 C.F.R. 482.12(c)(4); 19 C.S.R. 30-20.082(10).

# 2.9 **Psychologist Services**

- (a) In order to be granted Clinical Privileges in psychology, a Psychologist shall complete the initial Application process as outlined in the Medical Staff Bylaws.
- (b) All patients admitted to the Hospital by a Psychologist must also be under the care of a Physician Member of the Medical Staff.
- (c) A Psychologist may perform the medical history and physical examination, to the extent that the Psychologist has such Clinical Privileges in accordance with the Medical Staff Bylaws, in order to assess the psychological condition and treatment of the patient. An admitting history and physical shall be submitted or supplemented by a Physician Member of the Medical Staff who is selected by the Psychologist.
- (d) A Psychologist may write orders and prescribe medications within the limits of the Psychologist's licensure and Clinical Privileges granted by the Medical Staff.

**Comment:** A Psychologist may be permitted by a Hospital Medical Staff to admit patients, but a doctor of medicine or osteopathy also must be involved in the care of the patient. Requiring the Physician's involvement in the History & Physical (H&P) is one (1) option to document the Physician's involvement. 42 C.F.R. 482.12(c)(3); 19 C.S.R. 30-20.082(10).

# 2.10 **Discharge from Care**

- (a) Patients shall be discharged only on the order of a Practitioner and, in the case of post-operative patients, by approved post-operative discharge criteria.
- (b) Discharge planning and evaluation shall begin at the time of admission. Patients who are likely to suffer adverse health consequences upon discharge, if there is no adequate discharge planning, must be identified. A registered nurse ("R.N."), social worker or other appropriately qualified personnel must develop or supervise the development of a discharge planning evaluation for such patients. Medical Staff shall support and coordinate discharge planning with Hospital Staff.

**Comment:** 42 C.F.R. 482.43 and 19 C.S.R. 30-20.104 describe the requirements of a discharge plan. This process is completed by Hospital multidisciplinary personnel but should involve coordination with the attending Physician. Depending on Services provided to the patient, additional discharge instructions or care may be required.

# ARTICLE 3 MEDICAL RECORDS

#### 3.1 General

**Comment:** Each hospital should have specific policies regarding who is an authorized individual for documentation and where in the record that individual is permitted to complete documentation. Physician documentation and orders typically are segregated from documentation of nursing and other clinical staff. The basic requirements for medical records can be found at 42 C.F.R. 482.24, 19 C.S.R. 30-20.094 and Joint Commission Standard

RC.02.01.01. Additional documentation may be required to support coverage for certain services or types of admissions. The Medicare Benefit Policy Manual and Medicare Claims Processing Manual should be reviewed for specific service lines to determine if additional requirements should be integrated into this Article 3.

(a) Entries in the medical record are made only by authorized individuals. All entries must be dated, authenticated by the person making the entry and legible. At a minimum, medical records shall contain sufficient information to identify the patient clearly, to justify the diagnosis and treatment, and to accurately document the results accurately.

**Comment:** The Joint Commission has implemented additional requirements regarding the specific demographic information included in a patient's medical record. Joint Commission Standard 02.01.01. Missouri Medicaid regulations implement additional elements of documentation that may be required, depending on the type of Service, for the record to be considered "accurate" under Missouri law. 13 CSR 70-3.030(2)(A).

(b) Each medical record shall include, at a minimum: admitting diagnosis; results of all consultative evaluations of the patient and appropriate findings by clinical and other staff involved in the care of the patient; documentation of complications, Hospital acquired infections, and unfavorable reactions to drugs and anesthesia; properly executed informed consent forms for procedures and treatments specified by the Medical Staff, or by federal or state law if applicable, to require written patient consent; all Practitioners' or AHPs' orders, nursing notes, reports of treatment, medication records, radiology and laboratory reports, and vital signs and other information necessary to monitor the patient's condition; discharge summary with outcome of hospitalization, disposition of case and provisions for follow-up care; and final diagnosis.

**Comment:** The above listed elements of a medical record are required under the Medicare Conditions of Participation. 42 C.F.R. 482.24(c). The Hospital may expand upon this list to incorporate its specific record sections. Variations also may be made depending on procedures performed, diagnosis or patient age.

(c) In addition to the above listed elements, each inpatient medical record shall include a unique identifying record number; pertinent identifying and personal data; history of present illness or complaint; if injury, how the injury occurred; past history; and family history.

**Comment:** Missouri law places these additional requirements on inpatient records. 19 C.S.R. 30-20.094(9). Other requirements are set forth at Joint Commission Standard RC.02.01.01, including requirements for the collection of demographic information.

(d) An appropriate record shall be maintained for every patient receiving emergency service and will be incorporated in the patient's medical record, if one exists. Its contents shall include patient identification, time and method of arrival, history, physical findings, treatment, and disposition. The emergency room record shall

be signed by the Physician in attendance who is responsible for its clinical accuracy. This record shall be separate from and in addition to any centralized Emergency Services Log.

**Comment:** Hospitals are required to maintain an emergency record integrated into the patient's medical record. Applicable requirements of the overall medical record should be documented. Additional specific requirements may be incorporated to support the medical necessity of Services provided by the Hospital in the Emergency Department. 19 C.S.R. 30-20.092(10).

(e) An appropriate record for outpatient observation (less than twenty-four (24) hours) shall be maintained.

**Comment:** Applicable elements of the requirements of medical record documentation should be recorded on outpatient visits. Requirements specific to a procedure or Service may be incorporated to support reimbursement for the service. The Medical Staff should identify outpatient services requiring the documentation of medical history and physical examination. The list of outpatient Services requiring history and physical may be adjusted based on the particular Services of the Hospital.

(f) Each medical record shall be treated as confidential in accordance with the Hospital's privacy and security policies for the protection of individually identifiable health information. Records may only be accessed by individuals authorized by Hospital policies and procedures.

**Comment:** All medical records must be confidential. Specifics regarding security, confidentiality and access of medical records is typically included in the Hospital's HIPA.A. Compliance policies and/or Medical Records Department policies. 42 C.F.R. 482.24(b)(3); 19 C.S.R. 30-20.094(7).

(g) Records may be removed from the Hospital's jurisdiction and safekeeping only in accordance with a court order, subpoena or statute. All records are the property of the Hospital and shall not otherwise be taken away without permission of the Chief Executive Officer. In case of re-admission of a patient, all previous records shall be available for the use of the attending Physician. This shall apply whether the patient is attended by the same Physician or another Physician.

**Comment:** The Hospital is responsible for the security of medical records, and records are only permitted to be removed from Hospital property as authorized by the Governing Body or where specifically authorized by Missouri law or court order. 19 C.S.R. 30-20.094(6).

(h) All Discharge Summaries shall identify the patient, and contain sufficient information to support the diagnosis, justify the treatment, document the course and results of the treatment, and permit adequate continuity of care among health care providers. Discharge Summaries also shall contain instructions given to the patient relating to physical activity, medication, diet and follow-up care.

12

(i) Only abbreviations and symbols approved by the Medical Staff may be used in the medical records. Each abbreviation or symbol shall have only one (1) meaning, and an explanatory legend shall be available for use by all concerned. There shall be a list of abbreviations and symbols that shall not be used in handwritten communications.

**Comment:** Clarity in the use of abbreviations is required under Missouri law. 19 C.S.R. 30-20.094(4).

### 3.2 **History and Physical**

- (a) A medical history and physical examination must be completed and documented no more than thirty (30) days before or twenty-four (24) hours after admission or registration, but prior to a surgery or a procedure requiring anesthesia Services. The medical history and physical examination must be placed in the patient's medical record within twenty-four (24) hours after admission or registration, but prior to surgery or a procedure requiring anesthesia Services.
- (b) An updated history and physical examination must be completed and documented, including any changes in the patient's condition, when the medical history and physical examination are completed within thirty (30) days before admission or registration. Documentation of the updated examination must be placed in the patient's medical record within twenty-four (24) hours after admission or registration, but prior to surgery or a procedure requiring anesthesia Services.
- (c) At minimum, the history and physical must include \_\_\_\_ [include required elements of the history and physical per Hospital policy].

**Comment:** Joint Commission Standard MS 03.01.01EP 6 requires that the Medical Staff specifies the minimal content of medical history and physical examinations, which may vary by setting or level of care, treatment, and services.

(d) An Advanced Practice Registered Nurse ("APRN") or Physician Assistant ("PA) with appropriate Clinical Privileges working in collaboration with a Physician may perform a history and physical examination. Physician shall co-sign the history and physical examination.

Comment: The types of Practitioners or AHPs other than Physicians who are permitted to document history and physical will depend on the Medical Staff Bylaws and assigned Clinical Privileges. The Joint Commission, in accordance with state law and policies, permits the Medical Staff to choose to allow AHPs to perform part or all of a patient's medical history and physical examination under the supervision of, or through appropriate delegation by, a specific qualified doctor of medicine or osteopathy who is accountable for the patient's medical history and physical examination. Joint Commission Standard MS.03.01.01. The requirement that the Physician cosign the history and physical examination is one (1) method of demonstrating Physician involvement in the patient's admission.

© 2017 Missouri Hospital Association and Lathrop Gage LLP

## 3.3 **Operative Records**

(a) Pre-operative and post-operative medical records must be completed in an accurate and timely manner. An accurate and complete description of findings and techniques of operation shall be completed immediately after surgery, and the surgeon must enter a progress note in the patient's chart prior to transferring the patient to the next level of care. The surgeon must dictate or hand write the operative report to be placed in the medical record immediately after surgery.

**Comment:** Medicare Conditions of Participation require immediate dictation or writing of the operative report. 42 C.F.R. 482.51(b)(6). Missouri law requires completion of the documentation, but does not specify a period of time. 19 C.S.R. 30-20.140(7).

#### 3.4 Anesthesia Records

- (a) The Anesthesia Department shall maintain a complete anesthesia record to include evidence of pre-anesthetic evaluation, inter-operative anesthesia and post-anesthetic follow-up of the patient's condition. The surgeon/Physician should note use of anticipated general, spinal or other regional anesthesia, and the Anesthesiologist or Anesthetist shall document the type of anesthesia administered on the anesthetic record.
- (b) A record of events taking place during the induction and maintenance of and emergence from anesthesia, including the dosage and duration of all anesthetic agents, other drugs, intravenous fluids and blood or blood fractions, shall be made.

**Comment:** The anesthesia record must be completed according to Missouri law and Medicare Conditions of Participation. Missouri law requires the pre-anesthesia record to be completed by a Physician within forty-eight (48) hours before surgery, except in emergency situations. Interoperative and post-anesthetic records can be completed by an individual qualified to provide anesthesia services. Post-anesthesia records must be completed within twenty-four (24) hours after surgery. 19 C.S.R. 30-20.120; 42 C.F.R. 482.52(b).

#### 3.5 **Signature Requirements**

- (a) All clinical entries in the patient's medical record shall be accurately dated and authenticated by the individual making the entry. The method of acceptable authentication used shall be either:
  - (i) Legible full signature;
  - (ii) Legible first initial and last name;
  - (iii) Illegible signature or initials over a typed or printed name;
  - (iv) Illegible signature or initials matching a signature log maintained on file by the Medical Records Department; or

- (v) An electronic signature.
- (b) Where an authentication is made by electronic signature, the individual authenticating the record must have signed an electronic signature agreement agreeing to maintain the confidentiality and integrity of the username and password. Medical Staff Members found to be in violation of the Hospital's electronic signature policies may be subject to discipline under the Medical Staff Bylaws.

**Comment:** Where the Hospital utilizes an electronic medical record, it should implement a policy regarding control of electronic signatures to ensure the integrity of each signature. This is frequently accomplished through an agreement with each user who is permitted to authenticate medical records. This provision may require revision to ensure consistency with Hospital policy.

(c) Use of rubber stamp signatures is strictly prohibited.

**Comment:** All medical record documentation must be appropriately authenticated to be considered valid. Failure to appropriately authenticate a record may result in its inadmissibility in court or invalidity in the event of an audit. Authentication requirements can be found at 19 C.S.R. 30-20.094(3); 42 C.F.R. 482.24(c)(1) and Medicare Program Integrity Manual, Pub. 100-08, Chapter 3, Section 3.3.2.4. Missouri allows stamped signatures so long as the stamp is not shared with other individuals; however, Medicare and most other payers will not recognize stamped signatures, so the practice is commonly not permitted by Hospitals.

(d) Where possible, authentication should be made at the time of the creation of the record. All authentication must be made within thirty (30) days of the creation of the record or as otherwise required by these Rules and Regulations or Hospital policy. Where authentication is not made within thirty (30) days, the individual authenticating the record must attest to the accuracy of the information utilizing Hospital approved attestation language.

**Comment:** The thirty (30) day completion requirement is the outside, longest deadline for most medical record authentication. This time can be shortened consistent with Hospital policy.

#### 3.6 **Delinquent Records**

(a) Medical records are considered delinquent if they remain incomplete for more than thirty (30) days following the patient's discharge from the Hospital.

Incomplete record items include: missing dictation or reports, such as discharge summaries, history and physical examination records, operative reports, and consultations; unsigned dictation and reports; unsigned verbal orders; and other unsigned written entries in the chart (e.g. progress notes, post-operative notes and consents). In addition, history and physical examination records are considered delinquent if they are not dictated within twenty-four (24) hours of admission and operative reports are considered delinquent if they are not dictated immediately following the procedure. Medical records which are unavailable to the

15

Practitioner or AHP, and remain incomplete for thirty (30) days following the patient's discharge, will not be considered delinquent.

**Comment:** The above timeline is the outside limit for timely documentation under the Medicare Conditions of Participation. The Hospital may desire to shorten the timely documentation requirement generally, or specific to certain records. Missouri Medicaid regulations indicate that "adequate" documentation must be completed contemporaneously with the time of service, which is defined as within five (5) business days. 13 C.S.R. 70-3.030(2)(D).

- (b) The Practitioner or AHP whose records are delinquent will be notified via written notice that records must be completed within one (1) week from the date on which the notice is sent. If the records are not complete within one (1) week, admitting Privileges will be suspended.
- (c) Suspension of a Practitioner shall be interpreted in accordance with Subsection 12.11(c) of Article XII of the Medical Staff Bylaws to mean that the Practitioner may not admit patients under the Practitioner's own or any other Practitioner's name during the period of suspension, may not provide consultation Services, and may not schedule surgeries or procedures. The suspended Practitioner will continue to be responsible for attending the Practitioner's own patients admitted to the Hospital prior to the suspension, proceed with patient surgeries/procedures scheduled prior to the suspension, deliver pregnancies as applicable, and provide evaluation and treatment of emergency cases at the request of the Chief Executive Officer or President of the Medical Staff. The suspended Practitioner also will be consulted regarding which alternate Practitioner should assume responsibility for admitting the suspended Practitioner's patient's to the Hospital during the period of suspension. The Practitioner's Privileges will be automatically reinstated upon completion of the delinquent records.

**Comment:** All medical record documentation must be completed within thirty (30) days of discharge. 42 C.F.R. 482.24(c)(4)(viii); 19 C.S.R. 30-20.094(19). A Hospital may impose a shorter time period for completion of medical records and consider medical records delinquent prior to the thirty (30) days. Records completed more than thirty (30) days from date of Service also may involve signature concerns, discussed above.

# ARTICLE 4 ORDERS

#### 4.1 **Medication and Treatment Orders**

(a) Medication or treatment shall be administered only upon written and signed orders of a Practitioner or AHP who is acting within the scope of that Practitioner's or AHP's license and who is qualified according to the Medical Staff Bylaws.

**Comment:** All orders must be dated and authenticated by the ordering Practitioner or AHP. See Section 3.5 for authentication requirements. 19 C.S.R. 30-20.094(3).

- (b) Each Practitioner or AHP is responsible for the monitoring and review of the medications the Practitioner or AHP has ordered for a patient. Pharmacy Staff may periodically request a Practitioner or AHP review an order for continuation, and each Practitioner or AHP is expected to cooperate with such request.
- (c) Medication orders shall be written according to policies and procedures, and those written by persons who do not have independent statutory authority to prescribe shall be included in the quality improvement program.
- (d) Automatic stop orders for all medications shall be established and shall include a procedure to notify the prescriber of an impending stop order. A maximum stop order shall be effective for all medications which do not have a shorter stop order. Automatic stop orders are not required when the Pharmacist continuously monitors medications to ensure that the medications are not inappropriately continued.

**Comment:** The frequency for review depends on the particular medication being prescribed. The Hospital may customize this Section 4.1 of Article 4 to integrate any standard review times for certain medications such as antibiotics, anti-coagulants, pain medications and other medications requiring laboratory monitoring. If these requirements are contained in a separate pharmacy or Medical Staff policy, such policy can be referenced here. Missouri law requirements regarding automatic stop orders have been incorporated here. 19 C.S.R. 30-20.100(33) to (36).

(e) Respiratory Services must only be provided under the orders of a qualified and licensed Practitioner or AHP who is responsible for the care of the patient and acting within the Practitioner's or AHP's scope of practice. All respiratory care orders must be documented in the patient's medical record. If an order for therapy is not adequate or explicit, further written explanation and instructions will be requested from the Practitioner or AHP.

**Comment:** The ability to order respiratory therapy Services must be designated as Clinical Privileges through the Medical Staff Bylaw credentialing process. Such Services must have an order clearly documented by the treating Practitioner or AHP. 42 C.F.R. 482.57(b); 19 C.S.R. 30-20.136.

**Comment:** Additional provisions may be added to specify frequency of review of certain orders or other limitations or requirements of the Practitioner's or AHP's ability to prescribe medications and treatment in the Hospital.

(f) The ordering Physician or independently licensed provider is not required to maintain Clinical Privileges at the Hospital to order outpatient Services. Certain outpatient services may require the ordering Physician or independently licensed provider to coordinate with a Practitioner or AHP who has appropriate Clinical Privileges to coordinate the outpatient Services to be provided.

17

## 4.2 **Standing Orders**

- (a) All medication orders shall be written in the medical record and signed by the ordering Practitioner or AHP with the exception of influenza and pneumococcal polysaccharide vaccines, which may be administered per Physician-approved Hospital policy/protocol after an assessment for contraindications. When medication therapy is based on a protocol or standing order and a specific medication order is not written, a signed copy of the protocol or of an abbreviated protocol containing the medication order parameters or of the standing order shall be placed in the medical record with the exception of Physician-approved policies/protocols for the administration of influenza and pneumococcal polysaccharide vaccines after an assessment for contraindications. The assessment for contraindications shall be dated and signed by the R.N. performing the assessment and placed in the medical record.
- (b) Hospitals may adopt policies and procedures that permit the use of standing orders to address well-defined clinical scenarios involving medication administration. An order that has been initiated for a specific patient must be added to the patient's medical record at the time of initiation, or as soon as possible thereafter. The Practitioner or AHP is responsible for acknowledging and authenticating all standing orders, with the exception of influenza and pneumococcal polysaccharide vaccines. When Hospital-based agreements, protocols or standing orders are used, they shall be approved by the pharmacy and therapeutics or equivalent committee.

Comment: Hospitals may use pre-printed and electronic standing orders, order sets and protocols for patient orders only if the Hospital: (i) Establishes that such orders and protocols have been reviewed and approved by the Medical Staff and the Hospital's nursing and pharmacy leadership; (ii) Demonstrates that such orders and protocols are consistent with nationally recognized and evidence-based guidelines; (iii) Ensures that the periodic and regular review of such orders and protocols is conducted by the Medical Staff and the Hospital's nursing and pharmacy leadership to determine the continuing usefulness and safety of the orders and protocols; and (iv) Ensures that such orders and protocols are dated, timed and authenticated promptly in the patient's medical record by the ordering Practitioner or AHP or by another Practitioner responsible for the care of the patient only if such a Practitioner or AHP is acting in accordance with state law, including scope-of-practice laws, Hospital policies, and Medical Staff Bylaws and Medical Staff Rules and Regulations. 42 C.F.R. 482.23(c); Medicare State Operations Manual, Appendix A. Missouri law does not require separate authentication by a Physician for influenza and pneumococcal polysaccharide vaccines. 19 C.S.R. 30-20.100(33) & (34).

#### 4.3 Verbal Orders

(a) Telephone/verbal orders for medication or treatment shall be accepted only from authorized individuals when it is impractical for such orders to be given in writing. Telephone/verbal orders may be accepted by an R.N. or other person qualified and authorized under Hospital policy to accept such orders. All verbal orders shall include the name of the dictating Practitioner or AHP, the date and

- time the order was taken, and the name and signature of the authorized person transcribing the order.
- (b) Verbal/telephone orders shall include the date, time and signature of the person recording them. The prescribing or covering Practitioner or AHP shall authenticate the order within seventy-two (72) hours of the patient's discharge or within thirty (30) days, whichever occurs first.

**Comment:** All orders, including verbal orders, must be dated, timed and authenticated promptly by the ordering Practitioner or AHP. 42 C.F.R. 482.24(c). Verbal orders must be accepted by an authorized individual and authenticated by the prescribing Practitioner or AHP within seventy-two (72) hours of discharge or thirty (30) days, whichever is less. 42 C.F.R. 482.23(c)(2)(i); 19 C.S.R. 30-20.100. Hospital policy must identify classes of persons authorized to accept a verbal order for drugs or biological. 42 C.F.R. 482.23(c)(3).

#### 4.4 **Do Not Resuscitate Orders**

- (a) The patient has the right to accept medical care or to refuse it to the extent permitted by law and to be informed of the medical consequences of refusal. The patient has the right to appoint a surrogate to make health care decisions on the patient's behalf to the extent permitted by law. Each Practitioner or AHP is expected to cooperate with the patient's rights or notify Hospital if the request conflicts with the Practitioner's or AHP's personal belief or ethics.
- (b) Where a patient diagnosed with a terminal illness executes a Declaration directing the withholding or withdrawal of death-prolonging procedures, the patient's Physician is responsible for completing additional documentation required by Hospital policy. If the Physician refuses to comply with the directive, the Physician must transfer the care of the patient to another Physician on the Medical Staff.
- (c) If a conflict arises with a Do Not Resuscitate Order or Advance Directive, the issue may be directed to the Hospital Ethics Committee.

**Comment:** Missouri law provides the patient rights to make a Declaration regarding life support where the patient has a terminal illness. RSMo 459.010, et seq. A patient also has the right to accept medical care or refuse it to the extent permitted by law and to be informed of the medical consequences of refusal. 19 C.S.R. 30-20.084(1)(1). The Hospital should have a policy regarding Advance Directives or Declarations of patients to specify the Hospital's procedure related to these requests. Provisions here are limited to the Provider's involvement and expectations in these situations.

# ARTICLE 5 ALLIED HEALTH PROFESSIONALS

5.1 **Categories of AHP**. The Hospital currently credentials the categories of AHPs as provided in this Article 5. Other licensed health care professionals may provide patient care services at the Hospital as Hospital employees or under contract with the Hospital, but are not issued

Privileges to provide clinical services to Hospital patients independently. All Privileges extended to AHPs must be provided consistent with the Medical Staff Bylaws.

# 5.2 Advanced Practice Registered Nurse

- (a) An Advanced Practice Registered Nurse "APRN" may provide services to a Hospital inpatient or outpatient under a collaborative practice agreement that complies with the provisions of Missouri law. If the collaborating Physician is not on the Hospital's Medical Staff, the APRN must have a Physician on the Medical Staff serve as a sponsor for supervision of Hospital patients.
- (b) An APRN may order diagnostic tests and therapies at Hospital consistent with the APRN's licensure and collaborative practice agreement. An APRN only may order medications to the extent authorized under the collaborative practice agreement. To order controlled substances, the APRN must have appropriate authority under the collaborative practice agreement and hold a current BNDD and DEA registration.
- (c) A separate privileging form is maintained for APRNs, and an APRN only may be granted privileges for services designated on the APRN privileging form.

# 5.3 Chiropractor

- (a) For a Chiropractor to provide Services to a Hospital inpatient or outpatient, the patient must be under the medical care of a Physician on the Hospital's Medical Staff.
- (b) The Hospital may provide outpatient Services to a patient on the order of a Chiropractor so long as the order is within the scope of practice of the Chiropractor. The Hospital may designate outpatient services that require the order of a Chiropractor to include designation of the Physician responsible for the medical care of the patient.

**Comment:** Medicare covers a very limited scope of services for Chiropractors and does not cover most diagnostic tests if they are ordered by a Chiropractor. The Service may be within the scope of practice to order, but may be non-covered by Medicare. The Hospital may consider scope of practice and reimbursement limitations in determining the scope of services that it permits a Chiropractor to perform or order.

(c) A separate privileging form is maintained for Chiropractors and a Chiropractor only may be granted Privileges for services designated on the Chiropractor privileging form.

# 5.4 Certified Registered Nurse Anesthetist

(a) A Certified Registered Nurse Anesthetist "CRNA" is a qualified anesthesia provider in the Hospital. A CRNA may be granted Privileges relating to the provision of anesthesia, including pre- and post-anesthesia care.

- (b) All services performed by a CRNA must be supervised by the attending surgeon. Both the CRNA and the attending surgeon are responsible for ensuring that the procedure and the level of anesthesia to be administered are within the scope of Services for which the surgeon is credentialed to perform and supervise.
- (c) A separate privileging form is maintained for CRNAs, and a CRNA may only be granted Privileges for Services designated on the CRNA privileging form.

#### 5.5 Anesthesia Assistant

- (a) An Anesthesia Assistant ("A.A.") is a qualified anesthesia provider in the Hospital. An A.A. may be granted privileges relating to the provision of anesthesia, including pre- and post-anesthesia care, to the extent such Services are within the delegation by a licensed Anesthesiologist.
- (b) An A.A. may only provide services in the Hospital under the delegation of an Anesthesiologist. An Anesthesiologist must be immediately available at all times during which the A.A. is providing Services.
- (c) A separate privileging form is maintained for A.A.s, and an A.A. may only be granted Privileges for Services designated on the A.A. privileging form.

#### 5.6 **Optometrist**

- (a) For an Optometrist to provide services to a Hospital inpatient or outpatient, the patient must be under the medical care of a Physician on the Hospital's Medical Staff.
- (b) The Hospital may provide outpatient services to a patient on the order of an Optometrist so long as the order is within the scope of practice of the Optometrist. The Hospital may designate outpatient services that require the order of an Optometrist to include designation of the Physician responsible for the medical care of the patient.
- (c) A separate privileging form is maintained for Optometrist and an Optometrist may only be granted Privileges for Services designated on the Optometrist privileging form.

#### 5.7 Physician Assistant

- (a) A Physician Assistant ("P.A.") may provide Services to a Hospital inpatient or outpatient under a Physician supervision agreement with a Physician on the Hospital's Medical Staff. The P.A.'s supervising Physician must be available in person or by phone at all times during which the P.A. provides patient care Services at the Hospital.
- (b) The Services provided by a PA are limited to:

- (i) Taking patient histories;
- (ii) Performing physical examinations of a patient;
- (iii) Performing or assisting in the performance of routine office laboratory and patient screening procedures;
- (iv) Performing routine therapeutic procedures;
- (v) Recording diagnostic impressions and evaluating situations calling for attention of a Physician to institute treatment procedures;
- (vi) Instructing and counseling patients regarding mental and physical health using procedures reviewed and approved by a Physician;
- (vii) Assisting the supervising Physician in institutional settings, including reviewing of treatment plans, ordering of tests and diagnostic laboratory and radiological Services, and ordering of therapies, using procedures reviewed and approved by Physician;
- (viii) Assisting in surgery;
- (ix) Performing such other tasks not prohibited by law under the supervision of a Physician as the P.A. has been trained, is proficient to perform and has been granted applicable Clinical Privileges; and
- (x) Prescribing drugs only to the extent permitted under the Physician supervision agreement. Any controlled substances may only be prescribed as permitted under the Physician supervision agreement and the P.A.'s BNDD and DEA registration.
- (c) Subject to the limitations in Subsection (b), a P.A. may order diagnostic tests and therapies at Hospital consistent with the P.A.'s licensure and physician supervision agreement. A P.A. may only order medications to the extent authorized under the Physician supervision agreement. To order controlled substances, the P.A. must have appropriate authority under the Physician supervision agreement and hold a current BNDD and DEA registration.
- (d) A separate privileging form is maintained for P.A.s, and a P.A. may only be granted Privileges for Services designated on the P.A. privileging form.

**Comment:** The Medical Staff Bylaws require implementation of policy or rules and regulations to specify which categories of potential AHPs are actually offered Privileges within the Hospital. This Article 5 provides an example of how a Hospital could develop more specific policies regarding AHPs, which could be included in the Rules and Regulations or independent Medical Staff policy.

# ARTICLE 6 RESTRAINTS AND SECLUSION

- 6.1 **Patient Rights.** All patients have the right to be free from physical or mental abuse and corporal punishment. All patients have the right to be free from restraint or seclusion of any form, imposed as a means of coercion, discipline, convenience or retaliation by Staff. Restraint or seclusion may only be imposed to ensure the immediate physical safety of the patient, Hospital personnel, Staff Members or others, and must be discontinued at the earliest possible time. Restraint or seclusion shall never be used as a punishment or for the convenience of the staff.
  - The term "restraint" includes either a physical restraint or a drug that is being (a) used as a restraint.
  - A physical restraint is any manual method or physical or mechanical device, (b) material or equipment attached or adjacent to the patient's body that the patient cannot easily remove that restricts freedom of movement or normal access to one's body.
  - (c) A drug used as a restraint is a medication used to control behavior or to restrict the patient's freedom of movement and is not a standard treatment for the patient's medical or psychiatric condition.
  - (d) A restraint does not include devices, such as orthopedically prescribed devices, surgical dressing or bandages, protective helmets, or other methods, that involve the physical holding of a patient for the purpose of conducting routine physical examinations or tests, or to protect the patient from falling out of bed, or to permit the patient to participate in activities without the risk of physical harm (this does not include a physical escort).
  - "Seclusion" is the involuntary isolation of a patient alone in a room where the (e) patient's freedom to leave is restricted.
- 6.2 Use of Restraints or Seclusion. Restraint or seclusion may only be used when less restrictive interventions have been determined to be ineffective to protect the patient, Hospital personnel or others from harm. The type or technique of restraint or seclusion used must be the least restrictive intervention that will be effective to protect the patient, Hospital personnel or others from harm.
- 6.3 **Physician Order.** Use of restraint or seclusion must be in accordance with the order of a Physician with appropriate Clinical Privileges under the Medical Staff Bylaws. The order for restraint or seclusion must be:
  - (a) Followed by consultation with the patient's treating Physician, as soon as possible, if restraint or seclusion is not ordered by the patient's treating Physician;
  - In accordance with a written modification to the patient's plan of care; (b)

- (c) Signed within one (1) hour of implementation;
- (d) Implemented in the least restrictive manner possible;
- (e) In accordance with safe and appropriate restraining techniques;
- (f) Ended at the earliest possible time; and
- (g) Itime limited and never written as a standing order or on a PRN basis.

**Comment:** The use of restraint or seclusion must be in accordance with the order of a Physician or other licensed independent Practitioner who is responsible for the care of the patient and authorized to order restraint or seclusion by Hospital policy in accordance with State law. 42 C.F.R. 482.13(e)(5). With the exception of certain emergency situations, AHPs are not considered independent Practitioners for purposes of ordering restraints under Missouri law.

- **Renewal of Order.** The order may only be renewed in accordance with the following limits for up to a total of twenty-four (24) hours:
  - (a) three (3) hours for adults eighteen (18) years of age and older;
  - (b) two (2) hours for children and adolescents nine (9) to seventeen (17) years of age;
  - (c) one (1) hour for children under nine (9) years of age; or
  - (d) after twenty-four (24) hours, before writing a new order for the use of restraint or seclusion, the Physician responsible for the care of the patient must see and assess the patient.
- 6.5 **Face-to-Face Evaluation**. When restraint or seclusion is used for the management of violent or self-destructive behavior that jeopardizes the immediate physical safety of a patient, Hospital personnel or others, the patient must be seen face-to-face within one (1) hour after the initiation of restraint or seclusion by a Physician, APRN, a licensed independent Practitioner, R.N. or P.A/ who has been trained on the restraint and seclusion policies and is approved by the Medical Staff for such evaluation. The evaluation must include the patient's immediate situation, the patient's reaction to the intervention, the patient's medical and behavioral condition; and the need to continue or terminate the restraint or seclusion. If the face-to-face evaluation is performed by an APRN, R.N. or P.A., such individual must consult the attending Physician as soon as possible following the completion of the examination.
- Monitoring. The condition of the patient who is restrained or secluded must be monitored by a Physician, other licensed independent Practitioner or trained Staff. Such monitoring of the patient's condition shall be at a frequency determined by the treating Physician, which shall be no less than once per each fifteen (15) minutes.
- 6.7 **Documentation**. Documentation must be present showing that the patient is in danger to the patient and/or others. Specifically, when restraint or seclusion is used, there must be

documentation in the patient's medical record of the following: (i) the one (1)-hour faceto-face medical and behavioral evaluation if restraint or seclusion is used to manage violent or self-destructive behavior; (ii) a description of the patient's behavior and the intervention used; (iii) alternatives or other less-restrictive interventions attempted (as applicable); (iv) the patient's condition or symptoms(s) that warranted the use of the restraint and seclusion; (v) the patient's response to the interventions(s), including the rationale for continued use of the intervention; and (vi) the reason for the restriction, and the time of starting and ending the restriction. Nursing notes should reflect the continuous monitoring of the secluded or restrained patient. A Physician must sign a statement explaining the necessity for the use of any restraint or seclusion and shall make such statement part of the patient's permanent medical records.

- 6.8 Simultaneous Restraint and Seclusion. Simultaneous restraint and seclusion use is only permitted if the patient is continually monitored (i) face-to-face by an assigned, trained Staff Member; or (ii) by trained Staff using both video and audio equipment in close proximity to the patient.
- 6.9 **Intake and Output**. Patients must have the opportunity for regular meals, adequate fluids and use of the bathroom.
- 6.10 **Training**. Physicians and other Hospital personnel shall complete training on use of restraints and seclusion as required under Hospital Policy.

**Comment:** A patient's rights include the freedom from restraint or seclusion. The benefits of restraint to protect the patient or others must outweigh the limitation of the patient's rights. The above sections address requirements for the use of restraints and seclusion generally set forth at 42 C.F.R. 482.13(f) and 19 C.S.R. 30-20.132(1)(B). Hospitals should develop and implement a training program regarding restraints and seclusion as required under 42 C.F.R. 482.13(f).

## **ARTICLE 7 EMERGENCY SERVICES**

7.1 **Organization**. Emergency Services shall be organized under the direction of a qualified member of the Medical Staff and shall be integrated with other Departments of the Hospital. The duties of the directing member of the Medical Staff shall be adopted by the Governing Body.

If emergency Services are provided at a Hospital, the Services must be organized **Comment:** under the direction of a qualified member of the Medical Staff and must be integrated with other Departments of the Hospital. The policies and procedures governing medical care in the Emergency Department must be established by and be a continuing responsibility of the Medical Staff. 42 C.S.R. 482.55; 19 C.F.R 30-20.092(3).

7.2 On-Call Coverage. Medical Staff Members, including specialists and sub-specialists, are required to participate in on-call coverage as determined necessary by Hospital and in accordance with the Medical Staff Bylaws.

Hospitals operating dedicated Emergency Departments are required to maintain **Comment:** an on-call list of Physicians who are on the Hospital's Medical Staff or who have Privileges at the Hospital (or who are on the Medical Staff or have Privileges at another hospital participating in a formal community call plan) available to provide treatment necessary after the initial examination to stabilize individuals with emergency medical conditions who are receiving services under the Emergency Medical Treatment and Labor Act ("EMTALA") in accordance with the resources available to the Hospital. 42 C.F.R. 489.20(r)(2); 19 C.S.R. 30-20.092(4). This Section 7.2 of Article 7 may be customized to incorporate Hospital policy regarding the ability of Physicians to schedule elective surgery while on-call, the ability of Physicians to have simultaneous on-call duties, or incorporation of a formal community call plan if on-call coverage is coordinated with other area hospitals. 42 C.F.R. 489.24(j); 42 C.F.R. 489.20(r)(2).

7.3 **Response Time**. On-call Practitioners shall respond within a reasonable period of time after receiving a page or telephone call from emergency personnel. Surgeons assigned to emergency surgery call coverage must be available to arrive at the Hospital within thirty (30) minutes of being summoned.

**Comment:** Hospitals are required to establish policies and procedures requiring response from on-call Physicians within a reasonable period of time. An EMTALA enforcement action may be taken against both a Hospital and a Physician when a Physician who is on the Hospital's on-call list fails or refuses to appear within a reasonable period of time after being notified to appear. Policies and procedures also should address how Hospital personnel respond to situations in which a particular specialty is not available or the on-call Physician cannot respond due to circumstances beyond the Physician's control. 42 C.F.R. 489.24(j). General call coverage allows the Hospital to specify reasonable response times, but Missouri law requires surgeons to arrive at the Hospital within thirty (30) minutes. 19 C.S.R. 30-20.092(5).

- 7.4 **Screening Examination**. The Hospital will provide an appropriate medical screening examination within its capability, including ancillary Services routinely available to the Emergency Department, for persons (who are not already inpatients) on the Hospital's property requesting examination for what might be an emergency medical condition.
  - (a) An "emergency medical condition" is a medical condition manifesting itself by acute symptoms of sufficient severity (including severe pain, psychiatric disturbances and/or symptoms of substance abuse) such that the absence of immediate medical attention could reasonably be expected to result in: (i) placing the health of the individual (or, with respect to a pregnant woman, the health of the woman or her unborn child) in serious jeopardy; (ii) serious impairment to bodily functions; or (iii) serious dysfunction of any bodily organ or part. With respect to a pregnant woman who is having contractions, an emergency medical condition exists if there is inadequate time to affect a safe transfer to another hospital before delivery or transfer may pose a threat to the health or safety of the woman or the unborn child.
  - (b) The "Hospital's property" means the entire main Hospital campus, including the parking lot, sidewalk and driveway, but excluding other areas or structures of the Hospital's main building that are not part of the Hospital, such as Physician offices, rural health centers, skilled nursing facilities or other entities that

- participate separately under Medicare, as well as restaurants, shops or other nonmedical facilities.
- The screening examination may be performed by Medical Staff Practitioners, (c) emergency room Physicians, emergency room Registered Professional Nurses, and labor and delivery Registered Professional Nurses. Psychiatric patients will be screened by individuals qualified by training and experience, including Physicians, psychiatric Registered Professional Nurses, master's prepared Psychologists and master's prepared social workers.

**Comment:** The medical screening examination must be conducted by individuals determined to be qualified by Hospital Bylaws or rules and regulations. 42 C.F.R. 489.24(a)(i); 19 C.S.R. 30-20.092(12)(C).

- 7.5 **Stabilizing Treatment**. If an emergency medical condition is found to exist, the Hospital will provide necessary stabilizing treatment or an appropriate transfer.
  - "Stabilizing treatment" is considered as the treatment necessary to assure, within a (a) reasonable medical probability, that no material deterioration of the condition is likely to result from or occur during the transfer of the individual from a facility or that, with respect to a pregnant woman who is having contractions, the woman delivers the child and the placenta. If an individual at Hospital has an emergency medical condition that has not been stabilized, the Hospital will not transfer the individual unless it is an "appropriate transfer" (as defined below).
  - (b) To make an "appropriate transfer" to another medical care facility, the Hospital must (i) provide the stabilizing medical treatment within its capacity minimizing the risk to the individual or to the woman and unborn child; (ii) verify that the receiving facility has the space and qualified personnel available for the treatment of the individual; (iii) verify that the receiving hospital has agreed to accept the transfer of the individual and to provide the appropriate medical treatment; (iv) send pertinent medical records available at the time of the transfer to the receiving hospital (including available history, records related to the emergency medical condition, observations of signs or symptoms, preliminary diagnosis, results of diagnostic studies, treatment provided, and the name/address of any on-call Physician who refused or failed to appear within a reasonable time to provide necessary stabilizing treatment); and (v) effect the transfer through qualified persons and transportation equipment, including life support measures.
  - (c) An appropriate transfer, defined above, is only permitted where the patient (or the patient's authorized representative) requests the transfer, or a Physician has signed a certification that, based upon information available at the time of transfer, the medical benefits reasonably expected from the provision of appropriate medical treatment at another facility outweigh the increased risk to the individual or, in the case of a woman in labor, to the woman or unborn child.

- (i) A patient's (or authorized Representative's) request for transfer must be in writing and must indicate the reasons for the request and indicate that the patient is aware of the risks and benefits of transfer.
- (ii) A written Physician certification must summarize the risks and benefits of the transfer.
- (iii) If a Physician is not physically present in the Emergency Department at the time of transfer, a qualified medical person may sign the certification after consultation with a Physician who agrees with the certification. A qualified medical person is a P.A., APRN, emergency room Registered Professional Nurse, or labor and delivery Registered Professional Nurse. The consulting Physician must subsequently countersign the certification.

**Comment:** Missouri law incorporates the requirements of the Emergency Medical Treatment and Active Labor Act ("EMTALA") to all hospitals. 19 C.S.R. 30-20.092(12)(C). Medicare-participating hospitals operating Emergency Departments are required to provide appropriate stabilization treatment or transfer of the patient. 42 C.F.R. 489.24(d) & (e). The Hospital Medical Staff Bylaws or Rules and Regulations must specify the qualifications for individuals permitted to sign the transfer for certification. 42 C.F.R. 489.24(e)(1)(C). Failure to appropriately provide and document screening, stabilization and/or transfer of patients with emergency medical conditions may result in termination of the Hospital's Medicare provider agreement. 42 C.F.R. 489.24(g).

7.6 **Off-Campus Departments**. When a patient with an emergency medical condition presents to an off-campus Hospital department that does not include an Emergency Department, the patient will be provided screening and stabilization services at such off-campus department consistent with available resources and Hospital policy.

**Comment:** Medicare Conditions of Participation require that if the Hospital provides emergency services, but operates off-campus Hospital Departments that do not include emergency services, the Medical Staff must adopt written policies and procedures regarding how the off-campus Department evaluates and refers patients with emergency medical conditions. 42 C.F.R. 482.12(f)(3).

7.7 **Reporting**. The Hospital will report to the Centers for Medicare & Medicaid Services or the Missouri Department of Health and Senior Services any time the Hospital believes it has received an individual who has been transferred from another hospital in violation of EMTALA. This report must be made within seventy-two (72) hours of the occurrence.

**Comment:** Hospitals are required to report patient transfers in violation of EMTALA. 42 C.F.R. 489.20(m), State Operations Manual, Appendix V, Tag A-2401/C-2401.

# ARTICLE 8 AUTOPSIES

8.1 **Securing Autopsies**. Every Member of the Medical Staff is expected to be actively interested in securing autopsies, and should attempt to secure an autopsy in all cases of

unusual deaths and of medical-legal and educational interest. No autopsy shall be performed without written consent of a relative or legally authorized agent. All autopsies shall be performed by the Hospital pathologist or by a pathologist to whom the Physician may delegate the duty. The attending Physician shall be notified when an autopsy is being performed.

**Comment:** Hospitals are required to encourage the performance of autopsies. The Hospital must establish policies regarding the method of documenting consent to perform an autopsy. Specific procedures may be included in this Section 8.1 of Article 8 or specified in a separate Hospital policy. 42 C.F.R. 482.22(d); 19 C.S.R. 30-20.098(12).